ABSTRACT

Title of dissertation:	DISTRIBUTED ESTIMATION AND STABILITY OF EVOLUTIONARY GAME DYNAMICS WITH APPLICATIONS TO STUDY OF ANIMAL MOTIO						
	Shinkyu Park, Doctor of Philosophy, 2015						
Dissertation directed by:	Professor Nuno C. Martins Department of Electrical and Computer Engineering						

In this dissertation, we consider three problems: in the first we investigate distributed state estimation of linear time-invariant (LTI) plants; in the second we study optimal remote state estimation of Markov processes; while in the third we examine stability of evolutionary game dynamics in large populations.

Problem 1: Consider that an autonomous LTI plant is given and that each member of a network of LTI observers accesses a portion of the output of the plant. The dissemination of information within the network is dictated by a pre-specified directed graph in which each vertex represents an observer. This work proposes a distributed estimation scheme that is a natural generalization of consensus in which each observer computes its own state estimate using only the portion of the output vector accessible to it and the state estimates of other observers that are available to it, according to the graph. Unlike straightforward high-order solutions in which each observer broadcasts its measurements throughout the network, the average size of the state of each observer in the proposed scheme does not exceed the order of the plant plus one. We determine necessary and sufficient conditions for the existence of a parameter choice for which the proposed scheme attains asymptotic omniscience of the state of the plant at all observers. The conditions reduce to certain detectability requirements that imply that if omniscience is not possible under the proposed scheme then it is not viable under any other scheme – including higher order LTI, nonlinear, and time-varying ones – subject to the same graph. We apply the proposed scheme to distributed tracking of a group of water buffaloes.

Problem 2: Consider a two-block remote estimation framework in which a sensing unit accesses the full state of a Markov process and decides whether to transmit information about the state to a remotely located estimator given that each transmission incurs a communication cost. The estimator finds the best state estimate of the process using the information received from the sensing unit. The main purpose of this work is to design transmission policies and estimation rules that dictate decision making of the sensing unit and estimator, respectively, and that are optimal for a cost functional which combines the expectation of squared estimation error and communication costs. Our main results establish the existence of transmission policies and estimation policies and estimation rules that are jointly optimal, and propose an iterative procedure to find ones. Our convergence analysis shows that the sequence of sub-optimal solutions generated by the proposed procedure has a convergent subsequence, and the limit of any convergent subsequence is a person-by-person optimal solution.¹ We apply the proposed scheme to remote estimation of location of a water buffalo.

Problem 3: We investigate an energy conservation and dissipation (passivity) as-

¹The definitions of joint optimality and person-by-person optimality are given in Definition 3.1.2 and Definition 3.1.3, respectively.

pect of evolutionary dynamics in evolutionary game theory. We define a notion of passivity for evolutionary dynamics, and describe conditions under which dynamics exhibit passivity. For dynamics that are defined on a finite-dimensional state space, we show that the conditions can be characterized in connection with state-space realizations of the dynamics. In addition, we establish stability of passive dynamics in terms of dissipation of stored energy defined by passivity, and present stability results in population games. We provide implications of stability for various passive dynamics both analytically and by means of numerical simulations.

DISTRIBUTED ESTIMATION AND STABILITY OF EVOLUTIONARY GAME DYNAMICS WITH APPLICATIONS TO STUDY OF ANIMAL MOTION

by

Shinkyu Park

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park in partial fulfillment of the requirements for the degree of Doctor of Philosophy 2015

Advisory Committee: Professor Nuno C. Martins, Chair/Advisor Professor P. S. Krishnaprasad Professor Richard La Professor André Tits Professor Nikhil Chopra © Copyright by

Shinkyu Park

2015

To my family

Acknowledgments

I would like to thank Prof. Nuno C. Martins for giving me an opportunity to work in his research group, and for being my adviser. He taught me how to look for interesting research problems, and how to present findings in research community. Writing a paper with him was one of memorable moments during my PhD study. Thanks to his infinite support, I could enjoy every moment and overcome the most difficult time during the study. Prof. P. S. Krishnaprasad always encouraged me to "make a contribution to science." Because of his moral support and encouragement, I was able to achieve beyond what I could possibly imagine. Also I learned a lot from his control courses and from discussions outside the class. When I was first admitted to UMD, it was Prof. André Tits who encouraged me to take courses in mathematical analysis, from which I have established mathematical foundations for my research. I am thankful for always keeping his door open for me. Prof. Richard La gave me the most critical comments on my evolutionary game problem. Thanks to his comments, I was able to make a progress in some part of the problem. I would like to thank Prof. Nikhil Chopra for being in my dissertation committee as the Dean's representative. His feedback and support were very important to improve my dissertation. Prof. Jeff S. Shamma convinced me to pursue an interesting problem in evolutionary game theory, and is responsible for steering my research toward one of the main topics of my dissertation. I am grateful for his time for serious discussions during my visit to Georgia Tech. I would like to thank Dr. Konrad Aschenbach for the collaboration and help. Working with him in the research project was one of joyful moments during my study.

I also would like to thank ONR UMD-AppEl Center, the Multiscale Systems Center, one of six research centers funded under the Focus Center Research Program, Naval Air Warfare Center Aircraft Division / UMD Cooperative Agreement, AFOSR MURI FA955009108538, NSF grants 1135726 and 0931878 for financial supports.

Table of Contents

Lis	st of F	igures		viii										
1	Intro	duction		1										
	1.1	Design	of Distributed LTI Observers for State Omniscience	1										
		1.1.1	Summary of the Main Contributions	3										
	1.2	Optima	Optimal Remote State Estimation of Markov Processes											
		Summary of the Main Contributions	6											
	1.3	Evoluti	olutionary Game Dynamics and Passivity											
		1.3.1	Summary of the Main Contributions	9										
		1.3.2	Stability Concept and Landscape Metaphor	11										
2	Desi	gn of Di	istributed LTI Observers for State Omniscience	12										
	2.1	Proble	m Formulation	12										
		2.1.1	Notation	12										
		2.1.2	Problem Description	13										
		2.1.3	Comparative Survey of Related Work	14										
	2.2	Main F	Result	17										
		2.2.1	Additional Remarks on the Proposed Class of Distributed Observers	19										
			2.2.1.1 The Effect of Noise on the Estimation Performance	19										
			2.2.1.2 The Role of the Augmented States	19										
			2.2.1.3 Complexity of the Proposed Scheme	20										
	2.3	Applic	ation to the Synchronization of Coupled Multi-agent Systems	21										
	2.4	Proof o	of the Main Theorem	31										
		2.4.1	Key Results on Weighted Laplacian Matrices	32										
		2.4.2	A Brief Introduction to Stabilization via Decentralized Control	34										
		2.4.3	Additional Preliminary Results	36										
		2.4.4	Proof of Theorem 2.2.2	39										
	2.5	Applic	ation to Tracking of Animal Groups and Experimental Results	43										
	2.6	Summa	ary and Future Work	52										

3	Opti	mal Rer	note State Estimation of Markov Processes	53
	3.1	Proble	m Formulation	53
		3.1.1	Notation and Terminologies	53
		3.1.2	Problem Description	54
		3.1.3	Comparative Survey of Related Work	61
	3.2	Two-P	layer Optimal Stopping Problem	63
		3.2.1	Definitions and Preliminary Results	66
		3.2.2	Existence of a Jointly Optimal Solution	69
		3.2.3	Iterative Procedure for Finding a Person-by-Person Optimal So-	
			lution	71
	3.3	Optima	al Remote State Estimation	77
	3.4	Applic	ation to Specific System Models	78
		3.4.1	Linear System Models	78
		3.4.2	Self-Propelled Particle Models	81
	3.5	Applic	ation to Animal Tracking and Experimental Results	85
	3.6	Summ	ary and Future Work	94
4	Evol	utionary	y Game Dynamics and Passivity	95
	4.1	Backg	round	95
		4.1.1	Notation	95
		4.1.2	Population Games and Evolutionary Dynamics	96
			4.1.2.1 Population Games	96
			4.1.2.2 Evolutionary Dynamics	97
	4.2	Passivi	ity of Evolutionary Dynamics	99
		4.2.1	Definition of Passivity for Evolutionary Dynamics	99
		4.2.2	Characterization of Passivity of Evolutionary Dynamics	101
		4.2.3	Properties of Passive Evolutionary Dynamics	105
			4.2.3.1 Payoff Monotonicity and Passivity	105
			4.2.3.2 Equivalence to Closed-loop Stability	107
		~	4.2.3.3 Effect of Control Costs on Passivity	109
	4.3	Stabili	ty of Passive Evolutionary Dynamics	111
	4.4	Numer	rical Examples	117
		4.4.1	Replicator dynamics and BNN dynamics under a cumulative pay-	
			off function	117
		4.4.2	BNN dynamics and logit dynamics in the Hypnodisk game	121
	4.5	Summ	ary and Future Work	126
٨	٨	iliony De	asults for Chapter 2	127
A			utational Considerations	127
	A.1		Finding Source Components	127
		Λ 1 2	Computing an Omniscience, achieving Parameter Choice	127
		A.I. 2	A 1.2.1 Computation of $\mathbf{W} - (\mathbf{w}_{ij})$	12/
			A 1.2.2 Computation of $\mathbf{K} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$	120
	Δ 2	Nonde	$\mathbf{A}_{1,2,2} \text{computation of } \mathbf{I}_{1,1} \mathbf{x}_{i}, \mathbf{x}_{i}, \mathbf{x}_{i}_{i \in \mathbb{V}} \dots $	133
	Δ3	Prelim	inary Concepts and Proof of Proposition 2.3.2	130
	11.5	1 IVIIII	$\frac{1}{2}$	1.74

		A.3.1	Α	Ch	oic	e of	\mathbf{W}	_	(v	\mathbf{v}_{ij}) _{i.}	i∈ĭ	γ,	$\{\mathbf{I}\}$	\mathbf{X}_i	, F	\mathbf{P}_i	Q	$_{i},$	\mathbf{S}_i	$_{i\in I}$	V		•	•	•	•				137
		A.3.2	А	. Ch	oic	e of	`{ k	X_i^d, I	\mathbf{P}_i^a	l, \mathbf{C}	\mathbf{Q}_i^d	, S	S_i^{d}	}	-77						•										140
		A.3.3	P	roof	f of	Pro	pos	sitio	n	2.3	3.2	2	•	• 10	- · ·																140
	A.4	Proofs	s of	Ler	nm	as 2	.4.3	3 an	d ź	2.4	1.4	-																			142
	A.5	Prelimi	nina	ry I	Resi	ults	and	l Pr	00	of c	of '	Th	ee	ore	em	2	.4.	5													143
		A.5.1	S	truc	tura	al Co	ont	roll	ab	ili	ty	an	d	O	bse	erv	val	oil	ity	· .						•	•				144
		A.5.2	А	. Ke	y L	emr	na						•													•	•				145
		A.5.3	P	roof	f of	The	ore	em 2	2.4	1.5	•	•	•	•	•	•	•	•	•		•	•	•	•	•	•	•	•		•	148
В	Auxi	liary Re	esu	lts f	for (Char	pter	: 3																							151
	B .1	On Pro	odu	ct N	Лetr	ric S	Space	ce																							151
	B.2	On Rai	ndo	omi	zed	Pol	icie	es																							152
	B.3	Prelimi	nina	ry (Con	cept	ts a	nd]	Re	su	lts	5																			153
	B. 4	Proof o	of I	Prop	oosi	tion	3.2	2.14	Ι.																						156
	B.5	Proofs	s of	Pro	pos	itio	n 3.	.2.1	6 :	an	d I	Le	m	ma	a 3	.2	.1	8								•	•				158
	B.6	Proof o	of I	Lem	ıma	3.2	.24						•													•	•				167
	B.7	Proof o	of 7	Theo	orer	n 3.	3.1	•		•	•	•	•	•	•	•	•	•	•		•	•	•	•	•	•	•	•	•	•	178
С	Auxi	liary Re	esu	lts f	for (Chaj	pter	:4																							181
	C.1	Proof of	of I	Prop	posi	tion	4.2	2.4			•		•									•				•	•				181
	C.2	Proof o	of I	Prop	posi	tion	4.2	2.5		•																	•	•	•		182
	C.3	Proof of	of I	Prop	posi	tion	4.2	2.7			•		•									•				•	•				184
	C.4	Proof o	of I	Prop	posi	tion	4.2	2.8		•		•															•	•	•		185
	C.5	Proof o	of I	Prop	posi	tion	4.2	2.10).	•		•															•	•	•		187
	C.6	Proof o	of I	Prop	posi	tion	4.2	2.12	2.		•		•									•				•	•				189
	C.7	Proof o	of (Cord	olla	ry 4	.2.1	13		•																	•	•	•		189
	C.8	Proof o	of I	Prop	posi	tion	4.2	2.14	ι.	•																	•	•	•		190
	C.9	Proof o	of 7	The	orer	n 4.	3.1						•													•	•				192
	C.10	Proof o	of I	Prop	oosi	tion	4.3	3.2					•													•	•				195
	C.11	Proof o	of I	Prop	oosi	tion	4.3	3.3					•													•	•				196
	C.12	Proof o	of I	Prop	oosi	tion	4.3	3.5					•											•		•	•				197
	C.13	Proof o	of I	Prop	osi	tion	4.3	3.7			•		•	•	• •	•	•	•	•		•	•	•	•	•	•	•	•		•	198

Bibliography

List of Figures

1.1	A framework for distributed state estimation.	2
1.2	A framework for remote state estimation with a Markov process (P), a sensing unit (S), and an estimator (E).	5
2.1	A communication graph \mathcal{G} for Example 2.2.3	18
2.2	A diagram showing an example of an overall closed-loop system that con- sists of a LTI multi-agent system and distributed controller. See Exam-	22
22	A simulation result of Example 2.3.3 which deniets the synchronization	23
2.3	of the first components $\chi_1^{(1)}(k), \chi_1^{(2)}(k), \chi_1^{(3)}(k), \chi_1^{(4)}(k)$ of the states of	
	agent 1, 2, 3, 4	30
2.4	A precedence diagram for the proof of Theorem 2.2.2	32
2.5	System overview of the tracking device in the animal-borne wireless cam-	
	era network	44
2.6	A screenshot of GPS tracks of water buffaloes in the Google earth	
2.7	(Timespan: 2015-08-06100:002 \sim 2015-08-06104:00:00Z) Original trajectories of water buffaloes in a local North East Up (NEU) coordinate system (The origin of the coordinate system: Latitude = -	45
	18.9279877268328, Longitude = 34.5457567343343)	46
2.8	A communication graph \mathcal{G} for designing a distributed observer	49
2.9	Estimated trajectories of water buffaloes using the proposed distributed	
	estimation scheme in a local NEU coordinate system	50
2.10	Normalized estimation error at every node (Total traveled distance:	
	Buffalo $1 = 1694 m$, Buffalo $2 = 1505.8 m$, Buffalo $3 = 1981.8 m$, Buffalo	
	4 = 2129 m)	51
3.1	The problem solving strategy for Problem 3.1.1	60
3.2	A screenshot of the GPS track of a water buffalo in the Google earth $(\text{Timespan}; 2015, 08, 06\text{T}00:00; 007, 2015, 08, 06\text{T}06:00; 007)$	97
	$(111105pan, 2013-00-00100,002 \sim 2013-00-00100,002)$	0/

3.3	The x-coordinate $(p_{x,k})$, y-coordinate $(p_{y,k})$, and heading angle θ_k of a portion of the GPS track (contained in the red rectangle in Figure 3.2) in	
	a local NEU coordinate system. (Timespan: 2015-08-06T05:40:00Z \sim	
	2015-08-06T05:53:00Z / The origin of the coordinate system :	
	Latitude = -18.9401136372457, Longitude = 34.5337888580266)	88
3.4	Comparisons between the probability density functions of \mathbf{r}_k and ϕ_k un-	
	der the computed parameter choices and the GPS data.	91
3.5	Estimated trajectory of the water buffalo by the proposed remote estima-	
	tion scheme	92
3.6	Estimation error of the remote estimation scheme	93
4.1	A closed-loop obtained by a feedback interconnection of payoff operators	
	(4.1) and evolutionary dynamics (4.2) .	112
4.2	Simulation results for the replicator dynamics under a cumulative payoff	
	given by (4.29). $E(x) = \sum_{i=1}^{3} x_i^* \ln \frac{x_i^*}{x}$ where $x^* = \frac{1}{3} \cdot 1$	119
4.3	Simulation results for the BNN dynamics under a cumulative payoff given	
	by (4.29). $S_{BNN}(p,x) = \frac{1}{2} \sum_{i=1}^{3} [\hat{p}_i]_+^2 \dots \dots$	120
4.4	Simulation results for the BNN dynamics in the Hypnodisk game (4.31).	
	$S_{BNN}(p,x) = \frac{1}{2} \sum_{i=1}^{3} [\hat{p}_i]_{+} \dots \dots$	123
4.5	Simulation results for the logit dynamics ($\eta' = 0.36$) in the Hypnodisk	
	game (4.31). $S_{logit}(p, x) = \max_{y \in int(\mathbb{X})} \left[p^T y - \eta' \cdot \sum_{i=1}^3 y_i \ln y_i \right] - p^T x +$	
	$\eta' \cdot \sum_{i=1}^{3} x_i \ln x_i$	124
4.6	Simulation results for the perturbed BNN dynamics ($\eta' = 0.36$) in the	
	Hypnodisk game (4.31). $S_{PBNN}(p, x) = \frac{1}{2} \sum_{i=1}^{3} \left[\tilde{p}_i - \tilde{p}^T x \right]$	125
A.1	Diagrams depicting a design procedure for finding a distributed controller.	136

Chapter 1: Introduction

We consider the following three problems: (i) distributed state estimation of LTI plants (Chapter 2), (ii) remote state estimation of Markov processes (Chapter 3), and (iii) stability of evolutionary game dynamics (Chapter 4). The main objective and summary of main contributions to each problem are described in this chapter.

Main results of each chapter can be applied to study of animal motion: The estimation schemes that will be studied in Chapter 2 and Chapter 3 can be applied to tracking of animal groups. The data from tracking animal groups are then analyzed to identify and study collective animal motion. Based on results on stability of evolutionary game dynamics, which are presented in Chapter 4, we can find a reasoning over which certain collective motion emerges in animal groups.

1.1 Design of Distributed LTI Observers for State Omniscience

Consider the following linear time-invariant (LTI) plant in state-space form¹:

$$x(k+1) = Ax(k)$$

$$y(k) = Cx(k)$$
(1.1)

¹In order to simplify the notation, without loss of generality, we omit noise terms in the state-space equation (1.1). See Section 2.2.1.1 for more discussions.



Figure 1.1: A framework for distributed state estimation.

where $x(k) \in \mathbb{R}^n$ and $y(k) = \left(y_1^T(k) \dots y_m^T(k)\right)^T$, with $y_i(k) = C_i x(k) \in \mathbb{R}^{r_i}$, represent the state and output, respectively.

We consider the problem of designing a group of m observers so that each observer can asymptotically resolve the entire state x. Information exchange among observers is constrained by a pre-selected directed graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ with $\mathbb{V} = \{1, \dots, m\}$, where each vertex in \mathbb{V} represents an observer and the edges in $\mathbb{E} \subseteq \mathbb{V} \times \mathbb{V}$ determine the viability and direction of information transfer. We refer to a given \mathcal{G} as the *communication graph* and we denote a group of m observers equipped with \mathcal{G} , with each observer accessing an element of $\{y_1, \dots, y_m\}$, as a *distributed observer* (see Figure 1.1 for an illustration).

The internal state of an observer consists of a state estimate \hat{x}_i and an additional state z_i that is updated based on its portion y_i of the output of the plant and the state estimates of the other observers connected to it via the edges of \mathcal{G} . We later refer to z_i as the augmented state of observer *i*. A distributed observer is said to achieve *omniscience asymptotically* if $\lim_{k\to\infty} ||\hat{x}_i(k) - x(k)|| = 0$ holds for all *i* in \mathbb{V} , i.e., the state estimate at every observer converges to the state of the plant. **Our main goals are:** (i) Given a plant (1.1) and a graph \mathcal{G} , we wish to determine necessary and sufficient conditions for the existence of a LTI distributed observer that achieves omniscience asymptotically. (ii) Provided it exists, we want to find an omniscience-achieving solution in which the dimension μ_i of z_i , i.e., $z_i \in \mathbb{R}^{\mu_i}$, satisfies the following *scalability condition*:

$$\sum_{i=1}^{m} \mu_i < m \tag{1.2}$$

The main technical challenges are: (i) Each observer accesses only a portion of the output of the plant. Hence, unless the pair (A, C_i) is detectable for all i in \mathbb{V} , state omniscience requires information exchange among observers. The exception being the trivial case in which the state of the plant can be resolved from the portion of the output available to every observer. (ii) Notice that (1.2) rules out simple LTI schemes in which observers share their measurements throughout the network.² (iii) The existence of an omniscience-achieving scheme that conforms with both \mathcal{G} and (1.2) cannot be established by existing results on observer design.

1.1.1 Summary of the Main Contributions

In order to achieve the stated goals, this work focuses on the following two contributions: (i) We propose a parametrized class of LTI distributed observers within which information exchange conforms to a pre-specified directed communication graph \mathcal{G} . (ii) We find necessary and sufficient conditions for the existence of a parameter choice for the aforementioned class that is omniscience-achieving and satisfies the scalability constraint (1.2). We also outline a method to compute such a parameter choice, provided it exists.

²See Section 2.2.1.3 for more details.

In Section 2.4 we provide a detailed analysis that hinges on the fact that asymptotic omniscience for the proposed class of distributed observers can be cast as the stabilization of certain LTI systems via fully decentralized output feedback. Using this analogy, in Theorem 2.2.2 we show that an omniscience-achieving parameter choice satisfying (1.2) exists if and only if the state of the plant (1.1) is detectable from the combined output portions available to each source component³ of \mathcal{G} . We also ascertain that if such a detectability condition holds then there exists an omniscience-achieving solution for which the resulting aggregate dimension of all additional internal (augmented) states satisfies:

$$\sum_{i=1}^{m} \mu_i \le m - m_s \tag{1.3}$$

where m_s is the number of source components⁴ of \mathcal{G} . It follows from our analysis that if there is no omniscience-achieving solution in the proposed class satisfying (1.2), then omniscience cannot be attained by any other scheme – including higher order LTI, nonlinear, and time-varying ones – subject to the same graph.

We apply the distributed estimation scheme to tracking of 4 water buffaloes using animal-borne wireless camera network. In Section 2.5, we present experimental results using a data set collected from the deployment of animal-borne wireless camera network in the Gorongosa National Park (Mozambique)

1.2 Optimal Remote State Estimation of Markov Processes

We study a two-block remote state estimation problem: Suppose that a *sensing unit* accesses the full state \mathbf{x}_k of the process at time k, and decides whether to transmit

³The definition of the source component is given in Definition 2.2.1.

⁴The number of source components of \mathcal{G} ranges from 1 to m.

$$\mathbf{R}_{k} = \begin{cases} 1 & \text{Transmission} \\ 0 & \text{No Transmission} \\ & &$$

Figure 1.2: A framework for remote state estimation with a Markov process (P), a sensing unit (S), and an estimator (E).

 $(\mathbf{R}_k = 1)$ the accessed information \mathbf{x}_k to a *remotely located estimator* or not to transmit $(\mathbf{R}_k = 0)$, where each transmission incurs a positive communication cost c_k . The estimator computes a state estimate $\hat{\mathbf{x}}_k$ that is a causal function of information available to it. The diagram in Figure 1.2 depicts the framework adopted in this work.

Suppose that a transmission policy and an estimation rule at time k, denoted by \mathcal{T}_k and \mathcal{E}_k , respectively, dictate the random variable \mathbf{R}_k and estimate $\hat{\mathbf{x}}_k$ as follows:

$$\mathbf{R}_{k} = \boldsymbol{\mathcal{T}}_{k}\left(\left\{\mathbf{x}_{j}\right\}_{j=0}^{k}, \left\{\mathbf{R}_{j}\right\}_{j=1}^{k-1}\right)$$
(1.4a)

$$\hat{\mathbf{x}}_{k} = \mathcal{E}_{k}\left(\left\{\mathbf{x}_{j} \mid \mathbf{R}_{j} = 1, \ 1 \le j \le k\right\}, \left\{\mathbf{R}_{j}\right\}_{j=1}^{k}\right)$$
(1.4b)

Based on (1.4a) and (1.4b), let us consider a cost functional given as follows:

$$\sum_{k=1}^{N} \mathbb{E}\left[d^{2}\left(\mathbf{x}_{k}, \hat{\mathbf{x}}_{k}\right) + c_{k} \cdot \mathbf{R}_{k} \left| \mathbf{x}_{0} = x_{0}, \{\boldsymbol{\mathcal{T}}_{k}\}_{k=1}^{N}, \{\boldsymbol{\mathcal{E}}_{k}\}_{k=1}^{N}\right]\right]$$
(1.5)

subject to a Markov process $\{\mathbf{x}_k\}_{k=0}^N$ defined on a metric space (\mathbb{X}, d) . The total cost (1.5) consists of the expectation of squared estimation error $d^2(\mathbf{x}_k, \hat{\mathbf{x}}_k)$ and communication costs $c_k \cdot \mathbf{R}_k$.

Our goal is to find optimal transmission policies $\{\boldsymbol{\mathcal{T}}_k\}_{k=1}^N$ and estimation rules

 $\{\mathcal{E}_k\}_{k=1}^N$ for (1.5). To achieve this, we formulate this as a two-player team decision problem, and find optimal decision functions for both players – sensing unit and estimator. To assess optimality of solutions obtained in this work, we adopt the notions of joint optimality and person-by-person optimality: A jointly optimal solution consists of the transmission policies $\{\mathcal{T}_k\}_{k=1}^N$ and estimation rules $\{\mathcal{E}_k\}_{k=1}^N$ that achieve the minimum of (1.5); while a person-by-person optimal solution consists of the transmission policies $\{\mathcal{T}_k\}_{k=1}^N$ and estimation rules $\{\mathcal{E}_k\}_{k=1}^N$ for which given $\{\mathcal{T}_k\}_{k=1}^N$, $\{\mathcal{E}_k\}_{k=1}^N$ minimizes (1.5), and vice versa.

1.2.1 Summary of the Main Contributions

Our main strategy, which is described in Section 3.1, is to divide the aforementioned problem into a set of N sub-problems, and sequentially solve each sub-problem. In Section 3.2, adapting the notions of joint optimality and person-by-person optimality to each sub-problem, we focus on the following contributions for each sub-problem: (i) We show that there exists a jointly optimal solution. As jointly optimal solutions are also person-by-person optimal, this result ensures that the set of person-by-person optimal solutions is non-empty. (ii) We propose an iterative procedure to compute a person-by-person optimal solution. The procedure, which is inspired from Lloyd's algorithm originally used to compute Centroidal Voronoi Tessellations [1–3], alternates between finding a best response (transmission policy) of the sensing unit to a decision function (estimation rule) of the estimator and vice versa, and it generates a sequence of sub-optimal solutions. Our analysis will show that the sequence has a convergent subsequence, and the limit of any

convergent subsequence is a person-by-person optimal solution. In Section 3.3, we describe how to recover an optimal solution to the original problem from optimal solutions of the sub-problems. In Section 3.4, we consider two specific Markov process models – linear system models and self-propelled particle models, and verify that our main results are applicable to these models. Lastly, we apply the remote estimation scheme to tracking of a water buffalo using animal-borne wireless camera network. In Section 3.5, we present experimental results using a data set collected from the deployment of animalborne wireless camera network in the Gorongosa National Park (Mozambique).

1.3 Evolutionary Game Dynamics and Passivity

Of central interest, in evolutionary game theory [4, 5], is the study of dynamics of strategically interacting players in large populations. This line of study involves an investigation of properties of behavioral rules adopted by players and asymptotes of trajectories induced by the rules in an effort to identify stable equilibria. In this work, we conduct the investigation by adapting the notion of passivity originated from dynamical system theory [6, 7].

Consider multiple populations of players engaged in a game in which each player chooses a strategy from a finite set of strategies, and repeatedly revises its strategy choice in response to given payoffs. Evolutionary dynamics describe such strategy revision processes and determine the time-evolution of the population state – the distribution of strategy choices across populations. Assuming that there are infinitely many players in each population, we express evolutionary dynamics with differential equations and regard these

dynamics as dynamical systems. This point of view allows us to define passivity for evolutionary dynamics and to perform stability analysis based on passivity methods adopted from dynamical system theory literature.

The study of evolutionary dynamics and associated stability concepts has been one of active research areas in evolutionary game theory. Brown and von Neumann [8] studied *Brown-von Neumann-Nash (BNN) dynamics* to examine the existence of optimal strategies for a zero-sum two-player game. Taylor and Jonker [9] studied *replicator dynamics* and established a connection between *evolutionarily stable strategies* and stable equilibria of replicator dynamics. Later the result was strengthened by Zeeman [10] who also proposed a stability concept for games under replicator dynamics. Also Gilboa and Matshu [11] considered *cyclic stability* for games under dynamics exhibiting the best response choice.

In succeeding work, rather than working on specific forms of dynamics such as ones considered in [8–11], stability results were established for various classes of evolutionary dynamics. Swinkels [12] considered a class of *myopic adjustment dynamics*, and studied *strategic stability* of equilibria of these dynamics. Ritzberger and Weibull [13] considered a class of *sign-preserving selection dynamics*, and studied asymptotic stability of faces of the population state space. In particular, the authors discovered that a face spanned by a set of pure strategies is stable under sign-preserving selection dynamics if the face is closed under a *better-reply correspondence*.

In a recent development of evolutionary game theory, Hofbauer and Sandholm [14] studied *stable games* and established global stability results for a certain class of evolutionary dynamics, which includes *excess payoff/target (EPT) dynamics, pairwise compar*-

ison dynamics, and *perturbed best response (PBR) dynamics*. Fox and Shamma [15] later revealed that stable games and the aforementioned class of evolutionary dynamics exhibit passivity. Based on an input-output property of passive dynamical systems, the authors established \mathbb{L}_2 -stability of evolutionary dynamics in a class of (generalized) stable games. In addition, applications of evolutionary game theory to study of animal group motion are found in [16, 17], where stable strategy choices in animal pursuit-evasion problems are examined.

Inspired on the passivity analysis in [15], we further investigate passivity in evolutionary game theory. **Our main goals are (i)** to define passivity for evolutionary dynamics that admit realizations in a finite-dimensional state space and present systematic methods to examine passivity of evolutionary dynamics of interest; and **(ii)** to establish stability of passive dynamics in population games.

1.3.1 Summary of the Main Contributions

- We define three notions of passivity (ordinary) passivity, strict passivity, and strict output passivity – and explain how passivity defines stored energy of evolutionary dynamics, which will be used to establish stability of the dynamics. We characterize passivity in terms of vector fields that define state-space realizations of evolutionary dynamics. Based on this characterization, we show that the EPT dynamics, pairwise comparison dynamics, and PBR dynamics are passive; while the replicator dynamics are not.
- 2. We investigate properties of passive evolutionary dynamics in relation to the Nash

stationarity (**NS**) condition and *positive correlation* (**PC**) condition⁵. We first show that for passive dynamics satisfying (**NS**), their equilibrium points coincide with the set of states that achieve the lowest level of stored energy of the dynamics. In addition, if the dynamics also satisfy (**PC**) then we show that these dynamics cannot be strictly output passive.

- 3. We show an equivalence between passivity of evolutionary dynamics and (a weak form of) stability of a closed-loop resulting from a feedback interconnection of evolutionary dynamics and a certain class of payoff operators. This result leads us to re-define passivity of evolutionary dynamics using a class of (generalized) population games. Furthermore, we study the effect of control costs on passivity where we establish a relation between convexity of control costs and passivity of evolutionary dynamics.
- 4. Based on the above contributions, we present stability results for passive evolutionary dynamics in population games. In particular, we consider a class of games that generalizes stable games [14], and show that in this class of games, stored energy of passive dynamics converges to its lowest level. We provide an interpretation of the convergence of stored energy for formerly established dynamics both analytically and by means of numerical simulations.

⁵See (**NS**) and (**PC**) in Section 4.2.3 for their respective definitions.

1.3.2 Stability Concept and Landscape Metaphor

A landscape metaphor from genetics suggests that each individual in populations would move up toward the peak of *fitness landscape*, and would reside unless external force is applied [18, 19]. This metaphor suggests a reasoning over which the state of populations evolves and a "stable equilibrium" emerges.

In this work, we adopt a concept of stability that is similar to the idea suggested by the landscape metaphor: Stability implies that along the trajectory of the population state, stored energy of evolutionary dynamics converges to its lowest level. The convergence to the lowest energy level would have distinct interpretations which are specific to individual evolutionary dynamics. In some cases, the convergence implies that the population state approaches equilibrium points of dynamics; and hence it establishes asymptotic stability of the equilibrium points. As a case in point, in Section 4.3, we will show that for the BNN dynamics and Smith dynamics, the convergence implies that the population state converges to Nash equilibria; and for the logit dynamics, it implies that the population state state converges to the set of best-response strategy distributions.

Chapter 2: Design of Distributed LTI Observers for State Omniscience

2.1 Problem Formulation

2.1.1 Notation

- m is the number of observers that form the distributed observer.
- $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ is a graph¹ formed by a vertex set \mathbb{V} and an edge set $\mathbb{E} \subseteq \mathbb{V} \times \mathbb{V}$.
- I_p is the *p*-dimensional identity matrix.
- $\mathbf{1}_p$ is the *p*-dimensional vector with all entries equal to one.
- \otimes represents Kronecker product of matrices.
- For a set $\{K_1, \dots, K_p\}$ of matrices, we define the following block diagonal matrix:

$$diag(K_1, \cdots, K_p) \stackrel{def}{=} \begin{pmatrix} K_1 & \cdots & \mathbf{0} \\ \vdots & \ddots & \vdots \\ \mathbf{0} & \cdots & K_p \end{pmatrix}$$

Given a set V = {1, · · · , |V|}, W = (w_{ij})_{i,j∈V} is a matrix in R^{|V|×|V|} whose i, j-th entry is w_{ij}.

¹For notational convenience, we assume that every vertex of \mathcal{G} has a self-loop, i.e., $(i, i) \in \mathbb{E}$ for all i in

• For a set $\mathbb{J} = \{j_1, \cdots, j_p\} \subseteq \{1, \cdots, m\}$ and matrices B and C formed by concatenating conformal submatrices $\{B_i\}_{i=1}^m$ and $\{C_i\}_{i=1}^m$ as $B = \begin{pmatrix} B_1 & \cdots & B_m \end{pmatrix}$ and $C = \begin{pmatrix} C_1^T & \cdots & C_m^T \end{pmatrix}^T$, respectively, we define $B_{\mathbb{J}} \stackrel{def}{=} \begin{pmatrix} B_{j_1} & \cdots & B_{j_p} \end{pmatrix}$ and $C_{\mathbb{J}} \stackrel{def}{=} \begin{pmatrix} C_{j_1}^T & \cdots & C_{j_p}^T \end{pmatrix}^T$

2.1.2 Problem Description

We consider that a LTI plant (1.1) and a directed communication graph \mathcal{G} are given. Without loss of generality, we consider that the dynamic matrix A is nondegenerate (see Appendix A.2) and that the output matrices $\{C_i\}_{i=1}^m$ are nonzero. Each vertex i in \mathbb{V} is associated with an observer that accesses $y_i(k) = C_i x(k)$. We adopt the convention that \mathbb{E} includes edge (j, i) if information can be transmitted from observer j to observer i. The neighborhood of observer i, denoted as \mathbb{N}_i , is a subset of \mathbb{V} that contains i and all other vertices with an outgoing edge towards i. Essentially, elements of \mathbb{N}_i represent the observers that can transmit information to observer i.

In this work, we adopt the following parametrized class of distributed observers inspired on [20], where for each i in \mathbb{V} , observer i updates its internal state according to the following state-space equation:

$$\hat{x}_{i}(k+1) = A \sum_{j \in \mathbb{N}_{i}} \mathbf{w}_{ij} \underbrace{\hat{x}_{j}(k)}_{\text{state estimate}} + \mathbf{K}_{i} \underbrace{(y_{i}(k) - C_{i}\hat{x}_{i}(k))}_{\text{measurement innovation}} + \mathbf{P}_{i} \underbrace{z_{i}(k)}_{\text{augmented state}}$$
(2.1)
$$z_{i}(k+1) = \mathbf{Q}_{i} \left(y_{i}(k) - C_{i}\hat{x}_{i}(k) \right) + \mathbf{S}_{i}z_{i}(k)$$

where $\mathbf{w}_{ij} \in \mathbb{R}$, $\mathbf{K}_i \in \mathbb{R}^{n \times r_i}$, $\mathbf{P}_i \in \mathbb{R}^{n \times \mu_i}$, $\mathbf{Q}_i \in \mathbb{R}^{\mu_i \times r_i}$, $\mathbf{S}_i \in \mathbb{R}^{\mu_i \times \mu_i}$ are the design parameters and μ_i is the dimension of the augmented state z_i .² We refer to $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{N}}$

²We use bold font to represent the parameters to be designed. The role of the augmented states in (2.1)

as gain matrices and $\mathbf{W} = (\mathbf{w}_{ij})_{i,j\in\mathbb{V}}$ as a weight matrix³ that must satisfy $\sum_{j\in\mathbb{N}_i} \mathbf{w}_{ij} = 1$ for all i in \mathbb{V} . The update scheme (2.1) complies with \mathcal{G} because the estimate \hat{x}_i of observer i only depends on y_i and the estimates $\{\hat{x}_j\}_{j\in\mathbb{N}_i}$ of the observers in its neighborhood \mathbb{N}_i .

The following Definition of an omniscience-achieving parameter choice will be used throughout the chapter.

Definition 2.1.1 (Omniscience-achieving Parameter Choice). Consider a LTI plant (1.1) with state x and a distributed observer whose state estimates $\{\hat{x}_i\}_{i \in \mathbb{V}}$ are computed according to (2.1). A parameter choice $\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$ and $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$ for (2.1) is referred to as omniscience-achieving if the resulting distributed observer achieves omniscience asymptotically, i.e., $\lim_{k\to\infty} ||\hat{x}_i(k) - x(k)|| = 0$ holds for all i in \mathbb{V} .

The following is the main problem addressed in this work.

Problem 2.1.2. *Given a LTI plant* (1.1) *and a graph G, determine necessary and sufficient conditions for the existence of an omniscience-achieving parameter choice for* (2.1) *that satisfies the scalability condition* (1.2).

2.1.3 Comparative Survey of Related Work

The work in [21, 22] introduced a computationally tractable distributed state estimation scheme for linear plants. The proposed method, so called *Distributed Kalman* is explained in Section 2.2.1.2.

³We assume that $\mathbf{w}_{ij} = 0$ if $j \notin \mathbb{N}_i$ for all i in \mathbb{V} .

Filtering $(DKF)^4$, alternates between an estimation (Kalman filtering) step and a datafusion step that can be viewed as consensus [24].

Results on the performance and stability of the DKF are presented in [25–28]. In particular, the authors of [25] showed non-convexity of performance optimization for a simple system model, e.g., a first-order LTI plant. In [27, 28], stability properties of the DKF are studied when multiple data-fusion steps are allowed between two consecutive estimation steps.

Subsequent work [29–36] investigates similar estimation schemes which have the structure of an estimation-data fusion alternation as in [21, 22]. In [31], the authors performed a stability analysis in terms of the plant model and underlying communication graph to obtain gain parameters for the estimation step; and in [35], these parameters are obtained via optimization of a quadratic estimation cost. Besides, the data-fusion step is realized using weighted averaging [29], diffusion strategies [30], gossip algorithms [32], and internal model average consensus [33].

Other notable approaches to distributed estimation are proposed in [37–41]. The authors of [37] introduced a design method for the DKF which is based on spatial decomposition of the plant and a distributed algorithm for matrix computation. In [38], a distributed estimation algorithm is proposed for plants that consist of *overlapping* subsystems. In addition, a moving horizon estimation scheme was used in [39], and a H_2/H_{∞} optimization framework was adopted in [40,41] for distributed state estimation.

Moreover, in [42,43], the authors establish *necessary* conditions for achieving omniscience in distributed state estimation. These conditions specify observability/detectability

⁴An extensive review of the DKF schemes is found in [23].

requirements in terms of the plant model and underlying communication graph.

To achieve asymptotic omniscience, some of the existing schemes require (i) strong observability conditions [26, 30, 38], (ii) multiple data-fusion steps between two consecutive estimation steps [27,28], which imposes a two-time-scale structure, or (iii) imposition of algebraic constraints on the underlying graph [31,33,36], which are stronger than what is considered in our work.

In contrast to previous work, we propose a class of distributed observers that operate on a single time-scale, and we find necessary and sufficient conditions for the existence of an omniscience-achieving distributed observer in this class for which the scalability condition (1.2) holds. It will follow from our analysis that if asymptotic omniscience cannot be achieved under the proposed scheme then it is not possible under any other scheme – including higher order LTI, nonlinear, and time-varying ones – subject to the same communication graph.

The use of augmented states as in (2.1) was proposed in [20], where we also provided sufficient conditions for the existence of an omniscience-achieving parameter choice. In [44], we developed necessary and sufficient conditions for the existence of an omniscience-achieving parameter choice for the case where W is a pre-selected symmetric matrix. This work *extends* and *unifies* our prior results in the following way: we consider *directed communication graphs*, which allow asymmetric W, and we investigate necessary and sufficient conditions for the existence of an omniscience-achieving parameter choice for the case where W is a pre-selected symmetric matrix. This work *extends* and *unifies* our prior results in the following way: we consider *directed communication graphs*, which allow asymmetric W, and we investigate necessary and sufficient conditions for the existence of an omniscience-achieving parameter choice for (2.1). Unlike the methods proposed in [20] and [44], here we also consider the scalability constraint (1.2).

2.2 Main Result

In this section, we present our solution to Problem 2.1.2, and an example that illustrates it. We start by defining the source component of a graph.

Definition 2.2.1. Given a directed graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$, a strongly connected component $(\mathbb{V}^C, \mathbb{E}^C)$ of \mathcal{G} is said to be a source component⁵ if there is no edge from $\mathbb{V} \setminus \mathbb{V}^C$ to \mathbb{V}^C . Also we define a set of source component representatives⁶ as a subset \mathbb{V}^R of \mathbb{V} that contains exactly one element (representative) from each source component of \mathcal{G} .

The following is our main Theorem.

Theorem 2.2.2. Suppose that the plant is given as in (1.1), that the communication graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ is pre-selected, and that the following hold:

- (i) There are m_s source components⁷ of G which are represented as {(𝒱_l, 𝔅_l)}^{m_s}_{l=1}.
 Each source component G_l = (𝒱_l, 𝔅_l) is associated with a subsystem of the plant described by the pair (A, C_{𝒱_l}).
- (ii) Let \mathbb{V}^R be a set of source component representatives. For each *i* in \mathbb{V}^R , we define ν_i to be the order (number of vertices) of the source component to which vertex *i* belongs.

⁵We adopt the convention that if the graph G is strongly connected then G itself is a (unique) source component.

⁶The choice of \mathbb{V}^R may not be unique.

⁷According to Definition 2.2.1, every graph has at least one source component, i.e., $m_s \ge 1$.



Figure 2.1: A communication graph G for Example 2.2.3.

There is an omniscience-achieving parameter choice for (2.1) that satisfies (1.2) if and only if all the subsystems $\{(A, C_{\mathbb{V}_l})\}_{l=1}^{m_s}$ are detectable. In particular, if such a detectability condition holds then there is a parameter choice for which μ_i is given by

$$\mu_{i} = \begin{cases} \nu_{i} - 1 & \text{if } i \in \mathbb{V}^{R} \\ 0 & \text{otherwise} \end{cases}$$

$$(2.2)$$

for all i *in* \mathbb{V} *.*

The proof is given in Section 2.4. When the conditions of the Theorem are verified, the method outlined in Appendix A.1.2 can be used to compute omniscience-achieving parameters for which (2.2) is satisfied. Notice that because $\sum_{i \in \mathbb{V}^R} \nu_i \leq m$ holds, we can conclude that μ_i given by (2.2) satisfies (1.2). In fact, since \mathbb{V}^R has m_s elements, it also follows that (1.3) holds.

Example 2.2.3. Consider the communication graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ depicted in Figure 2.1 and a LTI plant (1.1) with m = 7. From Definition 2.2.1, we conclude that \mathcal{G}_1 and \mathcal{G}_2 are the source components of \mathcal{G} , and we select $\mathbb{V}^R = \{1, 6\}$. From Theorem 2.2.2, we conclude that there exists an omniscience-achieving parameter choice for which μ_i is given by

$$\mu_{i} = \begin{cases} 2 & \text{if } i = 1 \\ 1 & \text{if } i = 6 \\ 0 & \text{otherwise} \end{cases}$$

if and only if $(A, C_{\mathbb{V}_1})$ and $(A, C_{\mathbb{V}_2})$ are both detectable.

2.2.1 Additional Remarks on the Proposed Class of Distributed Observers

2.2.1.1 The Effect of Noise on the Estimation Performance

Although our formulation focuses on the noiseless case, the fact that the plant and the distributed observer are LTI guarantees graceful degradation with respect to noise in the communication links and/or measurements. In particular, if the noise amplitude is bounded by β then the limit $\max_{i \in \mathbb{V}} \lim_{k \to \infty} ||\hat{x}_i(k) - x(k)||$ may be positive, but one can find an upper bound that scales linearly with β . Also, the effect of noise can be quantified using classical frequency-domain methods.

2.2.1.2 The Role of the Augmented States

As will be discussed in Section 2.4.3, asymptotic omniscience for the proposed class of distributed observers can be cast as the stabilization of certain LTI systems via fully decentralized output feedback. The augmented states in (2.1) are directly related with the internal dynamics of such a decentralized controller which gives us additional freedom in designing the way local state estimates and measurements are fused.

2.2.1.3 Complexity of the Proposed Scheme

We evaluate the complexity of the proposed scheme in terms of the dimensions of the augmented states required to achieve asymptotic omniscience.

For the sake of argument, we compare our method with a simple relay-based centralized scheme described as follows: Suppose that under the same configuration as in Figure 1.1, every observer would transmit its local measurement to its neighbors and, at the same time, would relay local measurements received from neighboring observers in which each transmission/relay incurs a unit time delay. Under this setting, the fixedlag smoothing scheme [45] can be adopted at each observer to determine its update rule for state estimation. Similar to our scheme, the internal state of each observer in the centralized scheme consists of a state estimate and an augmented state to account for the time delay in transmission/relay. However, in what regards to achieving asymptotic omniscience, this centralized scheme would require augmented states whose dimensions would be much larger than our scheme. To see this, we note that in the centralized scheme, the dimension of the augmented state of each observer i is equal to the product of the order of the plant and the maximum length among the respective *shortest paths* from the other vertices to vertex i in the graph \mathcal{G} . In contrast, as stated in Theorem 2.2.2, in our scheme only one observer per source component needs an augmented state, whose dimension equals the order of the source component minus one. As a case in point, suppose that \mathcal{G} is a directed ring, and let n and m be the orders of the plant and graph \mathcal{G} , respectively. Then, for the centralized scheme, the aggregate dimension of all augmented states could be as large as $n \cdot m \cdot (m-1)$; whereas, for the scheme we propose it is no larger than m-1.

2.3 Application to the Synchronization of Coupled Multi-agent Systems

Given a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ with $\mathbb{V} = \{1, \dots, m\}$ and a set $\mathbb{V}^I = \{1, \dots, m_a\}$, consider a LTI multi-agent system whose state-space representation is given as follows:

$$\chi^{(i)}(k+1) = F_o \chi^{(i)}(k) + \sum_{j \in \mathbb{V}^I \setminus \{i\}} F_{ij} \left(\chi^{(j)}(k) - \chi^{(i)}(k) \right) + \sum_{j \in \mathbb{V}} G_{ij} u_j(k), \ i \in \mathbb{V}^I$$

$$y_{i}(k) = H_{i} \begin{pmatrix} \chi^{(1)}(k) \\ \vdots \\ \chi^{(m_{a})}(k) \end{pmatrix}, \ i \in \mathbb{V}$$

$$(2.3b)$$

For each i in \mathbb{V}^I , $\chi^{(i)}(k)$ takes a value in \mathbb{R}^n and represents the state of agent i. For each i in \mathbb{V} , $y_i(k)$ and $u_i(k)$ take values in \mathbb{R}^{r_i} and \mathbb{R}^{p_i} , and represent the output and control input of the system (2.3) associated with vertex i of \mathcal{G} , respectively.

For each i, j in \mathbb{V}^{I} , the matrix F_{ij} in (2.3a) quantifies the coupling between the state $\chi^{(i)}$ of agent i and the state $\chi^{(j)}$ of agent j. For notational convenience, $\mathcal{G}^{I} = (\mathbb{V}^{I}, \mathbb{E}^{I})$ represents the coupling among the states of agents in which, for each i in \mathbb{V}^{I} and j in $\mathbb{V}^{I} \setminus \{i\}$, an edge (j, i) belongs \mathbb{E}^{I} if and only if $F_{ij} \neq \mathbf{0}$ holds. We refer to \mathcal{G}^{I} as the *interaction graph* of the multi-agent system (2.3). We remark that if all the agents are synchronized at time k_0^{8} , i.e., $\chi^{(1)}(k_0) = \cdots = \chi^{(m_a)}(k_0)$, then they remain synchronized and the state of each agent i satisfies

$$\chi^{(i)}(k+1) = F_o \chi^{(i)}(k)$$

⁸In this case, we may assume that $u_i(k) = 0$, $\forall i \in \mathbb{V}$ and $k \ge k_0$, since there is no need to control synchronized agents.

for $k \ge k_0$. The agent model (2.3) is called *homogeneous* because the agents have the same internal dynamics as specified by the dynamic matrix F_o .

The goal is to design a set of controllers for which the agents of the system (2.3) are asymptotically synchronized, i.e., $\lim_{k\to\infty} ||\chi^{(i)}(k) - \chi^{(1)}(k)|| = 0$ holds for all i in $\mathbb{V}^I \setminus \{1\}$. In particular, we suppose that each controller i is represented by vertex i in \mathbb{V} and has the following state-space representation:

$$\xi_i(k+1) = \sum_{j \in \mathbb{N}_i} \mathbf{S}_{ij}^c \xi_j(k) + \mathbf{Q}_i^c y_i(k)$$

$$u_i(k) = \sum_{j \in \mathbb{N}_i} \mathbf{P}_{ij}^c \xi_j(k) + \mathbf{K}_i^c y_i(k)$$
(2.4)

where ξ_i is the internal state of controller *i*, and \mathbb{N}_i is the neighborhood of controller *i* defined by \mathcal{G} , which represents the controllers that can transmit information to controller *i*. We refer to a set of controllers equipped with \mathcal{G} as a *distributed controller*. The diagram in Figure 2.2 depicts the overall system considered here.



Figure 2.2: A diagram showing an example of an overall closed-loop system that consists of a LTI multi-agent system and distributed controller. See Example 2.3.3 for a numerical implementation of the closed-loop system.
We summarize the problem as follows.

Problem 2.3.1. *Given a graph* $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ *and a LTI multi-agent system as in* (2.3)*, we want to*

(i) determine parameters {K^c_i, P^c_{ij}, Q^c_i, S^c_{ij}}_{i,j∈V} for (2.4) such that the interconnection of (2.3) and (2.4) results in the asymptotic synchronization of the system (2.3), i.e.,

$$\lim_{k \to \infty} \left\| \chi^{(i)}(k) - \chi^{(1)}(k) \right\| = 0$$

for all i in $\mathbb{V}^{I} \setminus \{1\}$, and

(ii) show that the state of each agent converges to a solution of $\chi_o(k+1) = F_o\chi_o(k)$ for some initial value $\chi_o(0) \in \mathbb{R}^n$, i.e.,

$$\lim_{k \to \infty} \left\| \chi^{(i)}(k) - \chi_o(k) \right\| = 0$$

for all i in \mathbb{V}^I .

The literature on the problem of designing distributed controllers for synchronization of multi-agent systems is vast (see, for instance, [46–49] and references therein). To mention a few, the work of [50] considered synchronization of linearly coupled nonlinear agents, and the authors of [51] formulate synchronization as mixed-integer nonlinear optimization. Also, there is recent work [52–59] that focused on studying synchronization problems with LTI multi-agent models. The framework in these articles assumes that the states of agents are completely decoupled, and each agent has an associated controller that accesses its full state and has full control of it.

Here, we consider a LTI multi-agent system in which (i) agents are interacting according to (2.3a), (ii) for each j in \mathbb{V} , the j-th control input $u_j(k)$ affects the state of the system according to $\{G_{ij}\}_{i \in \mathbb{V}^I}$, and (iii) for each i in \mathbb{V} , the i-th output vector $y_i(k)$ depends on the state of the system according to H_i . Due to (i)-(iii), the formulation considered in Problem 2.3.1 may not be cast as one to which existing results for completely decoupled multi-agent models can be applied. More specifically, suppose that each agent has an associated controller that accesses its full state and has full control of it. To transform the agent model (2.3a) into a completely decoupled one, each controller needs to access the states of the agents on which the state of its associated agent depends, and generate control to cancel the coupling. However, this may not be possible since the interaction graph \mathcal{G}^I and graph \mathcal{G} , whose edges determine the viability and direction of information transfer among controllers, may not be identical as depicted in Figure 2.2.

The proposed scheme can be applied to frequency synchronization in power grids [60–62], which ensures stable operation of grids and efficient power transfer from generators to loads.

Our solution to Problem 2.3.1 is given as follows.

Proposition 2.3.2. Suppose that a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ is pre-selected, that a LTI multiagent system is given as in (2.3), and that \mathcal{G} and the matrices in (2.5) satisfy the following:

- *(i)* The pair (A', B') is stabilizable, where A' and B' are defined in (2.5b) and (2.5c), respectively.
- (ii) The graph G and the pair (A, C) satisfy the detectability condition of Theorem 2.2.2, where A and C are defined in (2.5a) and (2.5d), respectively.

There exists a distributed controller (2.4) that asymptotically synchronizes the system (2.3), i.e.,

$$\lim_{k \to \infty} \left\| \chi^{(i)}(k) - \chi^{(1)}(k) \right\| = 0$$

for all i in $\mathbb{V}^I \setminus \{1\}$. Furthermore, if all eigenvalues of F_o lie on or inside the unit circle in \mathbb{C} , then the state of each agent converges to a solution of

$$\chi_o(k+1) = F_o\chi_o(k)$$

for some initial value $\chi_o(0) \in \mathbb{R}^n$, i.e.,

$$\lim_{k \to \infty} \left\| \chi^{(i)}(k) - \chi_o(k) \right\| = 0$$

for all i in \mathbb{V}^{I} .

$$A = \left(\begin{array}{c|c|c} F_{o} & F_{12} & \cdots & F_{1m_{a}} \\ \hline 0 & A' \end{array} \right)$$
with $A' = \left(\begin{array}{c|c|c} A'_{1,1} & \cdots & A'_{1,m_{a}-1} \\ \vdots & \ddots & \vdots \\ A'_{m_{a}-1,1} & \cdots & A'_{m_{a}-1,m_{a}-1} \end{array} \right)$
and $A'_{i-1,j-1} = \begin{cases} F_{ij} - F_{1j} & \text{if } i \neq j \\ F_{o} - \sum_{l \in \mathbb{V}^{I} \setminus \{i\}} F_{il} - F_{1i} & \text{if } i = j \end{cases}$

$$B' = \left(\begin{array}{c} B'_{1} & \cdots & B'_{m} \end{array} \right) \text{ with } B'_{i} = \left((G_{2i} - G_{1i})^{T} & \cdots & (G_{m_{a}i} - G_{1i})^{T} \right)^{T}, \quad (2.5c)$$

$$C = \left(\begin{array}{c} C_{1}^{T} & \cdots & C_{m}^{T} \end{array} \right)^{T} \text{ with } C_{i} = H_{i} \left(\begin{pmatrix} 1 & 0 \\ 1_{m_{a}-1} & I_{m_{a}-1} \end{pmatrix} \otimes I_{n} \right) \quad (2.5d)$$

A constructive proof of Proposition 2.3.2 is given in Appendix A.3, where we use Theorem 2.2.2 to show the existence of $\{\mathbf{K}_{i}^{c}, \mathbf{P}_{ij}^{c}, \mathbf{Q}_{i}^{c}, \mathbf{S}_{ij}^{c}\}_{i,j \in \mathbb{V}}$ for (2.4) for which the interconnection of (2.3) and (2.4) results in the asymptotic synchronization of the system (2.3).

Example 2.3.3 (Numerical Example). *Consider a multi-agent system* (2.3) *and the graph G depicted in Figure 2.2, where the matrices in* (2.3) *are numerically specified as follows:*

$$F_{o} = \begin{pmatrix} 0.9950 & 0.0998 \\ -0.0998 & 0.9950 \end{pmatrix}$$
$$F_{ij} = \begin{cases} -0.1I_{2} & \text{if } (i,j) \in \{(1,3), (2,1), (3,2), (4,1), (4,3)\} \\ \mathbf{0} & \text{otherwise} \end{cases}$$

$$\begin{pmatrix} G_{11} & G_{12} & G_{13} \end{pmatrix} = \begin{pmatrix} 1 & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \end{pmatrix}$$

$$\begin{pmatrix} G_{21} & G_{22} & G_{23} \end{pmatrix} = \begin{pmatrix} -1 & 1 & 0 & 1 \\ 0 & 0 & 0 & 0 \end{pmatrix}$$

$$\begin{pmatrix} G_{31} & G_{32} & G_{33} \end{pmatrix} = \begin{pmatrix} 0 & -1 & 1 & 0 \\ 0 & 0 & 0 & 0 \end{pmatrix}$$

$$\begin{pmatrix} G_{41} & G_{42} & G_{43} \end{pmatrix} = \begin{pmatrix} 0 & 0 & -1 & -1 \\ 0 & 0 & 0 & 0 \end{pmatrix}$$

$$(H_{11} H_{12} H_{13} H_{14}) = \left(\begin{array}{cc|c} 1 & 0 & -1 & 0 & 0 & 0 & 0 \end{array}\right)$$
$$(H_{21} H_{22} H_{23} H_{24}) = \left(\begin{array}{cc|c} 0 & 0 & 1 & 0 & -1 & 0 & 0 & 0 \end{array}\right)$$
$$(H_{31} H_{32} H_{33} H_{34}) = \left(\begin{array}{cc|c} 1 & 0 & 0 & 0 & -1 & 0 & 0 & 0 \end{array}\right)$$

The assumptions (i) and (ii) of Proposition 2.3.2 are satisfied; hence, the existence of a distributed controller (2.4) that synchronizes the system (2.3) is guaranteed. We compute a parameter choice for (2.4) according to the procedure described in Appendix A.3.1 and Appendix A.3.2. The state trajectories of the resulting closed-loop system are depicted in Figure 2.3.



Figure 2.3: A simulation result of Example 2.3.3 which depicts the synchronization of the first components $\chi_1^{(1)}(k), \chi_1^{(2)}(k), \chi_1^{(3)}(k), \chi_1^{(4)}(k)$ of the states of agent 1, 2, 3, 4.

Remark 2.3.4. Since our results can be applied to any interaction graph \mathcal{G}^{I} , the assumption (ii) of Proposition 2.3.2 may be stronger than what would be needed for the cases in which the agents are completely decoupled (cf. Assumption 1 in [58]). As a case in point, consider a system configuration with the same number of agents and controllers and for which the agents are all decoupled, i.e., $\mathbb{V}^{I} = \mathbb{V}$ and $\mathbb{E}^{I} = \bigcup_{i \in \mathbb{V}^{I}} (i, i)$. In addition, assume that the input and output matrices of (2.3), respectively, satisfy $G_{ij} = \mathbf{0}$ if $i \neq j$ for all i in \mathbb{V}^{I} and j in \mathbb{V} , and $H_{i} = e_{i}^{T} \otimes H'_{i}$ for all i in \mathbb{V} and for a matrix H'_{i} in $\mathbb{R}^{r_{i} \times n}$, where e_{i} is the i-th column of the m_{a} -dimensional identity matrix. Under this setting, (ii) of Proposition 2.3.2 requires the graph \mathcal{G} to be strongly connected, while Assumption 1 in [58] only requires \mathcal{G} to have a directed spanning tree.

2.4 Proof of the Main Theorem

In this section, we provide a proof for Theorem 2.2.2. If the conditions of the Theorem hold then Appendix A.1.2 outlines a randomized method to obtain a choice of omniscience-achieving parameters.

The proof has two parts: It starts with Lemma 2.4.3, Lemma 2.4.4, and Theorem 2.4.5 that describe important spectral properties of a parametrized class of weight matrices W. The second part, which consists of Proposition 2.4.8, Theorem 2.4.10, and Remark 2.4.11, determines conditions for a parameter choice W and $\{K_i, P_i, Q_i, S_i\}_{i \in \mathbb{V}}$ to be omniscience-achieving. The structure of the proof is outlined in the diagram of Figure 2.4. Conditions for a parameter choice

 \mathbf{W} and $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$



Figure 2.4: A precedence diagram for the proof of Theorem 2.2.2

2.4.1 Key Results on Weighted Laplacian Matrices

Definition 2.4.1. Consider a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ with $\mathbb{V} = \{1, \dots, |\mathbb{V}|\}$. A matrix $L = (l_{ij})_{i,j \in \mathbb{V}}$ is said to be a Weighted Laplacian Matrix (WLM) of \mathcal{G} if the following three

conditions hold:

- (i) If $(j,i) \notin \mathbb{E}$ then $l_{ij} = 0$ for all i in \mathbb{V} and j in $\mathbb{V} \setminus \{i\}$.
- (ii) If $(j,i) \in \mathbb{E}$ then $l_{ij} < 0$ for all i in \mathbb{V} and j in $\mathbb{V} \setminus \{i\}$.
- (iii) It holds that $\sum_{j=1}^{|\mathbb{V}|} l_{ij} = 0$ for all i in \mathbb{V} .

For notational convenience, we define the set of WLMs of \mathcal{G} as follows:

$$\mathbb{L}(\mathcal{G}) \stackrel{\text{def}}{=} \left\{ L \in \mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|} \mid L \text{ is a WLM of } \mathcal{G} \right\}$$

Definition 2.4.2 (UEPP). Given square matrices A and B, $A \otimes B$ is said to have the so called Unique Eigenvalue Product Property (UEPP) if every nonzero eigenvalue λ of

 $A \otimes B$ can be uniquely expressed⁹ as a product $\lambda = \lambda_A \cdot \lambda_B$, where λ_A and λ_B are eigenvalues of A and B, respectively.

Lemma 2.4.3. Let $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ with $\mathbb{V} = \{1, \dots, |\mathbb{V}|\}$ be a directed graph, and A be a matrix in $\mathbb{R}^{n \times n}$. Suppose that a matrix \mathbf{W} in $\mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|}$ is defined as $\mathbf{W} = I_{|\mathbb{V}|} - \alpha L$, where α is a positive real number and $L = (l_{ij})_{i,j \in \mathbb{V}}$ is a WLM of \mathcal{G} . Given L and α' satisfying $0 < \alpha' \leq (\max_{1 \le i \le |\mathbb{V}|} l_{ii})^{-1}$ 10, for almost every α in $(0, \alpha')$, \mathbf{W} is a stochastic matrix and $\mathbf{W} \otimes A$ satisfies the UEPP.

Lemma 2.4.4. Let a matrix \mathbf{W} in $\mathbb{R}^{m \times m}$ and a matrix A in $\mathbb{R}^{n \times n}$ be given. If all eigenvalues of \mathbf{W} are simple¹¹ and $\mathbf{W} \otimes A$ satisfies the UEPP, then each eigenvector q of $\mathbf{W} \otimes A$ associated with a nonzero eigenvalue λ can be written as a Kronecker product $q = v \otimes p$, where v and p are, respectively, eigenvectors of \mathbf{W} and A (associated with the eigenvalues $\lambda_{\mathbf{W}}$ and λ_A for which $\lambda = \lambda_{\mathbf{W}} \cdot \lambda_A$ holds).

The proofs of Lemmas 2.4.3 and 2.4.4 are given in Appendix A.4.

Theorem 2.4.5. Let $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ be a strongly connected graph. For almost every element *L* of the set $\mathbb{L}(\mathcal{G})$, the following hold:

(i) Every right and left eigenvectors of L have no zero entries.

(*ii*) Every eigenvalue of L is simple.

⁹For an eigenvalue λ of $A \otimes B$, let λ_A , λ'_A be the eigenvalues of A and λ_B , λ'_B be the eigenvalues of B for which $\lambda = \lambda_A \cdot \lambda_B = \lambda'_A \cdot \lambda'_B$ holds. The eigenvalue λ is said to be uniquely expressed as a product $\lambda = \lambda_A \cdot \lambda_B$ if it holds that $\lambda_A = \lambda'_A$ and $\lambda_B = \lambda'_B$.

¹⁰If $l_{ii} = 0$ for all i in \mathbb{V} , then we consider that $\left(\max_{1 \le i \le |\mathbb{V}|} l_{ii}\right)^{-1} = \infty$.

¹¹An eigenvalue of a matrix is simple if both the geometric and algebraic multiplicities of the eigenvalue are equal to 1.

Since Theorem 2.4.5 hinges on structured linear system theory, in Appendix A.5 we provide a review of key concepts followed by a proof.

2.4.2 A Brief Introduction to Stabilization via Decentralized Control

We review certain classical results in decentralized control that will be used in the proof of Theorem 2.2.2. Of special relevance are the fundamental work of [63–66] that investigates the notion of fixed modes¹² for LTI plants, and the work of [68] that studies the effect of decentralized output feedback on LTI plants. To introduce these results, we consider the following state-space representation for a LTI plant:

$$\widetilde{x}(k+1) = \widetilde{A}\widetilde{x}(k) + \sum_{i=1}^{|\mathbb{V}|} \widetilde{B}_i \widetilde{u}_i(k)$$

$$\widetilde{y}_i(k) = \widetilde{C}_i \widetilde{x}(k)$$
(2.6)

for each i in $\mathbb{V} = \{1, \dots, |\mathbb{V}|\}$, where $\widetilde{x}(k) \in \mathbb{R}^{\widetilde{n}}, \widetilde{u}_i(k) \in \mathbb{R}^{\widetilde{p}_i}$, and $\widetilde{y}_i(k) \in \mathbb{R}^{\widetilde{r}_i}$ are the state, *i*-th control input, and *i*-th output, respectively.

Definition 2.4.6. [63, 64] A given $\lambda \in \mathbb{C}$ is a fixed mode of (2.6) if it is an eigenvalue of $\widetilde{A} + \sum_{i=1}^{|\mathbb{V}|} \widetilde{B}_i K_i \widetilde{C}_i$ for all K_i in $\mathbb{R}^{\widetilde{p}_i \times \widetilde{r}_i}$.

Remark 2.4.7. The fixed mode is an eigenvalue of the plant (2.6) which is invariant under the decentralized output feedback $\tilde{u}_i(k) = K_i \tilde{y}_i(k)$ for all i in \mathbb{V} , where K_i is a matrix in $\mathbb{R}^{\tilde{p}_i \times \tilde{r}_i}$. In addition, if the plant (2.6) has an unstable fixed mode then it cannot be stabilized by any decentralized controller that is causal and LTI (see [63, 64] for the details).

¹²The notion of fixed modes is analogous to the concept of uncontrollable or unobservable modes adopted in centralized control problems [67].

The fixed modes can be characterized by an algebraic rank test described in the following Proposition.

Proposition 2.4.8. [65, 66] Consider that a LTI plant is given as in (2.6). Let

$$\widetilde{B} = \begin{pmatrix} \widetilde{B}_1 & \cdots & \widetilde{B}_{|\mathbb{V}|} \end{pmatrix}$$
 and $\widetilde{C} = \begin{pmatrix} \widetilde{C}_1^T & \cdots & \widetilde{C}_{|\mathbb{V}|}^T \end{pmatrix}^T$

A given $\lambda \in \mathbb{C}$ is a fixed mode of the plant if and only if there exists a subset $\mathbb{J} \subseteq \mathbb{V}$ for which

$$rank \begin{pmatrix} \widetilde{A} - \lambda \cdot I_{\widetilde{n}} & \widetilde{B}_{\mathbb{J}} \\ \\ \widetilde{C}_{\mathbb{J}^{c}} & \mathbf{0} \end{pmatrix} < \widetilde{n}$$

$$(2.7)$$

holds, where \widetilde{n} is the dimension of \widetilde{A} , and $\mathbb{J}^c = \mathbb{V} \setminus \mathbb{J}$.

Definition 2.4.9. Let $(\mathbb{V}, \mathbb{E}^P)$ be a graph of a LTI plant (2.6) in which the edge set \mathbb{E}^P satisfies $(j, i) \in \mathbb{E}^P$ if and only if $\widetilde{C}_i \left(z \cdot I_{\widetilde{n}} - \widetilde{A} \right)^{-1} \widetilde{B}_j$ is nonzero. The plant (2.6) is said to be strongly connected if its graph $(\mathbb{V}, \mathbb{E}^P)$ is strongly connected.

In the following Theorem, based on Theorem 4 of [68], we specify the effect of decentralized output feedback of the following form on a strongly connected LTI plant.

$$z_{1}(k+1) = S_{1}z_{1}(k) + Q_{1}\tilde{y}_{1}(k)$$

$$\tilde{u}_{1}(k) = P_{1}z_{1}(k) + K_{1}\tilde{y}_{1}(k)$$
(2.8a)

$$\widetilde{u}_i(k) = K_i \widetilde{y}_i(k), \ i \in \mathbb{V} \setminus \{1\}$$
(2.8b)

where $z_1(k)$ takes a value in \mathbb{R}^{μ_1} for a nonnegative integer μ_1 .

Theorem 2.4.10. *Consider a LTI plant given as in* (2.6) *and decentralized output feedback* (2.8). *Suppose that the plant is strongly connected and has no unstable fixed mode.* Then, for almost every choice of $\{K_i\}_{i \in \mathbb{V} \setminus \{1\}}$, there exists a choice of K_1 , P_1 , Q_1 , S_1 for which the closed-loop system obtained from the interconnection of (2.6) and (2.8) described by

$$\begin{pmatrix} \widetilde{x}(k+1) \\ z_1(k+1) \end{pmatrix} = \begin{pmatrix} \widetilde{A} + \sum_{i=1}^{|\mathbb{V}|} \widetilde{B}_i K_i \widetilde{C}_i & \widetilde{B}_1 P_1 \\ Q_1 \widetilde{C}_1 & S_1 \end{pmatrix} \begin{pmatrix} \widetilde{x}(k) \\ z_1(k) \end{pmatrix}$$
(2.9)

is stable.

Remark 2.4.11. *The system* (2.9) *also can be viewed as the closed-loop system obtained by applying a (centralized) controller described by* (2.8a) *to a LTI system described by the triple*

$$\left(\widetilde{A} + \sum_{i=2}^{|\mathbb{V}|} \widetilde{B}_i K_i \widetilde{C}_i, \ \widetilde{B}_1, \ \widetilde{C}_1\right)$$
(2.10)

We can find a parameter choice K_1 , P_1 , Q_1 , S_1 for which (2.9) is stable using results on finding stabilizing (centralized) controllers for LTI systems. In particular, by adopting the result of [69], we can find a stabilizing controller (2.8a) of order μ_1 equal to the controllability index of (2.10) minus one.

2.4.3 Additional Preliminary Results

Let m_s be the number of source components of \mathcal{G} in which we denote each source component as $\mathcal{G}_l = (\mathbb{V}_l, \mathbb{E}_l)$, and \mathbb{V}^R be the set of source component representatives (see Definition 2.2.1). Notice that the source components of \mathcal{G} impose the following structure on W:

$$\mathbf{W} = \begin{pmatrix} \mathbf{W}_{1} & \cdots & \mathbf{0} & \mathbf{0} \\ \vdots & \ddots & \vdots & \vdots \\ \mathbf{0} & \cdots & \mathbf{W}_{m_{s}} & \mathbf{0} \\ \mathbf{W}_{m_{s}+1,1} & \cdots & \mathbf{W}_{m_{s}+1,m_{s}} & \mathbf{W}_{m_{s}+1,m_{s}+1} \end{pmatrix}$$
(2.11)

For each l in $\{1, \dots, m_s\}$, the sparsity pattern of $\mathbf{W}_l \in \mathbb{R}^{|\mathbb{V}_l| \times |\mathbb{V}_l|}$ must be consistent¹³ with \mathcal{G}_l so that under a suitable choice of $\{\mathbf{W}_{m_s+1,l}\}_{l=1}^{m_s+1}$, the sparsity pattern of \mathbf{W} given above is consistent with \mathcal{G} .

For notational convenience, we consider that $\mathbb{V}_l = \{1, \dots, |\mathbb{V}_l|\}$ and $\mathbb{V}_l \cap \mathbb{V}^R = \{1\}$. To analyze the asymptotic omniscience of the proposed estimation scheme, under the parameter choice of \mathbf{W} and $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$ described in Appendix A.1.2, we derive the state-space representation for the dynamics for the estimation error of (2.1) associated with \mathcal{G}_l as follows:

$$\begin{pmatrix} \widetilde{x}(k+1) \\ z_1(k+1) \end{pmatrix} = \begin{pmatrix} \mathbf{W}_l \otimes A - \sum_{i=1}^{|\mathbb{V}_l|} \overline{B}_i \mathbf{K}_i \overline{C}_i & -\overline{B}_1 \mathbf{P}_1 \\ \mathbf{Q}_1 \overline{C}_1 & \mathbf{S}_1 \end{pmatrix} \begin{pmatrix} \widetilde{x}(k) \\ z_1(k) \end{pmatrix}$$
(2.12)

where $\widetilde{x} = \left(\widetilde{x}_{1}^{T} \dots \widetilde{x}_{|\mathbb{V}_{l}|}^{T}\right)^{T}$ with $\widetilde{x}_{i} = x - \hat{x}_{i}$, and \mathbf{W}_{l} is a submatrix of \mathbf{W} associated with \mathcal{G}_{l} as in (2.11). For each i in \mathbb{V}_{l} , $\overline{B}_{i} = e_{i} \otimes I_{n}$ and $\overline{C}_{i} = e_{i}^{T} \otimes C_{i}$ where e_{i} is the i-th column of the $|\mathbb{V}_{l}|$ -dimensional identity matrix. Notice that (2.12) can be viewed as the state-space representation of the closed-loop system obtained by applying decentralized output feedback, parametrized by \mathbf{K}_{1} , \mathbf{P}_{1} , \mathbf{Q}_{1} , \mathbf{S}_{1} , $\{\mathbf{K}_{i}\}_{i \in \mathbb{V}_{l} \setminus \{1\}}$, to a LTI system described

¹³The sparsity pattern of a matrix $\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$ is consistent with a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ if $\mathbf{w}_{ij} = 0$ for $(j,i) \notin \mathbb{E}$.

by the triple

$$\begin{pmatrix} \mathbf{W}_{l} \otimes A, -\left(\overline{B}_{1} & \cdots & \overline{B}_{|\mathbb{V}_{l}|}\right), \\ \vdots \\ \overline{C}_{|\mathbb{V}_{l}|} \end{pmatrix} \end{pmatrix}$$
(2.13)

Hence, we can write (2.12) as in (2.9) by selecting $\widetilde{A} = \mathbf{W}_l \otimes A$, $P_1 = \mathbf{P}_1$, $Q_1 = \mathbf{Q}_1$, $S_1 = \mathbf{S}_1$ and $\widetilde{B}_i = -\overline{B}_i$, $\widetilde{C}_i = \overline{C}_i$, $K_i = \mathbf{K}_i$ for all i in \mathbb{V}_l . This idea, in conjunction with Theorem 2.4.10, allows us to connect the stability of the estimation error dynamics (2.12) with strong connectivity of (2.13) and the absence of unstable fixed modes in (2.13).

The following Lemma states certain spectral properties of W determined by Procedure 3 in Appendix A.1.2.1. The proof of Theorem 2.2.2 is then followed.

Lemma 2.4.12. The submatrices $\{\mathbf{W}_l\}_{l=1}^{m_s}$ and \mathbf{W}_{m_s+1,m_s+1} of \mathbf{W} in (2.11) satisfy the following with probability one:

- (P1) For each l in $\{1, \dots, m_s\}$, every right and left eigenvectors of \mathbf{W}_l have no zero entries.
- (P2) For each l in $\{1, \dots, m_s\}$, every eigenvector q of $\mathbf{W}_l \otimes A$ associated with an unstable eigenvalue λ can be written as a Kronecker product $q = v \otimes p$, where v and p are, respectively, eigenvectors of \mathbf{W}_l and A (associated with the eigenvalue $\lambda_{\mathbf{W}_l}$ and the unstable eigenvalue λ_A for which $\lambda = \lambda_{\mathbf{W}_l} \cdot \lambda_A$ holds).
- (P3) Every eigenvalue of $\mathbf{W}_{m_s+1,m_s+1} \otimes A$ is zero.

Proof. Notice that for each l in $\{1, \dots, m_s\}$, in Procedure 3 (Line 3-10), we have set $\mathbf{W}_l = I_{|\mathbb{V}_l|} - \alpha L$ where α is chosen according to a uniform distribution defined on (0, 1), and $L = (l_{ij})_{i,j \in \mathbb{V}_l}$ is a WLM of \mathcal{G}_l , each of its nonzero off-diagonal entries l_{ij} is chosen according to a uniform distribution defined on $\left(-\frac{1}{|\mathbb{N}_i|-1}, 0\right)$ independent of choices of other entries. By Theorem 2.4.5, L satisfies (i) of Theorem 2.4.5 which ensures that (P1) holds with probability one.

In addition, according to Lemma 2.4.3 and Theorem 2.4.5, this choice of α and L ensures that \mathbf{W}_l is a stochastic matrix and has all simple eigenvalues, and $\mathbf{W}_l \otimes A$ satisfies the UEPP (see Definition 2.4.2) with probability one. Since \mathbf{W}_l is stochastic, its eigenvalues lie on or inside the unit circle in \mathbb{C} ; hence, an unstable eigenvalue λ of $\mathbf{W}_l \otimes A$ can be written as $\lambda = \lambda_{\mathbf{W}_l} \cdot \lambda_A$ where $\lambda_{\mathbf{W}_l}$ is an eigenvalue of \mathbf{W}_l and λ_A is an unstable eigenvalue of A. Therefore, invoking Lemma 2.4.4, we conclude that (**P2**) holds with probability one.

Lastly, the way entries of \mathbf{W}_{m_s+1,m_s+1} are chosen by Procedure 3 (Line 11-14) ensures that all eigenvalues of \mathbf{W}_{m_s+1,m_s+1} are zero and (**P3**) holds.

2.4.4 Proof of Theorem 2.2.2

First of all notice that if the subsystem $(A, C_{\mathbb{V}_l})$ of the plant (1.1) is not detectable, then for any choice of \mathbb{W}_l , the system (2.13) has an unstable fixed mode. By Remark 2.4.7, there is no parameter choice for which the estimation error dynamics (2.12) is stable; hence, no omniscience-achieving parameter choice exists for (2.1). This proves the necessity of Theorem 2.2.2.

Let $\mathbb{V}_{m_s+1} = \mathbb{V} \setminus \bigcup_{l=1}^{m_s} \mathbb{V}_l$. Consider that $\{\mathbf{W}_{m_s+1,l}\}_{l=1}^{m_s+1}$ and $\{\mathbf{K}_i\}_{i \in \mathbb{V}_{m_s+1}}, \{\mu_i\}_{i \in \mathbb{V}_{m_s+1}}$ are determined by Procedure 3 (Line 11-14) and Procedure 4 (Line 9-11) in Appendix A.1.2, respectively. Notice that by (**P3**) of Lemma 2.4.12 and due to the choice of $\{\mathbf{K}_i\}_{i \in \mathbb{V}_{m_s+1}}$, $\{\mu_i\}_{i \in \mathbb{V}_{m_s+1}}$, to prove the sufficiency of Theorem 2.2.2, we only need to show that for each l in $\{1, \dots, m_s\}$, under the detectability condition of Theorem 2.2.2, there exists a choice of \mathbf{W}_l and $\{\mathbf{K}_i\}_{i \in \mathbb{V}_l}$, \mathbf{P}_1 , \mathbf{Q}_1 , \mathbf{S}_1 for which the estimation error dynamics (2.12) is stable.¹⁴

Suppose that the choice of \mathbf{W}_l and $\{\mathbf{K}_i\}_{i \in \mathbb{V}_l \setminus \{1\}}$, determined by Procedure 3 (Line 3-10) and Procedure 4 (Line 4-7), respectively, ensures that, with probability one, the LTI system (2.13) is (i) strongly connected and has (ii) no unstable fixed mode, and (iii) the controllability index of the LTI system described by the triple

$$\left(\mathbf{W}_{l} \otimes A - \sum_{i=2}^{|\mathbb{V}_{l}|} \overline{B}_{i} \mathbf{K}_{i} \overline{C}_{i}, \ -\overline{B}_{1}, \ \overline{C}_{1}\right)$$
(2.14)

is equal to $|\mathbb{V}_l|$. By Theorem 2.4.10 and Remark 2.4.11, there exist matrices \mathbf{K}_1 , \mathbf{P}_1 , \mathbf{Q}_1 , \mathbf{S}_1 with $\mu_1 = |\mathbb{V}_l| - 1$ that, in conjunction with the chosen \mathbf{W}_l and $\{\mathbf{K}_i\}_{i \in \mathbb{V}_l \setminus \{1\}}$, ensure that the estimation error dynamics (2.12) is stable, where these matrices can be determined by Procedure 4 (Line 8). Hence, we conclude that the detectability condition is sufficient for the existence of an omniscience-achieving parameter choice, and for the parameter choice determined by Procedure 3 and Procedure 4 to be omniscience-achieving with probability one. It remains to prove the arguments (i)-(iii).

Proof of (i): Suppose that the transfer function matrix given by

$$\overline{C}_{i}\left(z \cdot I_{|\mathbb{V}_{l}| \cdot n} - \mathbf{W}_{l} \otimes A\right)^{-1} \overline{B}_{j}$$
(2.15)

is zero for some i, j in \mathbb{V}_l , or equivalently $\overline{C}_i (\mathbf{W}_l \otimes A)^k \overline{B}_j = \mathbf{0}$ holds for all nonnegative

¹⁴From the overall estimation error dynamics for (2.1), it can be verified that if (2.12) is stable for every l in $\{1, \dots, m_s\}$, then it holds that $\lim_{k\to\infty} ||\hat{x}_i(k) - x(k)|| = 0$ for all $i \in \mathbb{V}_{m_s+1}$.

integer k. This yields that

$$\overline{C}_{i} \left(\mathbf{W}_{l} \otimes A \right)^{k} \overline{B}_{j} = \left(e_{i}^{T} \mathbf{W}_{l}^{k} e_{j} \right) C_{i} A^{k} = \mathbf{0}$$
(2.16)

where we use the fact that $\overline{B}_j = e_j \otimes I_n$ and $\overline{C}_i = e_i^T \otimes C_i$. Since \mathcal{G}_l is strongly connected, due to the choice of \mathbf{W}_l by Procedure 3 (Line 3-10), we can see that $e_i^T \mathbf{W}_l^{k_0} e_j \neq 0$ for a positive integer k_0 , and hence $C_i A^{k_0} = \mathbf{0}$ holds. However, this contradicts the fact that Ais nondegenerate and C_i is nonzero (see Section 2.1.2). Therefore, the transfer function matrix (2.15) is nonzero for all i, j in \mathbb{V}_l which, by definition, implies that the system (2.13) is strongly connected with probability one.

Proof of (ii): Let us define
$$\overline{B} = \left(\overline{B}_1 \quad \dots \quad \overline{B}_{|\mathbb{V}_l|}\right)$$
 and $\overline{C} = \left(\overline{C}_1^T \quad \dots \quad \overline{C}_{|\mathbb{V}_l|}^T\right)^T$.

According to Proposition 2.4.8, we need to show that the following inequality holds for every unstable eigenvalue λ of $\mathbf{W}_l \otimes A$:

$$rank \begin{pmatrix} \mathbf{W}_{l} \otimes A - \lambda \cdot I_{|\mathbb{V}_{l}| \cdot n} & \overline{B}_{\mathbb{J}} \\ \overline{C}_{\mathbb{J}^{c}} & \mathbf{0} \end{pmatrix} \ge |\mathbb{V}_{l}| \cdot n$$
(2.17)

where \mathbb{J} is an arbitrary subset of \mathbb{V}_l , and $\mathbb{J}^c = \mathbb{V}_l \setminus \mathbb{J}$ is its complement.

Suppose that \mathbb{J} is not empty then by (**P1**), (**P2**) of Lemma 2.4.12 and by the definition of \overline{B} , it holds that

$$rank \begin{pmatrix} \mathbf{W}_{l} \otimes A - \lambda \cdot I_{|\mathbb{V}_{l}| \cdot n} & \overline{B}_{\mathbb{J}} \\ \overline{C}_{\mathbb{J}^{c}} & \mathbf{0} \end{pmatrix}$$
$$\geq rank \begin{pmatrix} \mathbf{W}_{l} \otimes A - \lambda \cdot I_{|\mathbb{V}_{l}| \cdot n} & \overline{B}_{\mathbb{J}} \end{pmatrix} = |\mathbb{V}_{l}| \cdot n \qquad (2.18)$$

Otherwise, since $\mathbb{J}^c = \mathbb{V}_l$, by (P1), (P2) of Lemma 2.4.12, by the definition of \overline{C} , and by

the detectability of the subsystem $(A, C_{\mathbb{V}_l})$, it holds that

$$rank \begin{pmatrix} \mathbf{W}_{l} \otimes A - \lambda \cdot I_{|\mathbb{V}_{l}| \cdot n} & \overline{B}_{\mathbb{J}} \\ \overline{C}_{\mathbb{J}^{c}} & \mathbf{0} \end{pmatrix}$$
$$= rank \begin{pmatrix} \mathbf{W}_{l} \otimes A - \lambda \cdot I_{|\mathbb{V}_{l}| \cdot n} \\ \overline{C} \end{pmatrix} = |\mathbb{V}_{l}| \cdot n \qquad (2.19)$$

Therefore, from (2.18) and (2.19), we can observe that the inequality in (2.17) holds for every unstable eigenvalue λ of $\mathbf{W}_l \otimes A$, and by Proposition 2.4.8 the system (2.13) has no unstable fixed mode with probability one.

Proof of (iii): To verify this, we consider a matrix given by

$$\left(\overline{B}_{1} \quad (\mathbf{W}_{l} \otimes A) \,\overline{B}_{1} \quad \cdots \quad (\mathbf{W}_{l} \otimes A)^{|\mathbb{V}_{l}|-1} \,\overline{B}_{1}\right) \tag{2.20}$$

Note that (2.20) can be rewritten as

$$\left(\begin{pmatrix} e_1 & \mathbf{W}_l e_1 & \cdots & \mathbf{W}_l^{|\mathbb{V}_l|-1} e_1 \end{pmatrix} \otimes I_n \end{pmatrix} \cdot diag \left(I_n, A, \cdots, A^{|\mathbb{V}_l|-1} \right)$$
(2.21)

where we use the fact that $\overline{B}_1 = e_1 \otimes I_n$.

By the nondegeneracy of A, the rank of the matrix in (2.20) equals

$$rank\left(e_1 \quad \mathbf{W}_l e_1 \quad \cdots \quad \mathbf{W}_l^{|\mathbb{V}_l|-1} e_1\right) \cdot n$$

and by (**P1**) of Lemma 2.4.12, we can see that the matrix in (2.20) has rank $|\mathbb{V}_l| \cdot n$. Hence, the following matrix has *generic* rank $|\mathbb{V}_l| \cdot n$, i.e., for almost every choice of $\{\mathbf{K}_i\}_{i \in \mathbb{V}_l \setminus \{1\}}$, the matrix has rank $|\mathbb{V}_l| \cdot n$.

$$\left(\overline{B}_{1} \quad \left(\mathbf{W}_{l} \otimes A - \sum_{i=2}^{|\mathbb{V}_{l}|} \overline{B}_{i} \mathbf{K}_{i} \overline{C}_{i}\right) \overline{B}_{1} \quad \cdots \quad \left(\mathbf{W}_{l} \otimes A - \sum_{i=2}^{|\mathbb{V}_{l}|} \overline{B}_{i} \mathbf{K}_{i} \overline{C}_{i}\right)^{|\mathbb{V}_{l}|-1} \overline{B}_{1}\right)$$

$$(2.22)$$

Therefore, due to the choice of $\{\mathbf{K}_i\}_{i \in \mathbb{V}_l \setminus \{1\}}$ by Procedure 4 (Line 4-7), the controllability index of (2.14) is equal to $|\mathbb{V}_l|$ with probability one.

2.5 Application to Tracking of Animal Groups and Experimental Results

In this section, we apply the proposed distributed estimation scheme to tracking of animal groups, and show preliminary experimental results using a data set collected from the deployment of *animal-borne wireless camera network* in the Gorongosa National Park (Mozambique) in August 2015.¹⁵ The main purpose of the development and deployment of the system was to collect biologically meaningful measurements and videos using GPS, IMU, and Camera all integrated in a single tracking device, where the proposed estimation scheme can be used to determine how to fuse sensor measurements and location estimates of tracking devices within the network so that each tracking device in the network learns locations of all other devices connected to the same communication network (see Figure 2.5). The sensor measurements and videos are used to study animal group motion. During the deployment, 15 tracking devices were installed on waterbucks and water buffaloes. Figure 2.6 and Figure 2.7 show GPS tracks of 4 water buffaloes (Buffalo 1, 2, 3, 4).

¹⁵The development and deployment of animal-borne wireless camera network were performed under a research grant NSF ECCS 1135726.

Disclaimer: The author of this dissertation was NOT involved in the deployment in the Gorongosa National Park.



Figure 2.5: System overview of the tracking device in the animal-borne wireless camera network



Figure 2.6: A screenshot of GPS tracks of water buffaloes in the Google earth

(Timespan: 2015-08-06T00:00:00Z \sim 2015-08-06T04:00:00Z)



Figure 2.7: Original trajectories of water buffaloes in a local North East Up (NEU) coordinate system (The origin of the coordinate system: Latitude = -18.9279877268328, Longitude = 34.5457567343343)

To represent animal group motion, we adopt a continuous-time LTI model described by the following state-space equation: For each i in $\{1, \dots, 4\}$,

$$\dot{p}_x^{(i)}(t) = v_x^{(i)}(t)$$
 (2.23a)

$$\dot{p}_{y}^{(i)}(t) = v_{y}^{(i)}(t)$$
 (2.23b)

$$\dot{v}_x^{(i)}(t) = -\sum_{j=1}^4 a_{ij} \left(v_x^{(i)}(t) - v_x^{(j)}(t) \right)$$
(2.23c)

$$\dot{v}_{y}^{(i)}(t) = -\sum_{j=1}^{4} a_{ij} \left(v_{y}^{(i)}(t) - v_{y}^{(j)}(t) \right)$$
(2.23d)

where $\left(p_x^{(i)}(t) \quad p_y^{(i)}(t)\right)^T$ and $\left(v_x^{(i)}(t) \quad v_y^{(i)}(t)\right)^T$ denote the location and velocity of Buffalo *i*, respectively. We have assumed that $a_{ij} = a_{ji}$ for all *i*, *j* in $\{1, \dots, 4\}$, and that a node (observer) is associated with each water buffalo and the location and velocity measurements of each buffalo are available to its associated node every 10 seconds. By discretizing (2.23), we obtain a discrete-time LTI model described as follows:

$$x(k+1) = Ax(k) \tag{2.24a}$$

$$y_i(k) = C_i x(k) \tag{2.24b}$$

for i in $\{1, \cdots, 4\}$, where

$$x(k) = \begin{pmatrix} p_x^{(1)}(k) & p_y^{(1)}(k) & v_x^{(1)}(k) & v_y^{(1)}(k) & \cdots & p_x^{(4)}(k) & p_y^{(4)}(k) & v_x^{(4)}(k) & v_y^{(4)}(k) \end{pmatrix}^T$$
$$y_i(k) = \begin{pmatrix} p_x^{(i)}(k) & p_y^{(i)}(k) & v_x^{(i)}(k) & v_y^{(i)}(k) \end{pmatrix}^T$$

and the system matrices A and C_i are determined as follows:

$$A = \exp\left(10A_c\right)$$
$$C_i = e_i^T \otimes I_4$$

where

$$A_{C} = \begin{pmatrix} 0 & 1 & 0 & 0 & 0 & 0 & 0 & 0 \\ 0 & -\sum_{j \neq 1} a_{1j} & 0 & a_{12} & 0 & a_{13} & 0 & a_{14} \\ 0 & 0 & 0 & 1 & 0 & 0 & 0 & 0 \\ 0 & a_{21} & 0 & -\sum_{j \neq 2} a_{2j} & 0 & a_{23} & 0 & a_{24} \\ 0 & 0 & 0 & 0 & 0 & 1 & 0 & 0 \\ 0 & a_{31} & 0 & a_{32} & 0 & -\sum_{j \neq 3} a_{3j} & 0 & a_{34} \\ 0 & 0 & 0 & 0 & 0 & 0 & 0 & 1 \\ 0 & a_{41} & 0 & a_{42} & 0 & a_{43} & 0 & -\sum_{j \neq 4} a_{4j} \end{pmatrix} \otimes I_{2}$$

and e_i is the *i*-column of I_4 . Note that, according to C_i , each node has access to the location and velocity of its associated buffalo.

In practice, the model (2.24) may include a noise term w(k) which is due to the un-modeled dynamics of animal motion:

$$x(k+1) = Ax(k) + w(k)$$
$$y_i(k) = C_i x(k)$$

To minimize the noise w(k), we have chosen the entries a_{ij} of A_C that minimize the cost given by

$$\sum_{k=0}^{N-1} \|w(k)\|^2$$

where N is the number of available location and velocity measurements, and the resulting



Figure 2.8: A communication graph \mathcal{G} for designing a distributed observer.

choice of a_{ij} is given by

$$a_{12} = a_{21} = 0.002$$

 $a_{13} = a_{31} = 0.002$
 $a_{14} = a_{41} = 0$
 $a_{23} = a_{32} = 0.001$
 $a_{24} = a_{42} = 0$
 $a_{34} = a_{43} = 0.003$

To design a distributed observer, we assume that the communication graph \mathcal{G} is preselected as depicted in Figure 2.8. We find the omniscience-achieving parameter based on Procedure 3 and Procedure 4 described in Appendix A.1.2. Figure 2.9 shows the estimate $\left(\hat{p}_x^{(i)}(k) \ \hat{p}_y^{(i)}(k)\right)^T$ of the original trajectory of each buffalo, depicted in Figure 2.7, by every node; and Figure 2.10 shows normalized estimation error computed by

$$\frac{1}{\text{total traveled distance of Buffalo }i} \cdot \left\| \begin{pmatrix} p_x^{(i)}(k) \\ p_y^{(i)}(k) \end{pmatrix} - \begin{pmatrix} \hat{p}_x^{(i)}(k) \\ \hat{p}_y^{(i)}(k) \end{pmatrix} \right\|$$



Figure 2.9: Estimated trajectories of water buffaloes using the proposed distributed estimation scheme in a local NEU coordinate system



Figure 2.10: Normalized estimation error at every node (Total traveled distance:

Buffalo 1 = 1694 m, Buffalo 2 = 1505.8 m, Buffalo 3 = 1981.8 m, Buffalo 4 = 2129 m)

2.6 Summary and Future Work

We described a parametrized class of LTI distributed observers for state estimation of a LTI plant, where the information exchange among the members of a distributed observer is constrained by a pre-selected communication graph. We developed necessary and sufficient conditions for the existence of a parameter choice for a distributed observer that ensures asymptotic omniscience and satisfies the scalability constraint (1.2). These conditions can be described by the detectability of the subsystems of the plant that are associated with the source components of the graph.

As a future direction, we suggest performance analysis of the proposed scheme and parameter optimization to minimize estimation error in the presence of noise in measurement and communication link. Also it is important to consider distributed state estimation over time-varying communication graphs.

Chapter 3: Optimal Remote State Estimation of Markov Processes

3.1 Problem Formulation

3.1.1 Notation and Terminologies

- For elements a_1, \dots, a_N of X, we define $a_{1:N} \stackrel{def}{=} (a_1, \dots, a_N)$.
- For functions $\mathcal{A}_1, \cdots, \mathcal{A}_N$, we define $\mathcal{A}_{1:N} \stackrel{def}{=} (\mathcal{A}_1, \cdots, \mathcal{A}_N)$.
- When the random variable R_k is dictated by a policy T_k, we use same superscript for R_k and T_k, e.g., R^{*}_k and T^{*}_k, or R⁽ⁱ⁾_k and T⁽ⁱ⁾_k.
- We define

$$\boldsymbol{\tau}_{k} \stackrel{def}{=} \begin{cases} \max\left\{ j \in \{1, \cdots, k-1\} \middle| \mathbf{R}_{j} = 1 \right\} & \text{if } \mathbf{R}_{j} = 1 \text{ for some } j \in \{1, \cdots, k-1\} \\ 0 & \text{otherwise} \end{cases}$$

The value of τ_k indicates the most recent time when a transmission has occurred from the sensing unit to the estimator. We refer to τ_k as the *last transmission time* before time k.

3.1.2 Problem Description

In this section, we describe the problem formulation considered throughout the work in which we seek transmission policies $\mathcal{T}_{1:N}$ and (state) estimation rules $\mathcal{E}_{1:N}$ that dictate decision making of the sensing unit and estimator, respectively, and that are optimal for the cost functional (1.5). Throughout the work, without loss of optimality, we consider that transmission policies and estimation rules have the following structures¹: The transmission policy at time k depends only on the last transmission time τ_k , the information \mathbf{x}_{τ_k} transmitted to the estimator at time τ_k , and the current state \mathbf{x}_k of the process. In particular, we adopt a class of *randomized* transmission policies.² The estimation rule at time k depends only on the last transmission policies. The sensing unit at time τ_k . Given a transmission policy \mathcal{T}_k and an estimation rule \mathcal{E}_k , the decision variables \mathbf{R}_k and $\hat{\mathbf{x}}_k$ are dictated by \mathcal{T}_k and \mathcal{E}_k , respectively, as follows:

$$\mathbf{R}_{k} = \boldsymbol{\mathcal{T}}_{k} \left(\boldsymbol{\tau}_{k}, \mathbf{x}_{\boldsymbol{\tau}_{k}}, \mathbf{x}_{k} \right)$$
(3.1a)
$$\hat{\mathbf{x}}_{k} = \begin{cases} \mathcal{E}_{k} \left(\boldsymbol{\tau}_{k}, \mathbf{x}_{\boldsymbol{\tau}_{k}} \right) & \text{if } \mathbf{R}_{k} = 0 \\ \mathbf{x}_{k} & \text{otherwise} \end{cases}$$
(3.1b)

¹We do not lose optimality of resulting solutions from the imposition of these structures. This can be

verified by similar arguments as in Lemma 1 and Lemma 3 of [70].

²See Appendix B.2 for a detailed description of randomized policies.

We formally state our main problem as follows.

Problem 3.1.1. Given a Markov process $\{\mathbf{x}_k\}_{k=0}^N$, find optimal transmission policies $\mathcal{T}_{1:N}$ and estimation rules $\mathcal{E}_{1:N}$ for the cost functional given by

$$\mathcal{J}(x_0, \mathcal{T}_{1:N}, \mathcal{E}_{1:N}) = \sum_{k=1}^{N} \mathbb{E}\left[d^2(\mathbf{x}_k, \hat{\mathbf{x}}_k) + c_k \cdot \mathbf{R}_k \,\middle|\, \mathbf{x}_0 = x_0, \mathcal{T}_{1:N}, \mathcal{E}_{1:N}\right]$$
(3.2)

subject to (3.1).³

We consider two notions of optimality for solutions of Problem 3.1.1 described as follows.

Definition 3.1.2. Transmission policies $\mathcal{T}_{1:N}^*$ and estimation rules $\mathcal{E}_{1:N}^*$ are said to be jointly optimal for (3.2) if they achieve the global minimum for every x_0 in \mathbb{X} .

Definition 3.1.3. Transmission policies $\mathcal{T}_{1:N}^*$ and estimation rules $\mathcal{E}_{1:N}^*$ are said to be person-by-person optimal for (3.2) if the following relations hold for every x_0 in \mathbb{X} :

$$\mathcal{J}(x_0, \mathcal{T}^*_{1:N}, \mathcal{E}^*_{1:N}) = \min_{\mathcal{T}_{1:N}} \mathcal{J}(x_0, \mathcal{T}_{1:N}, \mathcal{E}^*_{1:N})$$
$$= \min_{\mathcal{E}_{1:N}} \mathcal{J}(x_0, \mathcal{T}^*_{1:N}, \mathcal{E}_{1:N})$$
(3.3)

Equation (3.3) implies that given decision functions $\mathcal{T}_{1:N}^*$ of one player (sensing unit), $\mathcal{E}_{1:N}^*$ are the best decision functions of the other player (estimator), and vice versa.

We maintain the following assumptions throughout the work.

³The initial condition $\mathbf{x}_0 = x_0$ and the process model is common knowledge to both the sensing unit and estimator.

Assumption 3.1.4. (\mathbb{X}, d) is a complete, separable, proper metric space.⁴

Assumption 3.1.5. Let $p_k : \mathbb{X} \times \mathfrak{B} \to \mathbb{R}$ be a transition probability of the process, where \mathfrak{B} is the Borel σ -algebra generated by the metric topology associated with (\mathbb{X}, d) . We assume that the following are true:

- 1. For every non-empty open set \mathbb{O} in \mathfrak{B} , the function $x \mapsto p_k(x, \mathbb{O})$ is positive for all x in \mathbb{X} .
- 2. For each \mathbb{A} in \mathfrak{B} , the function $x \mapsto p_k(x, \mathbb{A})$ is continuous.

Assumption 3.1.6. For each j in $\{k - 1, \dots, N\}$, k in $\{1, \dots, N\}$, and x_{k-1} in \mathbb{X} , we assume that there is a transformation $M_j(k - 1, x_{k-1}, \cdot) : \mathbb{X} \to \mathbb{X}$ for which

- 1. It holds that $M_{k-1}(k-1, x_{k-1}, x_{k-1}) = 0$.
- 2. The function $x_j \mapsto M_j (k-1, x_{k-1}, x_j)$ is continuous and has a continuous inverse. We denote the inverse by $M_j^{-1} (k-1, x_{k-1}, \cdot)$.
- 3. For every x_{j-1} in \mathbb{X} and \mathbb{A} in \mathfrak{B} , the transition probability p_j satisfies

$$p_j\left(x'_{j-1}, \mathbb{A}'\right) = p_j\left(x_{j-1}, \mathbb{A}\right)$$

where

$$x'_{j-1} = M_{j-1}(k-1, x_{k-1}, x_{j-1})$$
$$\mathbb{A}' = M_j (k-1, x_{k-1}, \mathbb{A})$$

⁴For notional convenience, we suppose that $0 \in \mathbb{X}$.

4. The metric d is invariant under M_j , i.e., it holds that

$$d(x_j, \hat{x}_j) = d(M_j(k-1, x_{k-1}, x_j), M_j(k-1, x_{k-1}, \hat{x}_j))$$

for all x_j, \hat{x}_j in \mathbb{X} .

To find a solution to Problem 3.1.1, we divide the problem into a set of N subproblems, and sequentially solve each sub-problem. We proceed by describing the socalled *Two-Player Optimal Stopping Problem* from time k, and show how each subproblem can be related to the optimal stopping problem.

Problem 3.1.7. Suppose that a Markov process $\{\mathbf{x}_j\}_{j=k-1}^N$ with a transition probability $p_j : \mathbb{X} \times \mathfrak{B} \to \mathbb{R}$ and positive constants $\{c'_j\}_{j=k}^N$ are given. Find optimal policies $\mathcal{T}_{k:N}^{< k-1>}$ and rules $\mathcal{E}_{k:N}^{< k-1>}$ for the cost functional given by

$$\mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^2\left(\mathbf{x}_j, \hat{\mathbf{x}}_j\right) + c'_{\mathbf{K}} \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1>}, \mathcal{E}_{k:N}^{< k-1>}\right]$$
(3.4)

where

$$\mathbf{K} = \begin{cases} \min\left\{j \in \{k, \cdots, N\} \middle| \mathbf{R}_j = 1\right\} & \text{if } \mathbf{R}_j = 1 \text{ for some } j \in \{k, \cdots, N\} \\ N & \text{otherwise} \end{cases}$$

The policy $\mathcal{T}_{j}^{\langle k-1 \rangle} : \mathbb{X} \times \mathbb{X} \to \{0,1\}$ and the rule $\mathcal{E}_{j}^{\langle k-1 \rangle} : \mathbb{X} \to \mathbb{X}$, respectively, dictate \mathbf{R}_{j} and $\hat{\mathbf{x}}_{j}$ as follows:

$$\mathbf{R}_{j} = \boldsymbol{\mathcal{T}}_{j}^{\langle k-1 \rangle} \left(x_{k-1}, \mathbf{x}_{j} \right)$$
(3.5a)

$$\hat{\mathbf{x}}_{j} = \begin{cases} \mathcal{E}_{j}^{\langle k-1 \rangle} \left(x_{k-1} \right) & \text{if } \mathbf{R}_{j} = 0 \\ \mathbf{x}_{j} & \text{otherwise} \end{cases}$$
(3.5b)

We adopt two notions of optimality for Problem 3.1.7 as follows.

Definition 3.1.8. Policies $\mathcal{T}_{k:N}^{* < k-1>}$ and rules $\mathcal{E}_{k:N}^{* < k-1>}$ are said to be jointly optimal for (3.4) if they achieve the global minimum for every x_{k-1} in \mathbb{X} .

Definition 3.1.9. Policies $\mathcal{T}_{k:N}^{* < k-1>}$ and rules $\mathcal{E}_{k:N}^{* < k-1>}$ are said to be person-by-person optimal for (3.4) if the following relations hold for every x_{k-1} in \mathbb{X} :

$$\mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}' \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{* < k-1 >}, \mathcal{E}_{k:N}^{* < k-1 >} \right]$$

$$= \min_{\mathcal{T}_{k:N}^{< k-1 >}} \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}' \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1 >}, \mathcal{E}_{k:N}^{* < k-1 >} \right]$$

$$= \min_{\mathcal{E}_{k:N}^{< k-1 >}} \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}' \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{* < k-1 >}, \mathcal{E}_{k:N}^{< k-1 >} \right]$$
(3.6)

Problem 3.1.7 can be viewed as a team decision problem [71] in which two players are involved and the main objective is to find optimal decision functions $\mathcal{T}_{k:N}^{\langle k-1 \rangle}$ and $\mathcal{E}_{k:N}^{\langle k-1 \rangle}$ for the players. Note that the total expected cost (3.4) consists of a *running cost* $d^2(\mathbf{x}_j, \hat{\mathbf{x}}_j)$ and a *stopping cost* c'_j .

Remark 3.1.10. In Section 3.2, we show that using the transformation described in Assumption 3.1.6, the value of (3.4) evaluated at an optimal solution does not depend on the initial condition $\mathbf{x}_{k-1} = x_{k-1}$ (see Remark 3.2.2).

Remark 3.1.11. For any polices $\mathcal{T}_{k:N}^{\langle k-1 \rangle}$ and rules $\mathcal{E}_{k:N}^{\langle k-1 \rangle}$, if $c'_j \leq c''_j$ holds for all j in

 $\{k, \dots, N\}$, then we have the following inequality:

$$\mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}' \cdot \mathbf{R}_{\mathbf{K}} \left| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1>}, \mathcal{E}_{k:N}^{< k-1>} \right] \\ \leq \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}'' \cdot \mathbf{R}_{\mathbf{K}} \left| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1>}, \mathcal{E}_{k:N}^{< k-1>} \right]$$
(3.7)

Associated with Problem 3.1.7, we describe the k-th sub-problem of Problem 3.1.1 as follows.

Sub-problem k: Given $\{\mathcal{T}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$ and $\{\mathcal{E}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$, let us define constants $\{c'_{j}\}_{j=k}^{N}$ as follows:

$$c'_{j} = c_{j} + \mathbb{E}\left[\sum_{l=j+1}^{\mathbf{K}_{j}} d^{2}\left(\mathbf{x}_{l}, \hat{\mathbf{x}}_{l}\right) + c'_{\mathbf{K}_{j}} \cdot \mathbf{R}_{\mathbf{K}_{j}} \left| \mathbf{x}_{j} = 0, \boldsymbol{\mathcal{T}}_{j+1:N}^{}, \mathcal{E}_{j+1:N}^{} \right]$$
(3.8)

with $c'_N = c_N$, where c_j is the communication cost at time j given as in (3.2), and

$$\mathbf{K}_{j} = \begin{cases} \min\left\{l \in \{j+1, \cdots, N\} \middle| \mathbf{R}_{l} = 1\right\} & \text{if } \mathbf{R}_{l} = 1 \text{ for some } l \in \{j+1, \cdots, N\} \\ N & \text{otherwise} \end{cases}$$

With the stopping costs $\{c'_j\}_{j=k}^N$ determined by (3.8), find a solution $\mathcal{T}_{k:N}^{\langle k-1 \rangle}$ and $\mathcal{E}_{k:N}^{\langle k-1 \rangle}$ to Problem 3.1.7.

Note that Sub-problem k assumes that $\{\mathcal{T}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$ and $\{\mathcal{E}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$ are given parameters.

Our main strategy in solving Problem 3.1.1 can be described as follows: We solve the Sub-problems backward in time starting from k = N, where for each Sub-problem k, we provide solutions $\{\mathcal{T}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$ and $\{\mathcal{E}_{j:N}^{< j-1>}\}_{j=k+1}^{N}$ for preceding Sub-problems
Remote Estimation Problem



Figure 3.1: The problem solving strategy for Problem 3.1.1

to construct the stopping costs $\{c'_j\}_{j=k}^N$. Once solutions to all the Sub-problems are found, we determine transmission policies $\mathcal{T}_{1:N}$ and estimation rules $\mathcal{E}_{1:N}$ for Problem 3.1.1 in the following way:

$$\boldsymbol{\mathcal{T}}_{j}(k-1, x_{k-1}, x_{j}) = \boldsymbol{\mathcal{T}}_{j}^{< k-1>}(x_{k-1}, x_{j})$$
(3.9a)

$$\mathcal{E}_j(k-1, x_{k-1}) = \mathcal{E}_j^{< k-1>}(x_{k-1})$$
 (3.9b)

for each j in $\{k, \dots, N\}$ and k in $\{1, \dots, N\}$.

In Section 3.2, we solve Sub-problem k. In particular, we show that there exists a jointly optimal solution and describe an iterative procedure for finding a person-byperson optimal solution. In Section 3.3, based on the results of Section 3.2, we verify that the transmission policies and estimation rules determined by (3.9) are a solution to Problem 3.1.1. The diagram in Figure 3.1 depicts the aforementioned problem solving strategy.

3.1.3 Comparative Survey of Related Work

The effect of communication costs in remote state estimation problems was studied in [70, 72–78]. Finite time-horizon problem formulations are consider in [70, 74, 76, 78]. The authors of [74] found a jointly optimal solution for first-order linear processes driven by Gaussian noise where it is shown that the transmission policy for jointly optimal solutions is of threshold-type. An iterative procedure for finding a transmission policy and estimation rule for first-order linear processes is proposed in [70]. The authors performed a convergence analysis on the proposed procedure for first-order linear processes driven by Gaussian noise, which essentially provides an alternative proof of the main result of [74]. The work of [76] considered a problem setting where the sensing unit has energy harvesting capability. The authors showed that the transmission policy for jointly optimal solutions is of threshold-type for a certain class of multi-dimensional linear processes. Preliminary results of our work were presented in [78] for linear processes.

Infinite time-horizon formulations are considered in [72, 73, 75, 77]. The authors of [72] studied the structure of optimal transmission policies for linear processes driven by Gaussian noise, and proposed a procedure to compute an optimal policy based on a value iteration algorithm. In [73], an algorithm for finding a sub-optimal solution was proposed.

For linear processes driven by Gaussian noise, the authors showed that the cost incurred by the proposed algorithm is within a constant factor of the optimum. While the question of whether the transmission policy for jointly optimal solutions is of threshold-type for multi-dimensional linear processes remains unanswered, the authors of [75] analyzed the performance of threshold-type transmission policies for such processes. A computationally efficient method for finding a sub-optimal transmission policy based on polynomial approximation is proposed in [77].

Other interesting remote estimation schemes are reported in [79–92]. The authors of [79] studied the structure of optimal transmission policies and estimation rules for the case where a finite number of transmissions is allowed to the sensing unit. The authors of [80] considered a problem setting where the sensing unit operates with two different sensing qualities, and found an optimal *time-periodic* transmission policies for linear processes driven by Gaussian noise. A remote estimation problem for continuous dynamical systems are studied in [85] where performance of various types of transmission policies is investigated. Results of [85] indicate that for remote estimation under a communication rate constraint, the transmission policy for jointly optimal solutions may not be of threshold-type. A framework in which the sensing unit observes noisy outputs of the process and transmits observed noisy outputs to the estimator is adopted in [83, 84, 87]. On the other hand, a framework in which the sensing unit accesses noisy observations of the state of the process and transmits its best state estimate to the estimator is adopted in [81,82,89]. The authors of [92] adopted a certain class of stochastic transmission policies which ensures that linear estimation rules are optimal. The authors of [86] proposed an approximate state estimation scheme based on a sum of Gaussians approach. Remote estimation over shared communication networks is considered in [90]. A problem of scheduling transmission power level for remote estimation was recently studied in [91].

The problem formulation considered in this work is technically different from previous ones found in literature in following ways:

- 1. We adopt random process models that may neither be linear nor have unimodal or symmetric probability distributions.
- 2. We consider classes of transmission policies and estimation rules on which no structural assumption is imposed under which the optimality of resulting solutions is lost.
- 3. We investigate optimization of the given performance criteria over both transmission policies and estimation rules.

3.2 Two-Player Optimal Stopping Problem

In this section, we find a solution to Sub-problem k where the constants $\{c'_j\}_{j=k}^N$ are determined by (3.8) using solutions $\{\mathcal{T}_{j:N}^{\leq j-1>}\}_{j=k+1}^N$ and $\{\mathcal{E}_{j:N}^{\leq j-1>}\}_{j=k+1}^N$ to preceding sub-problems – Sub-problem k + 1 to Sub-problem N. We consider two notions of optimality – joint optimality and person-by-person optimality. Our main results state the existence of a jointly optimal solution (Section 3.2.2) and describe an iterative procedure to find a person-by-person optimal solution (Section 3.2.3).

We proceed by re-writing (3.4) into a suitable form using the following Definition.

Definition 3.2.1. *Define a (random) function* $\mathcal{P}_j : \mathbb{X} \to \{0, 1\}$ *and a variable* $\hat{x}_j \in \mathbb{X}$ *for*

each j in $\{k, \cdots, N\}$ as follows:⁵

$$\boldsymbol{\mathcal{P}}_{j}\left(x_{j}\right) = \boldsymbol{\mathcal{T}}_{j}^{< k-1>}\left(0, x_{j}\right) \tag{3.10a}$$

$$\hat{x}_j = \mathcal{E}_j^{< k-1>}(0)$$
 (3.10b)

We refer to \mathcal{P}_j and \hat{x}_j as the (randomized) policy⁶ and estimate at time j (for the initial condition $\mathbf{x}_{k-1} = 0$), respectively.

Given that $\mathbf{x}_{k-1} = 0$, we can re-write (3.4) as follows:

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}, \hat{x}_{k:N}\right)\right]$$
(3.11)

where for each j in $\{k, \dots, N\}$,

$$J_{j}\left(\mathbf{x}_{j}, \boldsymbol{\mathcal{P}}_{j:N}, \hat{x}_{j:N}\right)$$

$$= \left(d^{2}\left(\mathbf{x}_{j}, \hat{x}_{j}\right) + \mathbb{E}_{\mathbf{x}_{j+1}}\left[J_{j+1}\left(\mathbf{x}_{j+1}, \boldsymbol{\mathcal{P}}_{j+1:N}, \hat{x}_{j+1:N}\right) \mid \mathbf{x}_{j}\right]\right) \cdot (1 - \mathbf{R}_{j}) + c'_{j} \cdot \mathbf{R}_{j}$$

$$(3.12)$$

with $J_{N+1} = 0$, and $\boldsymbol{\mathcal{P}}_j$ dictates \mathbf{R}_j as follows:

$$\mathbf{R}_{j} = \boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right) \tag{3.13}$$

Note that J_j satisfies

$$\mathbb{E}_{\mathbf{x}_{j}} \left[J_{j} \left(\mathbf{x}_{j}, \boldsymbol{\mathcal{P}}_{j:N}, \hat{x}_{j:N} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right]$$

$$= \left(\mathbb{E}_{\mathbf{x}_{j}} \left[d^{2} \left(\mathbf{x}_{j}, \hat{x}_{j} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right] \right.$$

$$+ \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1} \left(\mathbf{x}_{j+1}, \boldsymbol{\mathcal{P}}_{j+1:N}, \hat{x}_{j+1:N} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right] \right)$$

$$\cdot \mathbb{P} \left(\mathbf{R}_{j} = 0 \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right) + c'_{j} \cdot \mathbb{P} \left(\mathbf{R}_{j} = 1 \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)$$
(3.14)

 5^{5} As k is fixed in Sub-problem k, throughout the section, we drop the dependence of policies and estimates on k.

⁶See Appendix B.2 for a detailed description of randomized policies.

for all j in $\{k, \dots, N\}$.

Remark 3.2.2. Let $\mathcal{P}_{k:N}^*$ and $\hat{x}_{k:N}^*$ be optimal policies and estimates for (3.11), respectively. From (3.14) and our main results, we can see that

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}^{*}, \hat{x}_{k:N}^{*}\right)\right] = \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}^{*}\left(\mathbf{x}_{k}, \hat{x}_{k:N}^{*}\right)\right]$$
(3.15)

holds where

$$J_{j}^{*}\left(x_{j}, \hat{x}_{j:N}^{*}\right) = \min\left\{d^{2}\left(x_{j}, \hat{x}_{j}^{*}\right) + \mathbb{E}_{\mathbf{x}_{j+1}}\left[J_{j+1}^{*}\left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{*}\right) \middle| \mathbf{x}_{j} = x_{j}\right], c_{j}'\right\} (3.16)$$

with $J_{N+1}^* = 0$.

For any other initial condition $\mathbf{x}_{k-1} = x_{k-1}$, define estimates

$$\hat{x}_{j}^{\prime*} = M_{j}^{-1} \left(k - 1, x_{k-1}, \hat{x}_{j}^{*} \right)$$
(3.17)

for each j in $\{k, \dots, N\}$, where M_j is the transformation described in Assumption 3.1.6. Notice that from (3.16) and by the definition of the transformation M_j , we can observe that

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}^{*}\left(\mathbf{x}_{k},\hat{x}_{k:N}^{*}\right) \mid \mathbf{x}_{k-1}=0\right] = \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}^{*}\left(\mathbf{x}_{k},\hat{x}_{k:N}^{*}\right) \mid \mathbf{x}_{k-1}=x_{k-1}\right]$$

This implies that the value of (3.4) evaluated at an optimal solution does not depend on the initial condition, and by finding an optimal solution to (3.11), we can derive a solution to Sub-problem k using the following relation:

$$\mathcal{T}_{j}^{\langle k-1 \rangle}(x_{k-1}, x_{j}) = \mathcal{P}_{j}^{*}(M_{j}(k-1, x_{k-1}, x_{j}))$$
(3.18a)

$$\mathcal{E}_{j}^{\langle k-1 \rangle}(x_{k-1}) = M_{j}^{-1}\left(k-1, x_{k-1}, \hat{x}_{j}^{*}\right)$$
(3.18b)

Based on Remark 3.2.2, to solve Sub-problem k, we will find optimal policies and estimates for the initial condition $\mathbf{x}_{k-1} = 0$, and derive a solution to Sub-problem k using (3.18).

3.2.1 Definitions and Preliminary Results

We restate Definition 3.1.8 and Definition 3.1.9 as follows.

Definition 3.2.3. Policies $\mathcal{P}_{k:N}^*$ and estimates $\hat{x}_{k:N}^*$ are said to be a jointly optimal solution for (3.11) if they achieve the global minimum.

Definition 3.2.4. Policies $\mathcal{P}_{k:N}^*$ and estimates $\hat{x}_{k:N}^*$ are said to be a person-by-person optimal solution for (3.11) if the following relations hold:

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}^{*}, \hat{x}_{k:N}^{*}\right)\right] = \min_{\boldsymbol{\mathcal{P}}_{k:N}} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}, \hat{x}_{k:N}^{*}\right)\right]$$
$$= \min_{\hat{x}_{k:N}} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}^{*}, \hat{x}_{k:N}\right)\right]$$
(3.19)

Equation (3.19) implies that given decision functions $\mathcal{P}_{k:N}^*$ of one player (sensing unit), $\hat{x}_{k:N}^*$ are optimal decision variables of the other player (estimator), and vice versa.

To find an optimal solution for (3.11), we define *best response mappings* \mathfrak{P} and \mathfrak{X} as follows.

Definition 3.2.5. Given estimates $\hat{x}_{k:N}$, define $\mathfrak{P}(\hat{x}_{k:N})$ as a collection of policies $\mathcal{P}_{k:N}$ for which it holds that

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}, \hat{x}_{k:N}\right)\right] = \min_{\boldsymbol{\mathcal{P}}'_{k:N}} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}'_{k:N}, \hat{x}_{k:N}\right)\right]$$
(3.20)

Definition 3.2.6. Given policies $\mathcal{P}_{k:N}$, define $\mathfrak{X}(\mathcal{P}_{k:N})$ as a collection of estimates $\hat{x}_{k:N}$ for which it holds that

$$\mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}, \hat{x}_{k:N}\right)\right] = \min_{\hat{x}_{k:N}'} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}, \hat{x}_{k:N}'\right)\right]$$
(3.21)

Definition 3.2.7. Policies $\mathcal{P}_{k:N}$ are said to be degenerate if there exists $j_0 \in \{k, \dots, N\}$ for which it holds that

$$\mathbb{P}\left(\mathbf{R}_{j_0}=0 \middle| \mathbf{R}_k=0,\cdots,\mathbf{R}_{j_0-1}=0\right)=0$$
(3.22)

Remark 3.2.8. Let $\mathcal{P}_{k:N}$ be degenerate policies for which (3.22) holds. Then, from (3.14), we can derive that

$$\mathbb{E}_{\mathbf{x}_{j_0}}\left[J_{j_0}\left(\mathbf{x}_{j_0}, \mathcal{P}_{j_0:N}, \hat{x}_{j_0:N}\right) \middle| \mathbf{R}_k = 0, \cdots, \mathbf{R}_{j_0-1} = 0\right] = c'_{j_0}$$
(3.23)

Proposition 3.2.9. Consider that policies $\mathcal{P}_{k:N}$ and estimates $\hat{x}_{k:N}$ are given. Suppose that the policies are non-degenerate. Then $\mathcal{P}_{k:N}$ belong to $\mathfrak{P}(\hat{x}_{k:N})$ if and only if

$$\mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}\left(\mathbf{x}_{j}, \boldsymbol{\mathcal{P}}_{j:N}, \hat{x}_{j:N}\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right]$$
$$= \mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}^{*}\left(\mathbf{x}_{j}, \hat{x}_{j:N}\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right]$$
(3.24)

holds for all j in $\{k, \dots, N\}$, where

$$J_{j}^{*}(x_{j}, \hat{x}_{j:N}) = \min\left\{ d^{2}(x_{j}, \hat{x}_{j}) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*}(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}) \mid \mathbf{x}_{j} = x_{j} \right], c_{j}^{\prime} \right\}$$
(3.25)

with $J_{N+1}^* = 0$.

The proof follows from (3.14), Definition 3.2.5, and the fact that

$$\min_{\boldsymbol{\mathcal{P}}'_{k:N}} \mathbb{E}_{\mathbf{x}_{k}} \left[J_{k} \left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}'_{k:N}, \hat{x}_{k:N} \right) \right] = \mathbb{E}_{\mathbf{x}_{k}} \left[J_{k}^{*} \left(\mathbf{x}_{k}, \hat{x}_{k:N} \right) \right]$$

We omit the detail for brevity.

Corollary 3.2.10. Given estimates $\hat{x}_{k:N}$, consider (deterministic) policies $\mathcal{P}_{k:N}$ defined by

$$\mathcal{P}_{j}(\mathbf{x}_{j}) = \begin{cases} 0 & \text{if } \mathbf{x}_{j} \in \mathbb{D}_{j} \\ \\ 1 & \text{otherwise} \end{cases}$$
(3.26)

for each j in $\{k, \dots, N\}$, where \mathbb{D}_j is a measurable set for which $\underline{\mathbb{D}}_j \subseteq \mathbb{D}_j \subseteq \overline{\mathbb{D}}_j$ holds with

$$\overline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N} \right) \middle| \mathbf{x}_{j} = x_{j} \right] \le c_{j}^{\prime} \right\}$$
(3.27a)

$$\underline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N} \right) \middle| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$
(3.27b)

Then it holds that $\mathcal{P}_{k:N} \in \mathfrak{P}(\hat{x}_{k:N})$.

Remark 3.2.11. Given estimates $\hat{x}_{k:N}$, let $\mathcal{P}_{k:N}$ be non-degenerate policies for which $\mathcal{P}_{k:N} \in \mathfrak{P}(\hat{x}_{k:N})$ holds. Then Proposition 3.2.9 implies that

I.
$$\mathbb{P}\left(\mathbf{x}_{j} \in \overline{\mathbb{D}}_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right) = 1$$

2. $\mathbb{P}\left(\mathbf{x}_{j} \in \underline{\mathbb{D}}_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1\right) = 0$

where $\overline{\mathbb{D}}_j$ and $\underline{\mathbb{D}}_j$ are given in (3.27).

Proposition 3.2.12. Consider that policies $\mathcal{P}_{k:N}$ and estimates $\hat{x}_{k:N}$ are given. Suppose that the policies are non-degenerate. Then $\hat{x}_{k:N}$ belong to $\mathfrak{X}(\mathcal{P}_{k:N})$ if and only if

$$\mathbb{E}_{\mathbf{x}_{j}}\left[d^{2}\left(\mathbf{x}_{j},\hat{x}_{j}\right) \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]=\min_{\hat{x}_{j}^{\prime}\in\mathbb{X}}\mathbb{E}_{\mathbf{x}_{j}}\left[d^{2}\left(\mathbf{x}_{j},\hat{x}_{j}^{\prime}\right) \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]$$
(3.28)

holds for all j in $\{k, \dots, N\}$.

The proof follows from (3.14) and Definition 3.2.6. We omit the detail for brevity.

Corollary 3.2.13. Given non-degenerate policies $\mathcal{P}_{k:N}$, consider estimates $\hat{x}_{k:N}$ defined by

$$\hat{x}_{j} \in \underset{\hat{x}_{j} \in \mathbb{X}}{\operatorname{arg\,min}} \mathbb{E}_{\mathbf{x}_{j}} \left[d^{2} \left(\mathbf{x}_{j}, \hat{x}_{j} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
(3.29)

for each j in $\{k, \dots, N\}$. Then it holds that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$.

Proposition 3.2.14. Consider functions $\{\mathcal{G}_j\}_{j=k}^N$ defined by

$$\mathcal{G}_{j}\left(x_{j-1}, \hat{x}_{j:N}\right) \stackrel{def}{=} \mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}^{*}\left(\mathbf{x}_{j}, \hat{x}_{j:N}\right) \middle| \mathbf{x}_{j-1} = x_{j-1}\right]$$
(3.30)

where J_j^* is given in (3.25). $\{\mathcal{G}_j\}_{j=k}^N$ are all continuous functions.⁷

The proof is given in Appendix B.4. The following is a consequence of Proposition 3.2.14.

Corollary 3.2.15. Given estimates $\hat{x}_{k:N}$, the sets $\overline{\mathbb{D}}_j$ and $\underline{\mathbb{D}}_j$ defined in (3.27) are closed and open, respectively, for all j in $\{k, \dots, N\}$.

3.2.2 Existence of a Jointly Optimal Solution

Let us define

$$\mathcal{G}\left(\hat{x}_{k:N}\right) \stackrel{def}{=} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}^{*}\left(\mathbf{x}_{k}, \hat{x}_{k:N}\right)\right]$$
(3.31)

where J_k^* is given in (3.25). Note that $\mathcal{G}(\hat{x}_{k:N}) = \min_{\mathcal{P}'_{k:N}} \mathbb{E}_{\mathbf{x}_k} \left[J_k(\mathbf{x}_k, \mathcal{P}'_{k:N}, \hat{x}_{k:N}) \right].$

⁷Note that \mathcal{G}_j is a function defined on \mathbb{X}^{N-j+2} . See Appendix B.1 for some remarks on the continuity of functions on a product space.

Proposition 3.2.16. Let $\hat{x}_{k:N}^*$ be the estimates that achieve the global minimum of (3.31). The policies $\mathcal{P}_{k:N}^*$ satisfying

$$\mathcal{P}_{k:N}^* \in \mathfrak{P}(\hat{x}_{k:N}^*)$$

are not degenerate in the sense of Definition 3.2.7.

The proof is given in Appendix B.5.

Theorem 3.2.17. There exist estimates $\hat{x}_{k:N}^*$ that achieve the global minimum of (3.31). Furthermore, in conjunction with these estimates $\hat{x}_{k:N}^*$, the policies $\mathcal{P}_{k:N}^*$ satisfying

$$\boldsymbol{\mathcal{P}}_{k:N}^* \in \boldsymbol{\mathfrak{P}}\left(\hat{x}_{k:N}^*\right)$$

are a jointly optimal solution for (3.11).

To prove Theorem 3.2.17, we need the following Lemma.

Lemma 3.2.18. There exists a compact set $\mathbb{K} \subset \mathbb{X}^{N-k+1}$ for which

$$\min_{\hat{x}_{k:N} \in \mathbb{K}} \mathcal{G}\left(\hat{x}_{k:N}\right) \le \mathcal{G}\left(\hat{x}_{k:N}'\right)$$

holds for all $\hat{x}'_{k:N}$ in \mathbb{X}^{N-k+1} .

The proof is given in Appendix B.5.

Proof of Theorem 3.2.17: Recall the definitions of \mathcal{G}_k and \mathcal{G} given in (3.30) and (3.31), respectively. According to Proposition 3.2.14 and by the fact that $\mathcal{G}(\hat{x}_{k:N}) = \mathcal{G}_k(0, \hat{x}_{k:N})$, we can see that \mathcal{G} is a continuous function. Due to Lemma 3.2.18, in what regards to finding a global minimizer of \mathcal{G} , we may assume that the domain of \mathcal{G} is compact. Hence, by the continuity of \mathcal{G} and compactness of its domain, there exist estimates $\hat{x}^*_{k:N}$ that achieve the global minimum of \mathcal{G} . Let us choose policies $\mathcal{P}^*_{k:N}$ satisfying $\mathcal{P}_{k:N}^* \in \mathfrak{P}(\hat{x}_{k:N}^*)$ using, for instance, Corollary 3.2.10. Since $\hat{x}_{k:N}^*$ is a global minimizer of \mathcal{G} and $\mathcal{P}_{k:N}^* \in \mathfrak{P}(\hat{x}_{k:N}^*)$ holds, we conclude that the solution $\mathcal{P}_{k:N}^*$ and $x_{k:N}^*$ is jointly optimal for (3.11).

3.2.3 Iterative Procedure for Finding a Person-by-Person Optimal Solution

As numerically illustrated in [70], the function \mathcal{G} in (3.31) may be non-convex, and finding a jointly optimal solution to (3.11) would be computationally intractable. Instead, in this section, we seek a person-by-person optimal solution to (3.11). An iterative procedure for finding such a solution is described in Procedure 1, where η is a pre-selected non-negative constant that determines a stopping criterion of the procedure.

input $: \eta \ge 0, \hat{x}_{k:N}^{(0)}$ output: $\mathcal{P}_{k:N}^{(i+1)}, \hat{x}_{k:N}^{(i)}$ 1 begin $j \leftarrow N$ 2 while $j \ge k$ do 3 Choose $\mathcal{P}_{j}^{(1)}$ using (3.26) 4 $j \leftarrow j - 1$ 5 $i \leftarrow 0$ 6 repeat 7 $i \leftarrow i+1$ 8 $j \leftarrow k$ 9 while $j \leq N$ do 10 Choose $\hat{x}_{j}^{(i)}$ using (3.29) $j \leftarrow j + 1$ 11 12 $j \leftarrow N$ 13 while $j \ge k$ do 14 Choose $\mathcal{P}_{j}^{(i+1)}$ using (3.26) $j \leftarrow j - 1$ 15 16 $\mathbf{until} \left| \mathcal{G} \left(\hat{x}_{k:N}^{(i)} \right) - \mathcal{G} \left(\hat{x}_{k:N}^{(i-1)} \right) \right| \leq \eta$ 17

In what follows, we analyze convergence properties of the sequence of solutions computed by Line 8 - 16 of Procedure 1. We first define convergence of policies and estimates. To proceed, for each A in \mathfrak{B} , let us define

$$\mu_{j|j}^{(i)}(\mathbb{A}) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \middle| \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j}^{(i)} = 0\right)$$
$$\mu_{j|j}(\mathbb{A}) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right)$$

Definition 3.2.19. Let $\{\mathcal{P}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ be a sequence of policies. The sequence is said to converge to $\mathcal{P}_{k:N}$ if it holds that⁸

$$\mu_{j|j}^{(i)} \xrightarrow{w} \mu_{j|j} \tag{3.32a}$$

and

$$\mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0, \cdots, \mathbf{R}_{j-1}^{(i)}=0\right)$$

$$\stackrel{i \to \infty}{\longrightarrow} \mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right)$$
(3.32b)

for all j in $\{k, \dots, N\}$. In addition, two sets of policies $\mathcal{P}_{k:N}$ and $\mathcal{P}'_{k:N}$ are said to be equal if it holds that⁹

$$\mu_{j|j} = \mu'_{j|j} \tag{3.33a}$$

and

$$\mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j-1}=0\right)$$
$$=\mathbb{P}\left(\mathbf{R}_{j}'=0 \mid \mathbf{R}_{k}'=0,\cdots,\mathbf{R}_{j-1}'=0\right)$$
(3.33b)

for all j in $\{k, \cdots, N\}$.

⁸See Definition B.3.6 for the weak convergence of probability measures.

 $^{^{9}\}mu_{j|j} = \mu'_{j|j}$ implies that $\mu_{j|j}(\mathbb{A}) = \mu'_{j|j}(\mathbb{A})$ holds for all \mathbb{A} in \mathfrak{B} .

Remark 3.2.20 (Uniqueness of the Limit of Policies). Let $\{\mathcal{P}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ be a sequence of policies that converge to $\mathcal{P}_{k:N}$ and $\mathcal{P}'_{k:N}$. Then the two sets of the policies $\mathcal{P}_{k:N}$ and $\mathcal{P}'_{k:N}$ are equal. To see this, using the definition of the weak convergence of probability measures, we can derive that

$$\int_{\mathbb{X}} g \,\mathrm{d}\mu_{j|j} = \int_{\mathbb{X}} g \,\mathrm{d}\mu'_{j|j} \tag{3.34}$$

for every bounded, continuous function $g : \mathbb{X} \to \mathbb{R}$. Then, by applying Lemma 9.3.2 in [93], we can see that (3.33) holds for all j in $\{k, \dots, N\}$.

Definition 3.2.21. Let $\{\hat{x}_{k:N}^{(i)}\}_{i \in \mathbb{N}}$ be a sequence of estimates. The sequence is said to converge to $\hat{x}_{k:N}$ if it holds that

$$\lim_{i \to \infty} d\left(\hat{x}_j^{(i)}, \hat{x}_j\right) = 0 \tag{3.35}$$

for all j in $\{k, \dots, N\}$. In addition, two sets of estimates $\hat{x}_{k:N}$ and $\hat{x}'_{k:N}$ are said to be equal if it holds that

$$d\left(\hat{x}_{j}, \hat{x}_{j}'\right) = 0 \tag{3.36}$$

for all j in $\{k, \dots, N\}$.

In the following Theorem, we examine convergence properties of sequences $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ and $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ that satisfy

$$\boldsymbol{\mathcal{P}}_{k:N}^{(i)} \in \boldsymbol{\mathfrak{P}}\left(\hat{x}_{k:N}^{(i-1)}\right)$$
(3.37a)

$$\hat{x}_{k:N}^{(i)} \in \mathfrak{X}\left(\boldsymbol{\mathcal{P}}_{k:N}^{(i)}\right)$$
(3.37b)

To proceed, we make the following assumption.

Assumption 3.2.22. Consider sequences $\{\mathcal{P}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ and $\{\hat{x}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ that satisfy (3.37). Suppose that the subsequences $\{\mathcal{P}_{k:N}^{(i_l)}\}_{l\in\mathbb{N}}$, $\{\hat{x}_{k:N}^{(i_l)}\}_{l\in\mathbb{N}}$, and $\{\hat{x}_{k:N}^{(i_l-1)}\}_{l\in\mathbb{N}}$ converge to $\mathcal{P}_{k:N}$, $\hat{x}_{k:N}$, and $\hat{x}'_{k:N}$, respectively. We assume that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$.

Theorem 3.2.23. Let $\left\{ \boldsymbol{\mathcal{P}}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ and $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ be a sequence of solutions satisfying (3.37). Suppose that Assumption 3.2.22 holds and the policies $\left\{ \boldsymbol{\mathcal{P}}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ are strictly non-degenerate, i.e., there exists a positive constant ϵ for which

$$\mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0, \cdots, \mathbf{R}_{j-1}^{(i)}=0\right) > \epsilon$$
(3.38)

holds for all i in \mathbb{N} and j in $\{k, \dots, N\}$. Then, the sequence of the solutions $\{\mathcal{P}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ and $\{\hat{x}_{k:N}^{(i)}\}_{i\in\mathbb{N}}$ has a convergent subsequence, and the limit of any convergence subsequence is a person-by-person optimal solution.

To prove Theorem 3.2.23, we need the following Lemma.

Lemma 3.2.24. Consider sequences $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ and $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ satisfying (3.37). Suppose that $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ are strictly non-degenerate and $\left\{ \hat{x}_{k:N}^{(i-1)} \right\}_{i \in \mathbb{N}}$ converges to $\hat{x}'_{k:N}$. Then, the sequence $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ has a convergent subsequence, and the limit $\mathcal{P}_{k:N}$ of any convergence subsequence satisfies $\mathcal{P}_{k:N} \in \mathfrak{P}(\hat{x}'_{k:N})$.

The proof is given in Appendix B.6.

Proof of Theorem 3.2.23: We first note that according to Lemma B.5.2, the sequence of the estimates $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}}$ is contained in a compact set. By the compactness, there exists an infinite subset \mathbb{I} of \mathbb{N} for which the subsequences $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{I}}$ and $\left\{\hat{x}_{k:N}^{(i-1)}\right\}_{i\in\mathbb{I}}$ are convergent. Let $\hat{x}_{k:N}$ and $\hat{x}'_{k:N}$ be respective limits of $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{I}}$ and $\left\{\hat{x}_{k:N}^{(i-1)}\right\}_{i\in\mathbb{I}}$. Also, according to Lemma 3.2.24, there is an infinite subset \mathbb{I}' of \mathbb{I} for which the subsequence $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{I}'}$ is convergent. Let $\mathcal{P}_{k:N}$ be the limit of $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{I}'}$.

To complete the proof, it remains to show that $\mathcal{P}_{k:N}$ and $\hat{x}_{k:N}$ constitute a personby-person optimal solution, i.e., it holds that

$$\boldsymbol{\mathcal{P}}_{k:N} \in \boldsymbol{\mathfrak{P}}\left(\hat{x}_{k:N}\right) \tag{3.39a}$$

$$\hat{x}_{k:N} \in \mathfrak{X}\left(\boldsymbol{\mathcal{P}}_{k:N}\right) \tag{3.39b}$$

Equation (3.39b) is ensured by Assumption 3.2.22, and it remains to show that (3.39a) is true.

To see this, by contradiction, suppose that $\mathcal{P}_{k:N} \notin \mathfrak{P}(\hat{x}_{k:N})$ holds. Note that by Lemma 3.2.24, $\mathcal{P}_{k:N}$ belong to $\mathfrak{P}(\hat{x}'_{k:N})$. Then we can see that the following relations hold for policies $\mathcal{P}'_{k:N}$ satisfying $\mathcal{P}'_{k:N} \in \mathfrak{P}(\hat{x}_{k:N})$:

$$\mathcal{G}\left(\hat{x}_{k:N}\right) = \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}'_{k:N}, \hat{x}_{k:N}\right)\right]$$

$$\stackrel{(1)}{\leq} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}, \hat{x}_{k:N}\right)\right]$$

$$\stackrel{(2)}{\leq} \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}, \hat{x}'_{k:N}\right)\right] = \mathcal{G}\left(\hat{x}'_{k:N}\right)$$
(3.40)

(1) follows from the hypothesis that $\mathcal{P}_{k:N} \notin \mathfrak{P}(\hat{x}_{k:N})$; and (2) is due to (3.39b). On the other hand, since \mathcal{G} is decreasing along the sequence $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}}$, i.e.,

$$\mathcal{G}\left(\hat{x}_{k:N}^{(i)}\right) \ge \mathcal{G}\left(\hat{x}_{k:N}^{(i+1)}\right)$$

holds for all i in \mathbb{N} , it holds that $\lim_{i\to\infty} \mathcal{G}\left(\hat{x}_{k:N}^{(i)}\right) = \alpha$ for some real number α . In conjunction with the continuity of \mathcal{G} (see Proposition 3.2.14), this implies that

$$\mathcal{G}\left(\hat{x}_{k:N}\right) = \mathcal{G}\left(\hat{x}_{k:N}'\right) = \alpha$$

which contradicts (3.40). Therefore we conclude that $\mathcal{P}_{k:N} \in \mathfrak{P}(\hat{x}_{k:N})$.

3.3 Optimal Remote State Estimation

Based on the results of Section 3.2, we find a solution to Problem 3.1.1. To proceed, consider a procedure described in Procedure 2.

Procedure 2: Finding solutions to Sub-problems	
output : $\left\{ \mathcal{T}_{k:N}^{* < k-1>} \right\}_{k=1}^{N}$ and $\left\{ \mathcal{E}_{k:N}^{* < k-1>} \right\}_{k=1}^{N}$	
1 begin	
2	$k \leftarrow N$
3	while $k \ge 1$ do
4	Step 1: Compute constants $\left\{c_{j}^{\prime*}\right\}_{j=k}^{N}$ according to (3.8) for the solutions
	$\left\{ \boldsymbol{\mathcal{T}}_{j:N}^{* < j-1 >} \right\}_{j=k+1}^{N}$ and $\left\{ \boldsymbol{\mathcal{E}}_{j:N}^{* < j-1 >} \right\}_{j=k+1}^{N}$ to Sub-problem $k+1$ to
	Sub-problem N.
5	Step 2: Find $\mathcal{T}_{k:N}^{* < k-1 >}$ and $\mathcal{E}_{k:N}^{* < k-1 >}$ that is an optimal solution of Sub-problem k
	with the constants $\left\{c_{j}^{\prime*}\right\}_{j=k}^{N}$ obtain in Step 1.
6	$k \leftarrow k - 1$

Based on solutions obtained via Procedure 2, we can state the following Theorem.

Theorem 3.3.1. Let $\left\{ \mathcal{T}_{k:N}^{* < k-1>} \right\}_{k=1}^{N}$ and $\left\{ \mathcal{E}_{k:N}^{* < k-1>} \right\}_{k=1}^{N}$ be solutions to Sub-problems obtained via Procedure 2. For each k in $\{1, \dots, N\}$, if $\mathcal{T}_{k:N}^{* < k-1>}$ and $\mathcal{E}_{k:N}^{* < k-1>}$ are a jointly optimal (person-by-person optimal) solution of Sub-problem k, then the transmission policies $\mathcal{T}_{1:N}^{*}$ and the estimation rules $\mathcal{E}_{1:N}^{*}$ determined by

$$\mathcal{T}_{j}^{*}(k-1, x_{k-1}, x_{j}) = \mathcal{T}_{j}^{* < k-1 >}(x_{k-1}, x_{j})$$
 (3.41a)

$$\mathcal{E}_{j}^{*}(k-1, x_{k-1}) = \mathcal{E}_{j}^{* < k-1 >}(x_{k-1})$$
(3.41b)

for each j in $\{k, \dots, N\}$ and k in $\{1, \dots, N\}$ are jointly optimal (person-by-person optimal) for (3.2).

The proof is given in Appendix B.7

3.4 Application to Specific System Models

In this section, we apply our main results to linear system models and self-propelled particle models.

3.4.1 Linear System Models

Consider

$$\mathbf{x}_{k+1} = A_k \mathbf{x}_k + \mathbf{w}_k \tag{3.42}$$

where \mathbf{w}_k is a random variable with a Gaussian distribution in \mathbb{R}^n . We define the metric d using the Euclidean norm $\|\cdot\|_2$ as $d(x_k, x'_k) = \|x_k - x'_k\|_2$. We note that the metric is invariant under the transformation defined by

$$M_j (k - 1, x_{k-1}, x_j) = x_j - \prod_{l=k-1}^{j-1} A_l x_{k-1}$$

for j in $\{k - 1, \dots, N\}$ and k in $\{1, \dots, N\}$, where we adopt the convention that $\prod_{l=k-1}^{j-1} A_l = I_n$ if j = k - 1. It can be verified that Assumption 3.1.4 - Assumption 3.1.6 hold.

With the Euclidean norm, it is straight-forward to see that for given polices $\mathcal{P}_{k:N}$, for each j in $\{k, \dots, N\}$,

$$\hat{x}_j \in \operatorname*{arg\,min}_{\hat{x}_j \in \mathbb{R}^n} \mathbb{E}_{\mathbf{x}_j} \left[\|\mathbf{x}_j - \hat{x}_j\|^2 \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]$$

and

$$\hat{x}_j = \mathbb{E}_{\mathbf{x}_j} \left[\mathbf{x}_j \, \middle| \, \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]$$
(3.43)

are equivalent, provided that $\mathbb{E}_{\mathbf{x}_j} \left[\|\mathbf{x}_j - \hat{x}_j\|^2 \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]$ is well-defined for all \hat{x}_j in \mathbb{R}^n . In the following Proposition, we show that the statement in Assumption 3.2.22 is valid for the linear system models.

Proposition 3.4.1. Consider sequences $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ and $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ that satisfy

$$\boldsymbol{\mathcal{P}}_{k:N}^{(i)} \in \boldsymbol{\mathfrak{P}}\left(\hat{x}_{k:N}^{(i-1)}\right)$$
(3.44a)

$$\hat{x}_{k:N}^{(i)} \in \mathfrak{X}\left(\boldsymbol{\mathcal{P}}_{k:N}^{(i)}\right)$$
(3.44b)

Suppose that the subsequences $\left\{ \mathcal{P}_{k:N}^{(i_l)} \right\}_{l \in \mathbb{N}}$, $\left\{ \hat{x}_{k:N}^{(i_l)} \right\}_{l \in \mathbb{N}}$, and $\left\{ \hat{x}_{k:N}^{(i_l-1)} \right\}_{l \in \mathbb{N}}$ converge to $\mathcal{P}_{k:N}$, $\hat{x}_{k:N}$, and $\hat{x}'_{k:N}$, respectively. Then it holds that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$.

Proof. Let $\mu_{j|j}^{(i)}$ and $\mu_{j|j}$ be probability measures defined as

$$\mu_{j|j}^{(i)}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j}^{(i)} = 0\right)$$
(3.45)

$$\mu_{j|j}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right)$$
(3.46)

where \mathbb{A} belongs to \mathfrak{B} , and the random variables $\mathbf{R}_{j}^{(i)}$ and \mathbf{R}_{j} are dictated by $\mathcal{P}_{j}^{(i)}$ and \mathcal{P}_{j} , respectively. Since $\left\{\mathcal{P}_{k:N}^{(i_{l})}\right\}_{l\in\mathbb{N}}$ converges to $\mathcal{P}_{k:N}$, it holds that $\mu_{j|j}^{(i_{l})} \xrightarrow{w} \mu_{j|j}$ for all j in $\{k, \dots, N\}$. Since $(\mathbb{R}^{n}, \|\cdot\|_{2})$ is a complete, separable metric space, by the weak convergence of $\left\{\mu_{j|j}^{(i_{l})}\right\}_{l\in\mathbb{N}}$ and the Skorokhod representation theorem [94], there exist a sequence of random variables $\left\{\mathbf{y}_{j}^{(i_{l})}\right\}_{l\in\mathbb{N}}$ and a random variable \mathbf{y}_{j} all defined on a common probability space $(\Omega, \mathfrak{F}, \nu)$ in which the following three facts are true:

- (F1) $\mu_{j|j}^{(i_l)}$ is the probability measure of $\mathbf{y}_j^{(i_l)}$, i.e., $\nu\left(\left\{\omega \in \Omega \mid \mathbf{y}_j^{(i_l)}(\omega) \in \mathbb{A}\right\}\right) = \mu_{j|j}^{(i_l)}(\mathbb{A})$ for each \mathbb{A} in \mathfrak{B} .
- (F2) $\mu_{j|j}$ is the probability measure of \mathbf{y}_j , i.e., $\nu\left(\left\{\omega \in \Omega \mid \mathbf{y}_j(\omega) \in \mathbb{A}\right\}\right) = \mu_{j|j}(\mathbb{A})$ for each \mathbb{A} in \mathfrak{B} .
- (F3) $\left\{\mathbf{y}_{j}^{(i_{l})}\right\}_{l\in\mathbb{N}}$ converges to \mathbf{y}_{j} almost surely.

Hence, by Proposition 3.2.12 and (3.43), we can derive that

$$\hat{x}_{j}^{(i_{l})} = \mathbb{E}_{\mathbf{x}_{j}} \left[\mathbf{x}_{j} \middle| \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right]$$
$$= \int_{\mathbb{R}^{n}} x \, \mathrm{d}\mu_{j|j}^{(i_{l})}$$
$$= \int_{\Omega} \mathbf{y}_{j}^{(i_{l})}(\omega) \, \mathrm{d}\nu$$
(3.47)

Since $\{\hat{x}_{j}^{(i_{l}-1)}\}_{l\in\mathbb{N}}$ is a convergent sequence, it is bounded. By Remark 3.2.11, (3.44a), and Lemma B.3.9, there is a compact set \mathbb{K}_{j} for which $\mu_{j|j}^{(i_{l})}(\mathbb{K}_{j}) = 1$ for all l in \mathbb{N} . Hence, there is a positive real α for which it holds that

$$\int_{\left\{\omega\in\Omega\left|\left\|\mathbf{y}_{j}^{(i_{l})}(\omega)\right\|>\alpha\right\}}\mathbf{y}_{j}^{(i_{l})}(\omega)\,\mathrm{d}\nu = \int_{\left\{x\in\mathbb{X}\left|\left\|x\right\|>\alpha\right\}}x\,\mathrm{d}\mu_{j\mid j}^{(i_{l})} = 0\tag{3.48}$$

for all l in \mathbb{N} . In conjunction with (**F3**), by an application of Theorem 10.3.6 in [93], we have that

$$\lim_{l \to \infty} \int_{\Omega} \mathbf{y}_{j}^{(i_{l})}(\omega) \,\mathrm{d}\nu = \int_{\Omega} \mathbf{y}_{j}(\omega) \,\mathrm{d}\nu$$
(3.49)

Therefore, from (3.47) and (3.49), we can see that

$$\hat{x}_{j} = \lim_{l \to \infty} \hat{x}_{j}^{(i_{l})} = \lim_{l \to \infty} \int_{\Omega} \mathbf{y}_{j}^{(i_{l})}(\omega) \,\mathrm{d}\nu$$
$$= \int_{\Omega} \mathbf{y}_{j}(\omega) \,\mathrm{d}\nu$$
$$= \int_{\mathbb{X}} x \,\mathrm{d}\mu_{j|j}$$
$$= \mathbb{E} \left[\mathbf{x}_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
(3.50)

Since this holds for every j in $\{k, \dots, N\}$, by Proposition 3.2.12, we conclude that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N}).$

3.4.2 Self-Propelled Particle Models

Consider

$$\begin{pmatrix} \mathbf{p}_{1,k+1} \\ \mathbf{p}_{2,k+1} \\ \boldsymbol{\theta}_{k+1} \end{pmatrix} = \begin{pmatrix} \mathbf{p}_{1,k} + \mathbf{r}_k \cdot \cos\left(\boldsymbol{\theta}_k + \boldsymbol{\phi}_k\right) \\ \mathbf{p}_{2,k} + \mathbf{r}_k \cdot \sin\left(\boldsymbol{\theta}_k + \boldsymbol{\phi}_k\right) \\ \boldsymbol{\theta}_k + \boldsymbol{\phi}_k \end{pmatrix}$$
(3.51)

where $\mathbf{p}_{1,k}$, $\mathbf{p}_{2,k}$ take values in \mathbb{R} , and $\boldsymbol{\theta}_k$ takes a value in $[-\pi, \pi)$. \mathbf{r}_k and $\boldsymbol{\phi}_k$ are random variables with a Weibull distribution and Wrapped Cauchy distribution, respectively. We define the metric *d* using the Frobenius norm $\|\cdot\|_F$ as follows:

$$d\left(\begin{pmatrix}p_{1,k}\\p_{2,k}\\\theta_k\end{pmatrix},\begin{pmatrix}p'_{1,k}\\p'_{2,k}\\\theta'_k\end{pmatrix}\right) = \left\|\begin{pmatrix}\cos\theta_k & -\sin\theta_k & p_{1,k}\\\sin\theta_k & \cos\theta_k & p_{2,k}\\0 & 0 & 1\end{pmatrix} - \begin{pmatrix}\cos\theta'_k & -\sin\theta'_k & p'_{1,k}\\\sin\theta'_k & \cos\theta'_k & p'_{2,k}\\0 & 0 & 1\end{pmatrix}\right\|_F$$

where the metric is invariant under the transformation given by

$$M_{j}\left(k-1, \begin{pmatrix} p_{1,k-1} \\ p_{2,k-1} \\ \theta_{k-1} \end{pmatrix}, \begin{pmatrix} p_{1,j} \\ p_{2,j} \\ \theta_{j} \end{pmatrix}\right) = \begin{pmatrix} \cos\theta_{k-1} & \sin\theta_{k-1} & 0 \\ -\sin\theta_{k-1} & \cos\theta_{k-1} & 0 \\ 0 & 0 & 1 \end{pmatrix} \cdot \begin{pmatrix} p_{1,j}-p_{1,k-1} \\ p_{2,j}-p_{2,k-1} \\ \theta_{j}-\theta_{k-1} \end{pmatrix}$$

It can be verified that Assumption 3.1.4 - Assumption 3.1.6 hold.

By defining
$$\mathbf{x}_{j} = \begin{pmatrix} \mathbf{p}_{1,j} & \mathbf{p}_{2,j} & \boldsymbol{\theta}_{j} \end{pmatrix}^{T}$$
 and $\hat{x}_{j} = \begin{pmatrix} \hat{p}_{1,j} & \hat{p}_{2,j} & \hat{\theta}_{j} \end{pmatrix}^{T}$, we can derive

$$\mathbb{E} \begin{bmatrix} d^{2} (\mathbf{x}_{j}, \hat{x}_{j}) & | \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \end{bmatrix}$$

$$= \mathbb{E} \begin{bmatrix} (\mathbf{p}_{1,j} - \hat{p}_{1,j})^{2} + (\mathbf{p}_{2,j} - \hat{p}_{2,j})^{2} + 4 \cdot \left(1 - \cos\left(\boldsymbol{\theta}_{j} - \hat{\theta}_{j}\right)\right) & | \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \end{bmatrix}$$
(3.52)

provided that the conditional expectation is well-defined for all \hat{x}_j in $\mathbb{R} \times \mathbb{R} \times [-\pi, \pi)$. The first and second order conditions of optimality for (3.52) yield that a minimizer \hat{x}_j satisfies

$$\hat{p}_{1,j} = \mathbb{E}\left[\mathbf{p}_{1,j} \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0\right]$$
(3.53a)

$$\hat{p}_{2,j} = \mathbb{E} \left[\mathbf{p}_{2,j} \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]$$

$$\sin \hat{\theta}_j = \frac{\mathbb{E} \left[\sin \theta_j \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]}{\mathbb{E}^2 \left[\sin \theta_j \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right] + \mathbb{E}^2 \left[\cos \theta_j \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right]}$$
(3.53c)

$$\cos \hat{\theta}_{j} = \frac{\mathbb{E}\left[\cos \theta_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right]}{\mathbb{E}^{2}\left[\sin \theta_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right] + \mathbb{E}^{2}\left[\cos \theta_{j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right]}$$
(3.53d)

provided that at least one of

$$\mathbb{E}\left[\sin\boldsymbol{\theta}_{j} \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]$$
(3.54a)

$$\mathbb{E}\left[\cos\boldsymbol{\theta}_{j} \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]$$
(3.54b)

is non-zero. In this case, there exists a unique $\hat{\theta}_j$ that satisfies (3.53c) and (3.53d). If both (3.54a) and (3.54b) are zero, then the value of (3.52) does not depend on $\hat{\theta}_j$. In the following Proposition, we show that the statement in Assumption 3.2.22 is valid for the self-propelled particle models.

Proposition 3.4.2. Consider sequences $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ and $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ that satisfy

$$\mathcal{P}_{k:N}^{(i)} \in \mathfrak{P}\left(\hat{x}_{k:N}^{(i-1)}\right)$$
(3.55a)
$$\hat{x}_{k:N}^{(i)} \in \mathfrak{X}\left(\mathcal{P}_{k:N}^{(i)}\right)$$
(3.55b)

Suppose that the subsequences $\left\{ \mathcal{P}_{k:N}^{(i_l)} \right\}_{l \in \mathbb{N}}$, $\left\{ \hat{x}_{k:N}^{(i_l)} \right\}_{l \in \mathbb{N}}$, and $\left\{ \hat{x}_{k:N}^{(i_l-1)} \right\}_{l \in \mathbb{N}}$ converge to $\mathcal{P}_{k:N}$, $\hat{x}_{k:N}$, and $\hat{x}'_{k:N}$, respectively. Then it holds that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$.

Proof. By a similar argument as in the proof of Proposition 3.4.1, we can show that

$$\lim_{l \to \infty} \mathbb{E}\left[\mathbf{p}_{1,j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0\right] = \mathbb{E}\left[\mathbf{p}_{1,j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right]$$
(3.56a)

$$\lim_{l \to \infty} \mathbb{E}\left[\mathbf{p}_{2,j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0\right] = \mathbb{E}\left[\mathbf{p}_{2,j} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right]$$
(3.56b)

$$\lim_{l \to \infty} \mathbb{E} \left[\sin \boldsymbol{\theta}_j \, \middle| \, \mathbf{R}_k^{(i_l)} = 0, \cdots, \mathbf{R}_j^{(i_l)} = 0 \right] = \mathbb{E} \left[\sin \boldsymbol{\theta}_j \, \middle| \, \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0 \right] \quad (3.56c)$$

$$\lim_{l \to \infty} \mathbb{E}\left[\cos \theta_{j} \left| \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] = \mathbb{E}\left[\cos \theta_{j} \left| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right] \right]$$
(3.56d)

Suppose that at least one of

$$\mathbb{E}\left[\sin\boldsymbol{\theta}_{j} \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]$$
(3.57)

$$\mathbb{E}\left[\cos\boldsymbol{\theta}_{j} \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j}=0\right]$$
(3.58)

is non-zero. Note that by Proposition 3.2.12 and (3.53), $\hat{x}_j^{(i_l)} = \begin{pmatrix} \hat{p}_{1,j}^{(i_l)} & \hat{p}_{2,j}^{(i_l)} & \hat{\theta}_j^{(i_l)} \end{pmatrix}^T$ and

$$\begin{aligned} \hat{x}_{j} &= \left(\hat{p}_{1,j} \quad \hat{p}_{2,j} \quad \hat{\theta}_{j} \right)^{T} \text{ satisfy} \\ p_{1,j}^{(i_{l})} &= \mathbb{E} \left[\mathbf{p}_{1,j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \\ p_{2,j}^{(i_{l})} &= \mathbb{E} \left[\mathbf{p}_{2,j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \\ \sin \hat{\theta}_{j}^{(i_{l})} &= \frac{\mathbb{E} \left[\sin \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \\ \mathbb{E}^{2} \left[\sin \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] + \mathbb{E}^{2} \left[\cos \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \\ \cos \hat{\theta}_{j}^{(i_{l})} &= \frac{\mathbb{E} \left[\cos \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \\ \mathbb{E}^{2} \left[\sin \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] + \mathbb{E}^{2} \left[\cos \theta_{j} \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j}^{(i_{l})} = 0 \right] \end{aligned}$$

and

$$\lim_{l \to \infty} d\left(\hat{x}_j^{(i_l)}, \hat{x}_j\right) = 0 \tag{3.59}$$

For each j in $\{k, \dots, N\}$, let us define $\hat{x}_j^* = \begin{pmatrix} \hat{p}_{1,j}^* & \hat{p}_{2,j}^* & \hat{\theta}_j^* \end{pmatrix}$ as

$$p_{1,j}^{*} = \mathbb{E} \left[\mathbf{p}_{1,j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$

$$p_{2,j}^{*} = \mathbb{E} \left[\mathbf{p}_{2,j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$

$$\sin \hat{\theta}_{j}^{*} = \frac{\mathbb{E} \left[\sin \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]}{\mathbb{E}^{2} \left[\sin \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right] + \mathbb{E}^{2} \left[\cos \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]}$$

$$\cos \hat{\theta}_{j}^{*} = \frac{\mathbb{E} \left[\cos \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]}{\mathbb{E}^{2} \left[\sin \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right] + \mathbb{E}^{2} \left[\cos \theta_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]}$$

Note that by Proposition 3.2.12 and (3.53), it holds that $\hat{x}_{k:N}^* \in \mathfrak{X}(\mathcal{P}_{k:N})$. From (3.56) and (3.59), we can observe that

$$d\left(\hat{x}_{j}, \hat{x}_{j}^{*}\right) \leq d\left(\hat{x}_{j}^{(i_{l})}, \hat{x}_{j}^{*}\right) + d\left(\hat{x}_{j}^{(i_{l})}, \hat{x}_{j}\right) \stackrel{l \to \infty}{\longrightarrow} 0$$

Therefore, we conclude that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$.

If both (3.57) and (3.58) are zero, then the value of (3.52) does not depend on $\hat{\theta}_j$, and by Proposition 3.2.12, we can show that $\hat{x}_{k:N} \in \mathfrak{X}(\mathcal{P}_{k:N})$ if it holds that

$$p_{1,j} = \mathbb{E}\left[\mathbf{p}_{1,j} \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0\right]$$
$$p_{2,j} = \mathbb{E}\left[\mathbf{p}_{2,j} \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_j = 0\right]$$

This can be verified by similar arguments given above and (3.56).

3.5 Application to Animal Tracking and Experimental Results

In this section, we apply the proposed remote estimation scheme to animal tracking, and show preliminary experimental results using a data set collected from the deployment of *animal-borne wireless camera network* in the Gorongosa National Park (Mozambique) in August 2015.¹⁰ The main purpose of the development and deployment of the system was to collect biologically meaningful measurements and videos using GPS, IMU, and Camera all integrated in a single tracking device, where the proposed estimation scheme can be used to determine when to share sensor measurements between tracking devices and how to determine the best location estimates of nearby devices (see Figure 2.5). The sensor measurements and videos are used to study animal group motion. During the deployment, 15 tracking devices were installed on waterbucks and water buffaloes. Figure 3.2 shows the GPS track of a water buffalo, and Figure 3.3 depicts the *x*-coordinate ($p_{x,k}$),

¹⁰The development and deployment of animal-borne wireless camera network were performed under a research grant NSF ECCS 1135726.

Disclaimer: The author of this dissertation was NOT involved in the deployment in the Gorongosa National Park.

y-coordinate $(p_{y,k})$, and heading angle θ_k of a portion of the GPS track (contained in the red rectangle in Figure 3.2) in a local North East Up (NEU) coordinate system.



Figure 3.2: A screenshot of the GPS track of a water buffalo in the Google earth

(Timespan: 2015-08-06T00:00:00Z \sim 2015-08-06T06:00:00Z)



Figure 3.3: The x-coordinate $(p_{x,k})$, y-coordinate $(p_{y,k})$, and heading angle θ_k of a portion of the GPS track (contained in the red rectangle in Figure 3.2) in a local NEU coordinate system. (Timespan: 2015-08-06T05:40:00Z ~

2015-08-06T05:53:00Z / The origin of the coordinate system :

Latitude = -18.9401136372457, Longitude = 34.5337888580266)

We model the movement of the water buffalo using the self-propelled particle model described in Section 3.4.2:

$$\begin{pmatrix} \mathbf{p}_{x,k+1} \\ \mathbf{p}_{y,k+1} \\ \boldsymbol{\theta}_{k+1} \end{pmatrix} = \begin{pmatrix} \mathbf{p}_{x,k} + \mathbf{r}_k \cdot \cos\left(\boldsymbol{\theta}_k + \boldsymbol{\phi}_k\right) \\ \mathbf{p}_{y,k} + \mathbf{r}_k \cdot \sin\left(\boldsymbol{\theta}_k + \boldsymbol{\phi}_k\right) \\ \boldsymbol{\theta}_k + \boldsymbol{\phi}_k \end{pmatrix}$$
(3.60)

We assume that the sensing unit makes a decision on transmission of information to the estimator every 10 seconds.

The Weibull random variable \mathbf{r}_k and Wrapped Cauchy random variable ϕ_k for the model have respective probability density functions given as follows:

$$f_{\mathbf{r}_k}(r) = \frac{a}{b} \left(\frac{r}{b}\right)^{a-1} e^{-(r/b)^a}, \text{ for } r \ge 0$$
 (3.61a)

$$f_{\phi_k}(\phi) = \frac{1}{2\pi} \cdot \frac{\sinh \gamma}{\cosh \gamma - \cos(\phi - \mu)}$$
(3.61b)

Using the collected GPS data, we compute the maximum likelihood estimates (MLE) of the parameters for (3.61):

$$(a, b) = (10.3214, 5.9553)$$

 $(\mu, \gamma) = (0.004, 0.001)$

Under these parameter choices, the probability density functions are depicted in Figure 3.4.

Transmission policies and estimation rules are determined based on Procedure 1 and Procedure 2 for the communication costs $c_k = 5$ for all k in $\{1, \dots, N\}$ with N = 78. Figure 3.5 shows the estimate $\begin{pmatrix} \hat{p}_{x,k} & \hat{p}_{y,k} & \hat{\theta}_k \end{pmatrix}^T$ of the original trajectory of the buffalo, depicted in Figure 3.3; and Figure 3.6 shows the estimation error computed by

$$d(x_k, \hat{x}_k) = \left\| \begin{pmatrix} \cos \theta_k & -\sin \theta_k & p_{x,k} \\ \sin \theta_k & \cos \theta_k & p_{y,k} \\ 0 & 0 & 1 \end{pmatrix} - \begin{pmatrix} \cos \hat{\theta}_k & -\sin \hat{\theta}_k & \hat{p}_{x,k} \\ \sin \hat{\theta}_k & \cos \hat{\theta}_k & \hat{p}_{y,k} \\ 0 & 0 & 1 \end{pmatrix} \right\|_F$$

Note that $d(x_k, \hat{x}_k) = 0$ at time k in Figure 3.6 implies that the sensing unit transmitted information on the full state x_k of the process to the estimator, and the state estimate \hat{x}_k at the estimator was set to the state of the process, i.e., $\hat{x}_k = x_k$.



(b) The probability density function of ϕ_k

Figure 3.4: Comparisons between the probability density functions of \mathbf{r}_k and ϕ_k under

the computed parameter choices and the GPS data.



Figure 3.5: Estimated trajectory of the water buffalo by the proposed remote estimation scheme



Figure 3.6: Estimation error of the remote estimation scheme

3.6 Summary and Future Work

We have studied the remote state estimation problem formulated in Section 3.1. To find a solution to the problem, we re-write the problem into N sub-problems and sequentially solve each Sub-problem k. We show that optimal solutions to all the sub-problems constitute an optimal solution to the original problem. Based on this idea, our main results show the existence of a jointly optimal solution, and describe an iterative procedure for finding a person-by-person optimal solution. In addition we have applied the proposed scheme to the experimental data obtained from the real-world deployment.

As future work, we will find the convergence rate of the proposed procedure described in Procedure 1, and if it exists, search for a new algorithm that achieves a faster convergence rate. Also we are interested in extending the presented results to large scale dynamical systems which may consist of multiple sensing units and estimators.

Chapter 4: Evolutionary Game Dynamics and Passivity

4.1 Background

4.1.1 Notation

• For a vector a in \mathbb{R}^n , its *i*-th entry is denoted by a_i , and we define

$$[a_i]_+ \stackrel{def}{=} \max\{a_i, 0\}$$
$$[a]_+ \stackrel{def}{=} \left([a_1]_+ \cdots [a_n]_+ \right)^T$$

- We denote the gradient and Hessian of a real-valued function x → f(x) with respect to x by ∇_xf and ∇²_xf, respectively, provided they exist.
- We denote the interior and the boundary of a set A by int(A) and bd(A), respectively.
- 1 is the vector with all entries 1, *I* is the identity matrix, and *e_i* is the *i*-th column of *I*.
- $\|\cdot\|$ is the Euclidean norm.
4.1.2 Population Games and Evolutionary Dynamics

Consider a population of players engaged in a game where each player selects a (pure) strategy from the set of available strategies represented by $\{1, \dots, n\}$.¹ Suppose that the population consists of a continuum of players. Population states, which describe the distributions of strategy choices by players, constitute a simplex

$$\mathbb{X} \stackrel{def}{=} \left\{ x \in \mathbb{R}^n_+ \left| \left| \sum_{i=1}^n x_i \right| = 1 \right\} \right\}$$

We denote the tangent space of \mathbb{X} as $T\mathbb{X} = \left\{ z \in \mathbb{R}^n \mid \sum_{i=1}^n z_i = 0 \right\}$. A payoff vector $p \in \mathbb{R}^n$ is assigned to each population state x: p_i represents a payoff given to the *i*-strategists, the players choosing strategy *i*. Based on this notation, we describe population games and evolutionary dynamics in Section 4.1.2.1 and Section 4.1.2.2, respectively.

4.1.2.1 Population Games

We identify population games with payoff operators defined as follows:

$$p(\cdot) = \mathfrak{G}x(\cdot) \tag{4.1}$$

 $\mathfrak{G}: \mathcal{X} \to \mathcal{P}$ is a causal operator where \mathcal{X} is the set of all differentiable X-valued timedependent functions $x(\cdot): \mathbb{R}_+ \to X$, and \mathcal{P} is the set of all differentiable \mathbb{R}^n -valued timedependent functions $p(\cdot): \mathbb{R}_+ \to \mathbb{R}^n$. Equation (4.1) suggests that a payoff trajectory $p(\cdot)$ is a causal function of a population state trajectory $x(\cdot)$; hence, under (4.1), the payoff p(t)

¹Population games, in general, account multiple populations of players, and the strategy sets are allowed to be distinct across populations. However, for simple and clear presentation, we restrict our attention to single-population games. Results for the single-population cases can be extended to multi-population cases.

at each time t may depend on the entire history of a population state. We note that this formalism of population games generalizes the conventional ones presented, for instance, in [4,95].

The following are a few examples of payoff operators (4.1).

Example 4.1.1. Let F be a C^1 mapping from \mathbb{X} to \mathbb{R}^n .

Time-Delayed Payoff:	p(t) = F(x(t-d))
Contrarian Effect Payoff [15]:	$p(t) = F(x(t)) - \Lambda(x(t) - x(t - d))$
Cumulative Payoff:	$\dot{p}(t) = F(x(t))$
Anticipatory Payoff [15]:	$\dot{q}(t) = \lambda(F(x(t)) - q(t))$
	p(t) = F(x(t)) + k(F(x(t)) - q(t))

where Λ is a matrix in $\mathbb{R}^{n \times n}$, d is a positive constant, and k, λ are real numbers.

We adopt the notion of Nash equilibrium in the following way.²

Definition 4.1.2 (Nash Equilibrium). Let $x \in \mathbb{X}$ be a population state and $p \in \mathbb{R}^n$ be a payoff vector assigned to it. The population state x is a Nash equilibrium if every strategy in use receives the maximum payoff, i.e., if x_i is positive then $p_i = \max_{j \in \{1, \dots, n\}} p_j$ holds.

4.1.2.2 Evolutionary Dynamics

Evolutionary dynamics describe how the population state evolves over time in response to a payoff trajectory. Throughout the chapter, we consider dynamics that can be

²When a game is described by a payoff function p = F(x), Definition 4.1.2 coincides with the conventional definition of Nash equilibrium.

represented by a differential equation³ given by

$$\dot{x} = V(p, x), \ x(0) = x_0 \in \mathbb{X}$$
(4.2)

where p(t), x(t), and $\dot{x}(t)$ take values in \mathbb{R}^n , \mathbb{X} , and $T\mathbb{X}$, respectively. We assume that the vector field $V : \mathbb{R}^n \times \mathbb{X} \to T\mathbb{X}$ is *well-defined* in a sense that for each initial value x_0 in \mathbb{X} and payoff trajectory $p(\cdot)$ in \mathcal{P} , there exists a unique solution $x(\cdot)$ to (4.2) that belongs to \mathcal{X} .

We define the set of equilibrium points of (4.2) as

$$\mathbb{S} \stackrel{def}{=} \left\{ (p, x) \in \mathbb{R}^n \times \mathbb{X} \, \big| \, V(p, x) = \mathbf{0} \right\}$$

and for each x in \mathbb{X} , its projection on $\mathbb{R}^n \times \{x\}$ as $\mathbb{S}_x \stackrel{def}{=} \{p \in \mathbb{R}^n \mid (p, x) \in \mathbb{S}\}$. We assume that for each x in \mathbb{X} , the set \mathbb{S}_x is path-connected, i.e., for every p_0, p_1 in \mathbb{S}_x , there exists a piece-wise smooth path from p_0 to p_1 . As a case in point, consider a set \mathbb{S}_{NE} that consists of (p, x) for which given the payoff p, the population state x is a Nash equilibrium. It can be verified that if (p_0, x) and (p_1, x) both belong to \mathbb{S}_{NE} then so does $(\lambda \cdot p_0 + (1 - \lambda) \cdot p_1, x)$ for all λ in [0, 1]. Hence \mathbb{S}_{NE} satisfies the path-connectedness assumption.

The following are a few examples of evolutionary dynamics that are found in literature.

Example 4.1.3. The replicator dynamics [9], BNN dynamics [8], Smith dynamics [96], and logit dynamics [97] are representative instances of evolutionary dynamics. The state-

³In the literature of evolutionary game theory, it is a convention to represent evolutionary dynamics as $\dot{x} = V(F(x), x)$ to make the dependence on a payoff function $F : \mathbb{X} \to \mathbb{R}^n$ explicit. While, in this work, we remove the dependence to study evolutionary dynamics under a generalized class of games (4.1).

space representations of these dynamics are given as follows:

Replicator:
$$\dot{x}_i = x_i \hat{p}_i$$
 (4.3)

BNN:
$$\dot{x}_i = [\hat{p}_i]_+ - x_i \sum_{j=1}^n [\hat{p}_j]_+$$
 (4.4)

Smith:
$$\dot{x}_i = \sum_{j=1}^n x_j [p_i - p_j]_+ - x_i \sum_{j=1}^n [p_j - p_i]_+$$
 (4.5)

Logit:
$$\dot{x}_i = \frac{\exp(\eta^{-1} \cdot p_i)}{\sum_{j=1}^n \exp(\eta^{-1} \cdot p_j)} - x_i$$
 (4.6)

for each *i* in $\{1, \dots, n\}$, where \hat{p} is the excess payoff vector defined as $\hat{p} = p - p^T x \cdot \mathbf{1}$, and η is a positive real number.

4.2 Passivity of Evolutionary Dynamics

We define a notion of passivity for evolutionary dynamics (4.2), and characterize passivity in terms of the vector field V in (4.2). Based on the characterization, we examine passivity of previously established dynamics, and investigate properties of passive dynamics.

4.2.1 Definition of Passivity for Evolutionary Dynamics

To define passivity for (4.2), let us consider the following inequality for a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ and a constant η :

$$S_{ED}(p(t), x(t)) \le S_{ED}(p(t_0), x(t_0)) + \int_{t_0}^t \left[\dot{p}^T(\tau) \dot{x}(\tau) - \eta \cdot \dot{x}^T(\tau) \dot{x}(\tau) \right] \, \mathrm{d}\tau \tag{4.7}$$

for $t \ge t_0 \ge 0$, where $x(\cdot) \in \mathcal{X}$ is the trajectory of the population state determined by (4.2) in response to a payoff trajectory $p(\cdot) \in \mathcal{P}$. In terms of (4.7), we state the definition of passivity as follows.

Definition 4.2.1. *Consider an evolutionary dynamic given as in* (4.2).

- 1. The dynamic is said to be passive if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which (4.7) holds with $\eta = 0$ for $t \ge t_0 \ge 0$ and every payoff trajectory $p(\cdot)$ in \mathcal{P} .
- 2. The dynamic is said to be strictly passive if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which (4.7) holds with $\eta = 0$ for $t \ge t_0 \ge 0$ and every payoff trajectory $p(\cdot)$ in \mathcal{P} , and if $\nabla_x^T S_{ED}(p, x) V(p, x) = 0$ implies V(p, x) = 0.
- 3. The dynamic is said to be strictly output passive if there exists a C^1 function S_{ED} : $\mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which (4.7) holds with $\eta > 0$ for $t \ge t_0 \ge 0$ and every payoff trajectory $p(\cdot)$ in \mathcal{P} .

We refer to S_{ED} as a *storage function* and (4.7) as the *passivity inequality*. Since S_{ED} is a non-negative function, without loss of generality, we assume that

$$\inf_{(p,x)\in\mathbb{R}^n\times\mathbb{X}}S_{ED}(p,x)=0$$

It follows from Definition 4.2.1 that strict output passivity entails strict passivity and strict passivity entails passivity.

Remark 4.2.2. The definition of passivity for evolutionary dynamics is closely related with the notion of dissipativity from dynamical system theory [98]. To see this, let us rewrite (4.2) in the following form:

$$\begin{pmatrix} \dot{p} \\ \dot{x} \end{pmatrix} = \begin{pmatrix} \mathbf{0} \\ V(p, x) \end{pmatrix} + \begin{pmatrix} u \\ \mathbf{0} \end{pmatrix}$$
(4.8a)

$$y = V(p, x) \tag{4.8b}$$

Note that (4.8) is a state-space representation of a control-affine nonlinear system with the input u, state (p, x), and output y. By the traditional notion of dissipativity [98], the system (4.8) is dissipative with respect to the supply rate $s(u, y) = u^T y - \eta \cdot y^T y$ if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which

$$S_{ED}(p(t), x(t)) \le S_{ED}(p(t_0), x(t_0)) + \int_{t_0}^t \left[u^T(\tau) y(\tau) - \eta \cdot y^T(\tau) y(\tau) \right] \, \mathrm{d}\tau \tag{4.9}$$

holds for $t \ge t_0 \ge 0$ and every real-valued function $u : \mathbb{R}_+ \to \mathbb{R}^n$. Then, by the equivalence between (4.2) and (4.8), we can verify that the passivity inequality (4.7) is satisfied for (4.2) if and only if the inequality (4.9) is satisfied for (4.8).

4.2.2 Characterization of Passivity of Evolutionary Dynamics

Let us consider the following two conditions:

$$\nabla_p S_{ED}(p,x) = V(p,x) \tag{P1}$$

$$\nabla_x^T S_{ED}(p, x) V(p, x) \le -\eta \cdot V^T(p, x) V(p, x)$$
(P2)

where $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ is a C^1 function, $V : \mathbb{R}^n \times \mathbb{X} \to T\mathbb{X}$ is the vector field given in (4.2), and η is a real number. In the following Theorem, we show that (**P1**) and (**P2**) are passivity requirements for evolutionary dynamics. This result not only provides an alternative definition of passivity but also is useful in studying properties of passive evolutionary dynamics.

Theorem 4.2.3. *Consider an evolutionary dynamic given as in* (4.2)*. The following statements are true:*

- (S1) The dynamic is passive if and only if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which the conditions (P1) and (P2) hold with $\eta = 0$.
- (S2) The dynamic is strictly passive if and only if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which the conditions (P1) and (P2) hold with $\eta = 0$, and the equality in (P2) holds only if V(p, x) = 0 holds.
- (S3) The dynamic is strictly output passive if and only if there exists a C^1 function S_{ED} : $\mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which the conditions (P1) and (P2) hold with $\eta > 0$.

Proof. To prove the Theorem, as noted in Remark 4.2.2, recall that (4.2) can be rewritten as in (4.8) and that passivity of evolutionary dynamics (4.2) is equivalent to dissipativity of control-affine nonlinear systems (4.8) with the supply rate $s(u, y) = u^T y - \eta \cdot y^T y$. Using dissipativity characterization Theorem (see, for instance, Theorem 1 in [99]), we can see that there exists a C^1 function S_{ED} for which the conditions (**P1**) and (**P2**) hold if and only if S_{ED} satisfies the passivity inequality (4.7) for $t \ge t_0 \ge 0$ and every payoff trajectory $p(\cdot)$ in \mathcal{P} . The rest of the proof follows from Definition 4.2.1 and the equivalence between passivity for (4.2) and dissipativity for (4.8).

Implications of the conditions (**P1**) and (**P2**) are as follows: For a fixed $x \in \mathbb{X}$, consider an integral $\int_P V(\mathbf{p}, x) \bullet d\mathbf{p}$ of the vector field V along a piece-wise smooth path P from p_0 to p_1 in the direction of \mathbf{p} . (**P1**) is equivalent to the fact that the value of the integral does not depend on the choice of the path.⁴ Dissatisfaction of (**P1**) could lead to a limit cycle and non-convergence to an equilibrium. (see Example 6.1 in [14]).

⁴A game-theoretic interpretation of (**P1**) is given in [100].

Next, suppose that the payoff vector p(t) is constant, i.e., $p(t) = p_0$ for all t in \mathbb{R}_+ . According to (**P2**), the population state x(t) evolves along a trajectory for which the stored energy quantified by $S_{ED}(p(t), x(t))$ decreases. In particular, if the dynamic (4.2) is strictly passive then we can establish asymptotic stability of the set S using LaSalle's theorem [101]. In Section 4.3, based on this observation, we establish stability of passive dynamics in a class of population games.

As an application of Theorem 4.2.3, we evaluate passivity of evolutionary dynamics found in literature. We start with the replicator dynamics.

Proposition 4.2.4. *The replicator dynamics* (4.3) *are not passive.*

The proof is given in Appendix C.1.

In what follows, we examine passivity of the EPT dynamics [102], (impartial) pairwise comparison dynamics [103], and PBR dynamics [97].

EPT Dynamics:

$$\dot{x}_i = \varrho_i(\hat{p}) - x_i \cdot \mathbf{1}^T \varrho(\hat{p}) \tag{4.10}$$

where the excess payoff vector is defined as $\hat{p} = p - p^T x \cdot \mathbf{1}$. The function $\varrho = \begin{pmatrix} \varrho_1 & \dots & \varrho_n \end{pmatrix}^T$ is called the *revision protocol* in which each entry is defined as $\varrho_i : \mathbb{R}^n \to \mathbb{R}_+$ and satisfies following two conditions – Integrability (**I**) and Acuteness (**A**):

$$\nabla_{\hat{p}}\gamma(\hat{p}) = \varrho(\hat{p}) \text{ for a } C^1 \text{ function } \gamma: \mathbb{R}^n \to \mathbb{R}$$
 (I)

$$\hat{p}^T \varrho(\hat{p}) > 0 \text{ if } \hat{p} \in \mathbb{R}^n \setminus \mathbb{R}^n_- \tag{A}$$

Proposition 4.2.5. The EPT dynamics (4.10) are strictly passive with a storage function⁵

$$S_{EPT}(p,x) = \gamma(\hat{p})$$

The proof is given in Appendix C.2.

Pairwise Comparison Dynamics:

$$\dot{x}_i = \sum_{j=1}^n x_j \varrho_i (p_i - p_j) - x_i \sum_{j=1}^n \varrho_j (p_j - p_i)$$
(4.11)

The function $\rho = \begin{pmatrix} \rho_1 & \dots & \rho_n \end{pmatrix}^T$ is called the *revision protocol* in which each entry is defined as $\rho_i : \mathbb{R} \to \mathbb{R}_+$ and satisfies the following condition – Sign Preservation (**SP**):⁶

$$\operatorname{sgn}\left(\varrho_i(p_i - p_j)\right) = \operatorname{sgn}\left(\left[p_i - p_j\right]_+\right)$$
(SP)

Proposition 4.2.6. [15] The pairwise comparison dynamics (4.11) are strictly passive with a storage function $S_{PC}(p, x) = \sum_{i=1}^{n} \sum_{j=1}^{n} x_i \int_{0}^{p_j - p_i} \varrho_j(s) ds.$

PBR Dynamics:

$$\dot{x} = C(p) - x \tag{4.12}$$

where $C : \mathbb{R}^n \to \mathbb{X}$ is defined as $C(p) = \arg \max_{y \in \operatorname{int}(\mathbb{X})} (p^T y - v(y))$, where v : $\operatorname{int}(\mathbb{X}) \to \mathbb{R}$ is a strictly convex C^2 function that satisfies $z^T \nabla_x^2 v(x) z > 0$ for all x in \mathbb{X} and z in $T\mathbb{X} \setminus \{\mathbf{0}\}$, and $\|\nabla_x v(x)\| \to \infty$ as $x \to \operatorname{bd}(\mathbb{X})$. We refer to such C as a *choice* function, and to such v as an *admissible (deterministic) perturbation*.

⁶sgn :
$$\mathbb{R} \to \{-1, 0, 1\}$$
 is defined as $\operatorname{sgn}(a) \stackrel{def}{=} \begin{cases} 1 & \text{if } a > 0 \\ 0 & \text{if } a = 0 \\ -1 & \text{if } a < 0 \end{cases}$

⁵In the proof of Proposition 4.2.5, we show that there is a non-negative potential function γ that satisfies both (I) and (A).

Proposition 4.2.7. The PBR dynamics (4.12) are strictly passive with a storage function

$$S_{PBR}(p,x) = \max_{y \in \operatorname{int}(\mathbb{X})} \left(p^T y - v(y) \right) - \left(p^T x - v(x) \right)$$
(4.13)

Suppose that v is strongly convex satisfying

$$z^T \nabla_x^2 v(x) z \ge \eta' \cdot z^T z$$

for all x in X and z in TX, where $\eta' > 0$. Then the dynamics are strictly output passive and satisfy the passivity inequality (4.7) for $\eta = \eta'$.

The proof is given in Appendix C.3.

4.2.3 Properties of Passive Evolutionary Dynamics

4.2.3.1 Payoff Monotonicity and Passivity

Using the characterization of passivity given in Theorem 4.2.3, we study properties of passive evolutionary dynamics in connection with the following two conditions⁷ – *Nash Stationarity* (**NS**) and *Positive Correlation* (**PC**):

$$V\left(p,x
ight) =\mathbf{0}$$
 if and only if

given the payoff p, the population state x is a Nash equilibrium (NS)

$$p^T V(p, x) \ge 0$$
 holds for all $(p, x) \in \mathbb{R}^n \times \mathbb{X}$ (PC)

Consider evolutionary dynamics (4.2) in a game in which the payoff is constant, i.e., $p(t) = p_0$ for all t in \mathbb{R}_+ . The conditions (**NS**) and (**PC**) have the following implications:

⁷These conditions are previously considered in literature to establish stability of evolutionary dynamics [14, 104].

(PC) implies that the population state trajectory $x(\cdot)$ determined by (4.2) evolves along which the average payoff $p_0^T x(t)$ is increasing, i.e., $\frac{d}{dt} p_0^T x(t) \ge 0$; and (NS) implies that the population state does not change if and only if the maximum average payoff is attained, i.e., $\dot{x}(t) = 0$ if and only if $p_0^T x(t) = \max_{x \in \mathbb{X}} p_0^T x$. We will refer to these phenomena as *payoff monotonicity*, and to the dynamics satisfying both (NS) and (PC) as *payoff monotonic*.

Proposition 4.2.8. Consider passive evolutionary dynamics (4.2) with a storage function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$. A global minimizer of S_{ED} is an equilibrium point of (4.2). In addition, if the dynamics satisfy (**NS**) then every equilibrium point of (4.2) is a global minimizer of S_{ED} .

The proof is given in Appendix C.4.

Remark 4.2.9. According to Proposition 4.2.8, given that $\min_{(p,x)\in\mathbb{R}^n\times\mathbb{X}} S_{ED}(p,x) = 0$, the inverse image $S_{ED}^{-1}(0)$ is a subset of the set \mathbb{S} of equilibrium points of (4.2), and it is identical to \mathbb{S} if the dynamics satisfy (**NS**).

Based on Definition 4.2.1, we note that strict output passivity is a stronger notion, and leads to stronger stability than does ordinary passivity in a sense that strictly output passive dynamics are stable in a larger class of population games. Hence, in what regards to achieving stability, it is desired to adopt strictly output passive dynamics. However, in the following Proposition, we show that the evolutionary dynamics exhibiting the payoff monotonicity cannot be strictly output passive. Therefore, the payoff monotonicity and strict output passivity cannot be attained simultaneously. **Proposition 4.2.10.** For $n \ge 3$, no payoff monotonic evolutionary dynamics are strictly output passive.

The proof is given in Appendix C.5.

The following is a direct consequence of Proposition 4.2.10.

Corollary 4.2.11. *The EPT dynamics* (4.10) *and the pairwise comparison dynamics* (4.11) *are not strictly output passive.*

As it can be verified that both dynamics are payoff monotonic, the proof directly follows from Proposition 4.2.10.

4.2.3.2 Equivalence to Closed-loop Stability

In Definition 4.2.1, we defined passivity as an input-output property of evolutionary dynamics: Satisfaction of the passivity inequality (4.7) for each payoff trajectory (input) $p(\cdot)$ in \mathcal{P} and the population state trajectory (output) $x(\cdot)$ determined by (4.2) in response to $p(\cdot)$. In evolutionary game theory, the main interest lies in examining the time-evolution of the population state induced in specific games. Hence, it is more natural to define passivity of evolutionary dynamics in connection with games of interest. To achieve this, let us consider population games identified by cumulative payoff functions of the following form:

$$\dot{p}(t) = F(x(t)) \tag{4.14}$$

where F admits a C^1 potential function, i.e., there is a C^1 function $f : \mathbb{X} \to \mathbb{R}$ that satisfies $\nabla_x f = \Phi F$ where $\Phi = I - \frac{1}{n} \mathbf{1} \mathbf{1}^T$. Then, with a C^1 function defined by

$$S_G(x) = \max_{x \in \mathbb{X}} f(x) - f(x) \tag{4.15}$$

we can derive that

$$S_{G}(x(t)) - S_{G}(x(t_{0})) = -\int_{t_{0}}^{t} \frac{\mathrm{d}}{\mathrm{d}\tau} f(x(\tau)) \,\mathrm{d}\tau$$
$$= -\int_{t_{0}}^{t} \dot{p}^{T}(\tau) \dot{x}(\tau) \,\mathrm{d}\tau \qquad (4.16)$$

In the following Proposition, we show that passivity can be defined as (a weak form of) stability of closed-loops formed by cumulative payoff functions (4.14) and evolutionary dynamics. This result implies that passivity of evolutionary dynamics is equivalent to satisfaction of the passivity inequality (4.7) in the class of population games identified by (4.14).

Proposition 4.2.12. *Consider the following closed loop formed by a cumulative payoff function* (4.14) *and an evolutionary dynamic* (4.2):

$$\dot{p} = F(x) \tag{4.17a}$$

$$\dot{x} = V(p, x) \tag{4.17b}$$

The dynamic (4.2) is passive if and only if for each cumulative payoff function (4.14), the closed-loop (4.17) has a Lyapunov function $E : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ of the form $E = S_G + E_{ED}$, where $S_G : \mathbb{X} \to \mathbb{R}_+$ is given in (4.15) and $E_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ is a fixed C^1 function.

The proof is given in Appendix C.6.

The following is a direct consequence of Proposition 4.2.12, where the proof is given in Corollary C.7.

Corollary 4.2.13. An evolutionary dynamic (4.2) is passive if and only if there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which the passivity inequality (4.7) holds for $\eta = 0$ in the class of population games identified by cumulative payoff functions (4.14), i.e., there exists a C^1 function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which

$$S_{ED}(p(t), x(t)) \le S_{ED}(p(t_0), x(t_0)) + \int_{t_0}^t F^T(x(\tau))\dot{x}(\tau) \,\mathrm{d}\tau$$

holds for every function $F : \mathbb{X} \to \mathbb{R}^n$ that admits a C^1 potential function.

4.2.3.3 Effect of Control Costs on Passivity

Consider the *total payoff function* $u : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}$ given by

$$u(p,x) = p^T x - v(x)$$
 (4.18)

where a C^2 function $v : \mathbb{X} \to \mathbb{R}$ is referred to as a *control cost* [105] or a *deterministic perturbation* [106]. A control cost is said to be admissible if it is strictly convex satisfying $z^T \nabla_x^2 v(x) z > 0$ for all x in \mathbb{X} and z in $T\mathbb{X} \setminus \{\mathbf{0}\}$, and $\|\nabla_x v(x)\| \to \infty$ as $x \to bd(\mathbb{X})$. Notice that when there is no control cost, i.e., v = 0, the total payoff coincides with the average payoff. The idea of imposing control costs on the total payoff appeared in game theory and economics to study the effect of random perturbations [97, 106] or disutility [105] on choice models, to model human choice behavior [107], and to analyze the effect of social norms in economic problems [108]

We consider evolutionary dynamics that depend on u, and investigate the effect of control costs on passivity of the dynamics. We refer to the dynamics as *unperturbed* if v = 0; otherwise they are called *perturbed*. To proceed, let us consider the state-space

representation of evolutionary dynamics in terms of revision protocols [14]: For each i in $\{1, \dots, n\}$,

$$\dot{x}_i = \sum_{j=1}^n x_j \varrho_{ji}(p, x) - x_i \sum_{j=1}^n \varrho_{ij}(p, x)$$
(4.19)

where $\rho_{ji} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ is called the revision protocol and denotes the rate at which *j*-strategists switch to strategy *i* given a payoff *p* and population state *x*.

For instance, the revision protocols can be realized as follows:

$$\varrho_{ji}(p,x) = \left[\nabla_x^T u(p,x) \left(e_i - x\right)\right]_+$$
(4.20a)

$$\varrho_{ji}(p,x) = \left[\nabla_x^T u(p,x) \left(e_i - e_j\right)\right]_+$$
(4.20b)

The protocol (4.20a) depends on the (instantaneous) increase of u(p, x) when the population state changes in the direction of $e_i - x$; and the protocol (4.20b) depends on the (instantaneous) increase of u(p, x) when the population state changes in the direction of $e_i - e_j$. Based on the revision protocols (4.20a) and (4.20b), we can derive the following evolutionary dynamics: For each i in $\{1, \dots, n\}$,

$$\dot{x}_{i} = \left[\nabla_{x}^{T} u(p, x) \left(e_{i} - x\right)\right]_{+} - x_{i} \sum_{j=1}^{n} \left[\nabla_{x}^{T} u(p, x) \left(e_{j} - x\right)\right]_{+}$$
(4.21a)

$$\dot{x}_{i} = \sum_{j=1}^{n} x_{j} \left[\nabla_{x}^{T} u(p, x) \left(e_{i} - e_{j} \right) \right]_{+} - x_{i} \sum_{j=1}^{n} \left[\nabla_{x}^{T} u(p, x) \left(e_{j} - e_{i} \right) \right]_{+}$$
(4.21b)

Note that when no control cost is imposed, i.e., v = 0, (4.21a) and (4.21b) become the state-space representations of the BNN dynamics (4.4) and Smith dynamics (4.5), respectively. According to Proposition 4.2.5 and Proposition 4.2.6, the unperturbed dynamics of (4.21a) and (4.21b) are strictly passive.

Suppose that the control costs are strongly convex. Then, it can be verified that the resulting perturbed dynamics of (4.21a) and (4.21b) are strictly output passive. In

what follows, we formalize this idea and show how convexity of control costs affects passivity of evolutionary dynamics. For our purpose, we consider evolutionary dynamics (4.19) whose revision protocols depend only on the gradient $\nabla_x u(p, x)$ of the total payoff function (4.18) and the population state x, i.e., ρ_{ji} is a function of $\nabla_x u(p, x)$ and x as in (4.20).

Proposition 4.2.14. Consider evolutionary dynamics (4.19) whose revision protocols ϱ_{ji} depend only on the gradient $\nabla_x u(p, x)$ of the total payoff function (4.18) and the population state x. Suppose that the unperturbed dynamics of (4.19) are passive. Then the following are true:

- 1. If the control cost is admissible then the resulting perturbed dynamics are strictly passive.
- 2. If the control cost is admissible and strongly convex then the resulting perturbed dynamics are strictly output passive.

The proof is given in Proposition C.8.

4.3 Stability of Passive Evolutionary Dynamics

In this section, we establish stability of passive evolutionary dynamics in population games in terms of dissipation of stored energy of the dynamics. To achieve this, we regard evolutionary dynamics and payoff operators as dynamical systems, and we investigate population state and payoff trajectories induced by a closed-loop formed by these dynamical systems (see Figure 4.1 for an illustration).



Figure 4.1: A closed-loop obtained by a feedback interconnection of payoff operators (4.1) and evolutionary dynamics (4.2).

Consider population games identified by (4.1) that satisfy the following inequality for some positive α : ⁸

$$\int_0^t \left[\dot{p}^T(\tau) \dot{x}(\tau) - \nu \cdot \dot{x}^T(\tau) \dot{x}(\tau) \right] \, \mathrm{d}\tau \le \alpha \tag{4.22}$$

for every population state trajectory $x(\cdot)$ in \mathcal{X} and t in \mathbb{R}_+ , where ν is non-negative real number and $p(\cdot) \in \mathcal{P}$ is a payoff trajectory determined by (4.1) in response to $x(\cdot)$. We represent the closed loop formed by (4.1) and (4.2) as follows:

$$p(\cdot) = \mathfrak{G}x(\cdot) \tag{4.23a}$$

$$\dot{x}(t) = V(p(t), x(t)) \tag{4.23b}$$

In the following Theorem, we state stability results for the closed-loop described by (4.23) in which

(CL1) (4.22) holds for $\nu = 0$, and (4.23b) is strictly passive.

⁸In [15], population games (4.1) satisfying (4.22) for $\nu = 0$ are called δ -anti-passive.

(CL2) (4.22) holds for $\nu > 0$, and (4.23b) is strictly output passive for a constant η satisfying $\eta > \nu$.

Theorem 4.3.1. Consider (4.23) in which (CL1) or (CL2) holds. Let $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ be a storage function of (4.23b). Suppose that the following assumptions hold: For any sequence $\{(p^{(l)}, x^{(l)})\}_{l \in \mathbb{N}}$ in $\mathbb{R}^n \times \mathbb{X}$,

(A1)
$$\|V(p^{(l)}, x^{(l)})\| \xrightarrow{l \to \infty} \infty$$
 implies $S_{ED}(p^{(l)}, x^{(l)}) \xrightarrow{l \to \infty} \infty$

(A2)
$$\nabla_x^T S_{ED}\left(p^{(l)}, x^{(l)}\right) V\left(p^{(l)}, x^{(l)}\right) \xrightarrow{l \to \infty} 0 \text{ implies } S_{ED}\left(p^{(l)}, x^{(l)}\right) \xrightarrow{l \to \infty} 0$$

If the time-derivative \dot{p} of the payoff is bounded, i.e, there is a positive real M for which $\|\dot{p}(t)\| < M$ holds for all t in \mathbb{R}_+ , then it holds that $\lim_{t\to\infty} S_{ED}(p(t), x(t)) = 0$.

The proof is given in Appendix C.9.

We note that the class of cumulative payoff functions presented in Section 4.2.3.2 satisfy (4.22). In the following Proposition, we present another class of payoff operators that satisfy (4.22).

Proposition 4.3.2. For C^1 mappings F_1 and F_2 , consider a (time-delayed) payoff function given by

$$p(t) = F_1(x(t)) + F_2(x(t-d))$$
(4.24)

where *d* is a positive real.⁹ Suppose that

$$z^T D F_1(x) z \le \nu_1 \cdot z^T z \tag{4.25a}$$

$$z^{T}DF_{2}^{T}(x)DF_{2}(x)z \le \nu_{2}^{2} \cdot z^{T}z$$
 (4.25b)

⁹Here we assume that x(t) for t in [-d, 0) satisfies $\int_{-d}^{0} \|\dot{x}(\tau)\|^2 d\tau < \beta$ for some positive real number

hold for all x in X and z in TX, where ν_1 is a real and ν_2 is a non-negative real. Then, the payoff function (4.24) satisfies (4.22) with $\nu = \nu_1 + \nu_2$.

The proof is given in Appendix C.10.

Based on Theorem 4.3.1, in what follows, we present stability results with the EPT dynamics (4.10), pairwise comparison dynamics (4.11), and PBR dynamics (4.12). As the EPT dynamics and pairwise comparison dynamics are at most strictly passive, we only consider (**CL1**) for these dynamics.

Proposition 4.3.3. Consider (4.23) with the EPT dynamics (4.10) in which (CL1) holds. Let $S_{EPT} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ be a storage function of (4.10). Suppose that the revision protocol $\varrho : \mathbb{R}^n \to \mathbb{R}^n_+$ of (4.10) satisfies the following conditions: For any sequence $\{\hat{p}^{(l)}\}_{l\in\mathbb{N}}$ in \mathbb{R}^n ,

- (C1) $\|\varrho\left(\hat{p}^{(l)}\right)\| \xrightarrow{l \to \infty} \infty$ implies $S\left(\hat{p}^{(l)}\right) \xrightarrow{l \to \infty} \infty$.
- (C2) $\|\varrho\left(\hat{p}^{(l)}\right)\| \stackrel{l \to \infty}{\longrightarrow} 0 \text{ implies } \left(\hat{p}^{(l)}\right)^T \varrho\left(\hat{p}^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0, \text{ and}$ $\left(\hat{p}^{(l)}\right)^T \varrho\left(\hat{p}^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0 \text{ implies } S\left(\hat{p}^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0.$

where $S(\hat{p}) = \int_{P} \varrho(\hat{\mathbf{p}}) \bullet d\hat{\mathbf{p}}$ is an integral of ϱ along a piece-wise smooth path P from **0** to \hat{p} in the direction of $\hat{\mathbf{p}}$.¹⁰ If the time-derivative \dot{p} of the payoff is bounded, then it holds that $\lim_{t\to\infty} S_{EPT}(p(t), x(t)) = 0$.

The proof is given in Appendix C.11.

Example 4.3.4. The BNN dynamics (4.4) are the EPT dynamics with a revision protocol given by $\rho(\hat{p}) = [\hat{p}]_+$ and a storage function given by $S_{EPT}(p, x) = \frac{1}{2} ||[\hat{p}]_+||^2$. Notice

¹⁰Note that by the condition (I) of the EPT dynamics, $\int_P \rho(\mathbf{p}) \bullet d\mathbf{p}$ does not depend on the choice of P.

that

$$S(\hat{p}) = \int_0^1 s \cdot \left\| [\hat{p}]_+ \right\|^2 \, \mathrm{d}s = \frac{1}{2} \left\| [\hat{p}]_+ \right\|^2 = \frac{1}{2} \left\| \varrho\left(\hat{p}\right) \right\|^2 \tag{4.26}$$

and

$$\hat{p}^{T}\varrho\left(\hat{p}\right) = \left\|\varrho\left(\hat{p}\right)\right\|^{2} \tag{4.27}$$

It follows from (4.26) and (4.27) that (C1) and (C2) of Proposition 4.3.3 hold. We also note that $\lim_{t\to\infty} S_{EPT}(p(t), x(t)) = 0$ implies $\lim_{t\to\infty} ||[\hat{p}(t)]_+|| = 0$. Notice that given a payoff p, a population state x is a Nash equilibrium if and only if the excess payoff $\hat{p} = p - p^T x \cdot \mathbf{1}$ satisfies $||[\hat{p}]_+|| = 0$. Hence the convergence of S_{EPT} implies that the population state trajectory converges to a set of Nash equilibria.

Proposition 4.3.5. Consider (4.23) with the pairwise comparison dynamics (4.11) in which (CL1) holds. Let $S_{PC} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ be a storage function of (4.11). Suppose that the revision protocol $\varrho = \left(\varrho_1 \ \cdots \ \varrho_n \right)^T$ of (4.11) satisfies the following conditions for each i, j in $\{1, \dots, n\}$: For any sequence $\{(p^{(l)}, x^{(l)})\}_{l \in \mathbb{N}}$ in $\mathbb{R}^n \times \mathbb{X}$,

(C1)
$$x_i^{(l)} \varrho_j \left(p_j^{(l)} - p_i^{(l)} \right) \xrightarrow{l \to \infty} \infty$$
 implies $S_j \left(p_j^{(l)} - p_i^{(l)}, x_i^{(l)} \right) \xrightarrow{l \to \infty} \infty$
(C2) $\varrho_j \left(p_j^{(l)} - p_i^{(l)} \right) \xrightarrow{l \to \infty} 0$ implies $S_j \left(p_j^{(l)} - p_i^{(l)}, x_i^{(l)} \right) \xrightarrow{l \to \infty} 0.$

where $S_j(p_j - p_i, x_i) = x_i \int_0^{p_j - p_i} \varrho_j(s) ds$. If the time-derivative \dot{p} of the payoff is bounded, then it holds that $\lim_{t\to\infty} S_{PC}(p(t), x(t)) = 0$.

The proof is given in Appendix C.12.

Example 4.3.6. The Smith dynamics (4.5) are the pairwise comparison dynamics (4.11) with a revision protocol given by $\varrho_j (p_j - p_i) = [p_j - p_i]_+$ and a storage function given

by $S_{PC}(p, x) = \frac{1}{2} \sum_{i=1}^{n} \sum_{j=1}^{n} x_i [p_j - p_i]_+^2$. We can derive the following:

$$S_j(p_j - p_i, x_i) = x_i \int_0^{p_j - p_i} [s]_+ \, \mathrm{d}s$$
(4.28a)

$$=\frac{1}{2}x_i[p_j - p_i]_+^2$$
(4.28b)

$$=\frac{1}{2}x_i\varrho_j^2(p_j-p_i) \tag{4.28c}$$

$$\geq \frac{1}{2} \left[x_i \varrho_j (p_j - p_i) \right]^2 \tag{4.28d}$$

It follows from (4.28c) and (4.28d) that (C1) and (C2) of Proposition 4.3.5 hold. By the fact that $\sum_{i=1}^{n} x_i [p_j - p_i]_+ \ge [\hat{p}_j]_+$, we note that $\lim_{t\to\infty} S_{PC}(p(t), x(t)) = 0$ implies $\lim_{t\to\infty} ||[\hat{p}(t)]_+|| = 0$. Hence, the convergence of S_{PC} implies that the population state trajectory converges to a set of Nash equilibria.

Proposition 4.3.7. Consider (4.23) with the PBR dynamics (4.12) in which (CL1) or (CL2) holds. Let $S_{PBR} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ be a storage function of (4.12). If the timederivative \dot{p} of the payoff is bounded, then it holds that $\lim_{t\to\infty} S_{PBR}(p(t), x(t)) = 0$.

The proof is given in Appendix C.13.

Remark 4.3.8. Due to (C.6b), by letting y = C(p), it holds that $[p - \nabla_x v(C(p))]^T z = 0$ for all z in TX, where C is the choice function of the PBR dynamics (4.12). Then, we can see that the storage function (4.13) satisfies

$$S_{PBR}(p,x) = \nabla_x^T v(y) (y-x) - (v(y) - v(x))$$

By strict convexity of v *and by the fact that* $y \in X$ *, we can see that*

$$\lim_{t \to \infty} S_{PBR}(p(t), x(t)) = 0 \text{ implies } \lim_{t \to \infty} \|C(p(t)) - x(t)\| = 0$$

i.e., the population state trajectory converges to the best response choice C(p).

4.4 Numerical Examples

To illustrate the main results, we provide numerical examples and simulations. We consider two different types of examples. In the first example, we consider the replicator dynamics and BNN dynamics in population games identified by a cumulative payoff function. In the second example, we consider the BNN dynamics and logit dynamics in the Hypnodisk game [104], which is used in the proof of Proposition 4.2.10. Simulation results will show that the population state trajectories induced by the BNN dynamics converge to a limit cycle; while the trajectories induced by the logit dynamics converge to an equilibrium point. In addition, we examine the case in which the BNN dynamics are perturbed by the control cost $v(x) = \eta \cdot \sum_{i=1}^{3} x_i \ln x_i$ as in (4.21a). Simulation results will show that the perturbed BNN dynamics have strong stability properties than the unperturbed ones.

4.4.1 Replicator dynamics and BNN dynamics under a cumulative payoff function

Consider a cumulative payoff function given by

$$\begin{pmatrix} \dot{p}_{1}(t) \\ \dot{p}_{2}(t) \\ \dot{p}_{3}(t) \end{pmatrix} = \begin{pmatrix} -x_{1}(t) + \frac{1}{3} \\ -x_{2}(t) + \frac{1}{3} \\ -x_{3}(t) + \frac{1}{3} \end{pmatrix}$$
(4.29)

We note that under (4.29), both the replicator dynamics (4.3) and BNN dynamics (4.4) are stationary at each element of the set given by

$$\left\{ (p,x) \in \mathbb{R}^3 \times \mathbb{X} \mid p_1 = p_2 = p_3 \text{ and } x_1 = x_2 = x_3 = \frac{1}{3} \right\}$$
(4.30)

The population state trajectories induced by these dynamics under (4.29) are illustrated in Figure 4.2 and Figure 4.3, respectively. From the illustrations, we can observe that for the replicator dynamics, the Kullback-Leibler (KL) divergence between the population state x(t) and the equilibrium $\frac{1}{3} \cdot 1$ does not converge to zero; while the storage function of the BNN dynamics converges to zero, which implies that the population state trajectory converges to the equilibrium $\frac{1}{3} \cdot 1$.



(a) Population state trajectory on simplex $\ensuremath{\mathbb{X}}$



(b) Time-evolution of the KL divergence E(x(t)) along the trajectory starting from $(p_0, x_0) = (-1, 0, 1, 0.2, 0.2, 0.6)$

Figure 4.2: Simulation results for the replicator dynamics under a cumulative payoff given by (4.29). $E(x) = \sum_{i=1}^{3} x_i^* \ln \frac{x_i^*}{x}$ where $x^* = \frac{1}{3} \cdot \mathbf{1}$



(a) Population state trajectory on simplex \mathbb{X}



(b) Time-evolution of storage function $S_{BNN}(p(t), x(t))$ along the trajectory starting from $(p_0, x_0) = (-1, 0, 1, 0.2, 0.2, 0.6)$

Figure 4.3: Simulation results for the BNN dynamics under a cumulative payoff given by (4.29). $S_{BNN}(p, x) = \frac{1}{2} \sum_{i=1}^{3} [\hat{p}_i]_+^2$

4.4.2 BNN dynamics and logit dynamics in the Hypnodisk game

Consider the Hypnodisk game whose payoff function is given by

$$\begin{pmatrix} p_{1}(t) \\ p_{2}(t) \\ p_{3}(t) \end{pmatrix} = H \begin{pmatrix} x_{1}(t) \\ x_{2}(t) \\ x_{3}(t) \end{pmatrix}$$

$$= \cos\left(\theta(x_{1}(t), x_{2}(t), x_{3}(t))\right) \begin{pmatrix} x_{1}(t) - \frac{1}{3} \\ x_{2}(t) - \frac{1}{3} \\ x_{3}(t) - \frac{1}{3} \end{pmatrix}$$

$$+ \frac{\sqrt{3}}{3} \sin\left(\theta(x_{1}(t), x_{2}(t), x_{3}(t))\right) \begin{pmatrix} x_{2}(t) - x_{3}(t) \\ x_{3}(t) - x_{1}(t) \\ x_{1}(t) - x_{2}(t) \end{pmatrix} + \frac{1}{3} \begin{pmatrix} 1 \\ 1 \\ 1 \\ 1 \end{pmatrix}$$
(4.31)

where θ is a smooth function as described in the proof of Proposition 4.2.10.

We note that under (4.31), both the BNN dynamics (4.4) and logit dynamics (4.6) have a unique equilibrium point at $(p, x) = (\frac{1}{3} \cdot \mathbf{1}, \frac{1}{3} \cdot \mathbf{1})$. The population state trajectories induced by these dynamics under (4.31) are illustrated in Figure 4.4 and Figure 4.5, respectively. From the illustrations, we can observe that the logit dynamics have a stronger stability property than do the BNN dynamics as the population state trajectory of the former converges to the equilibrium $x = \frac{1}{3} \cdot \mathbf{1}$; while that of the latter dynamics forms a limit cycle.

Now consider that a control cost given by $v(x) = \eta \cdot \sum_{i=1}^{3} x_i \ln x_i$ is imposed on

the BNN dynamics as in (4.21a). The resulting perturbed dynamics are represented by

$$\dot{x}_{i} = \left[\tilde{p}_{i} - \tilde{p}^{T}x\right]_{+} - x_{i}\sum_{j=1}^{3}\left[\tilde{p}_{j} - \tilde{p}^{T}x\right]_{+}$$
(4.32)

where $\tilde{p} = p - \nabla_x v$. Since v is a strongly convex function, according to Proposition 4.2.14, we can see that (4.32) is strictly output passive. The population state trajectory induced by (4.32) under the payoff function (4.31) is depicted in Figure 4.6, which shows that the trajectory converges to equilibrium $x = \frac{1}{3} \cdot \mathbf{1}$.



(a) Population state trajectory on simplex \mathbb{X}



(b) Time evolution of storage function $S_{BNN}(p(t), x(t))$ along the trajectory starting from $p_0 = H(x_0), x_0 = (0.5203, 0.3394, 0.1403)$

Figure 4.4: Simulation results for the BNN dynamics in the Hypnodisk game (4.31). $S_{BNN}(p,x) = \frac{1}{2} \sum_{i=1}^{3} [\hat{p}_i]_+$



(a) Population state trajectory on simplex X



(b) Time evolution of storage function $S_{logit}(p(t), x(t))$ along the trajectory starting from $p_0 = H(x_0), x_0 = (0.4677, 0.3709, 0.1614)$

Figure 4.5: Simulation results for the logit dynamics $(\eta' = 0.36)$ in the Hypnodisk game (4.31). $S_{logit}(p, x) = \max_{y \in int(\mathbb{X})} \left[p^T y - \eta' \cdot \sum_{i=1}^3 y_i \ln y_i \right] - p^T x + \eta' \cdot \sum_{i=1}^3 x_i \ln x_i$



(a) Population state trajectory on simplex \mathbb{X}



(b) Time evolution of storage function $S_{PBNN}(p(t), x(t))$ along the trajectory starting from $p_0 = H(x_0), x_0 = (0.4269, 0.3876, 0.1855)$

Figure 4.6: Simulation results for the perturbed BNN dynamics $(\eta' = 0.36)$ in the Hypnodisk game (4.31). $S_{PBNN}(p, x) = \frac{1}{2} \sum_{i=1}^{3} \left[\tilde{p}_i - \tilde{p}^T x \right]_+$

4.5 Summary and Future Work

We have exploited the notion of passivity in evolutionary game theory. We have defined passivity for evolutionary dynamics and characterized it in terms of vector fields that define the state-space realizations of the dynamics. Based on the characterization, we have studied certain properties of passive dynamics and established stability in a generalized class of population games. Numerical simulations are provided to illustrate the stability results.

To benefit from the presented passivity methods, as a future direction, we suggest to investigate the following design problems: How to design passive evolutionary dynamics whose storage function achieves its minimum at desired states; while maintaining their stability in population games of interest.

Appendix A: Auxiliary Results for Chapter 2

A.1 Computational Considerations

We proceed to outlining how to find the source components of a directed graph and how to compute an omniscience-achieving parameter choice for (2.1) that satisfies (1.2), provided that the conditions of Theorem 2.2.2 hold.

A.1.1 Finding Source Components

In Chapter 22.5 of [109], the Strongly-Connected Components (SCC) algorithm is described for finding all strongly connected components of a directed graph. For each strongly connected component given by the SCC algorithm, we can check whether there are no incoming edges from outside of it, in which case it is a source component. Since both the SCC algorithm and subsequent checks have linear-time complexity, the overall procedure for finding source components is a linear-time algorithm.

A.1.2 Computing an Omniscience-achieving Parameter Choice

We proceed to describing randomized procedures to compute a parameter choice for (2.1). Under the detectability condition of Theorem 2.2.2, it follows from our analysis in Section 2.4.3 that the parameter choice obtained from the following randomized procedures satisfies (1.2) and is omniscience-achieving with probability one.

A.1.2.1 Computation of $\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$

Let $\{\mathcal{G}_l\}_{l=1}^{m_s}$ with $\mathcal{G}_l = (\mathbb{V}_l, \mathbb{E}_l)$ be the source components of \mathcal{G} , and let us define $\mathbb{V}_{m_s+1} = \mathbb{V} \setminus \bigcup_{l=1}^{m_s} \mathbb{V}_l$. We first find a spanning subgraph \mathcal{G}' of \mathcal{G} for which $\{\mathcal{G}_l\}_{l=1}^{m_s}$ are the source components of \mathcal{G}' , and the subgraph of \mathcal{G}' induced¹ by \mathbb{V}_{m_s+1} has no cycle. Then we select a weight matrix \mathbf{W} whose sparsity pattern is consistent with \mathcal{G}' and, hence, with \mathcal{G} .

In order to obtain \mathcal{G}' , we perform multiple rounds of the depth-first search on \mathcal{G} where each round of the search starts from a (unvisited) vertex in \mathbb{V}_{m_s+1} that is a neighbor of a source component. We continue this search until every vertex in \mathbb{V}_{m_s+1} is visited exactly once. The overall search operation gives a disjoint collection of directed trees. Next, we eliminate certain edges of \mathcal{G} to obtain a new graph \mathcal{G}' such that \mathcal{G}' is same as \mathcal{G} except that the subgraph of \mathcal{G}' induced by \mathbb{V}_{m_s+1} is the union of the trees obtained from the aforementioned search operation. By our construction of \mathcal{G}' , it can be verified that $\{\mathcal{G}_l\}_{l=1}^{m_s}$ are the source components of \mathcal{G}' , and the subgraph of \mathcal{G}' induced by \mathbb{V}_{m_s+1} has no cycle.

For each i in \mathbb{V} , let \mathbb{N}'_i be the neighborhood of vertex i defined by \mathcal{G}' . We choose the submatrices of \mathbf{W} in (2.11) as described by the following randomized procedure (Procedure 3). Here, we use $q \sim U(a, b)$ to denote a randomization in which q is the realization of a random variable uniformly distributed in the interval (a, b). We assume that the ran-

 $^{^{1}(\}mathbb{V}',\mathbb{E}')$ is a subgraph of (\mathbb{V},\mathbb{E}) induced by $\mathbb{V}' \subseteq \mathbb{V}$ if \mathbb{E}' contains every edge (i,j) in \mathbb{E} with $i,j \in \mathbb{V}'$.

domizations presented in the procedure are drawn from independent random variables.

Procedure 3: Computation of $\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$

input : $\mathcal{G}' = (\mathbb{V}, \mathbb{E}')$ output: $\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}$ 1 begin for $l \in \{1, \cdots, m_s\}$ do 2 /* computation of \mathbf{W}_l in (2.11) */ $\alpha \sim U\left(0,1\right)$ 3 for $i \in \mathbb{V}_l$ do 4 for $j \in \mathbb{V}_l \setminus \{i\}$ do 5 if $j \in \mathbb{N}'_i \setminus \{i\}$ then 6 $\left| \begin{array}{c} q \sim U\left(-\frac{1}{|\mathbb{N}'_i|-1}, 0\right) \\ \mathbf{w}_{ij} \leftarrow -\alpha \cdot q \\ \\ \textbf{else } \mathbf{w}_{ij} \leftarrow 0 \\ \\ \mathbf{w}_{ii} \leftarrow 1 - \sum_{j \in \mathbb{N}'_i \setminus \{i\}} \mathbf{w}_{ij} \end{array} \right|$ 7 8 9 10 /* computation of $\{\mathbf{W}_{m_s+1,l}\}_{l=1}^{m_s+1}$ in (2.11) */ 11 for $i \in \mathbb{V}_{m_s+1}$ do for $j \in \mathbb{V}$ do 12 if $j \in \mathbb{N}'_i \setminus \{i\}$ then $\mathbf{w}_{ij} \leftarrow rac{1}{|\mathbb{N}'_i| - 1}$ 13 else $\mathbf{w}_{ij} \leftarrow 0$ 14

It follows from Lemmas 2.4.3 - 2.4.4 and Theorem 2.4.5 that Procedure 3 will select a weight matrix **W** randomly from a parametrized set for which almost all parameters lead to a suitable choice, with the possible exception of a subset of measure zero. Our particular choice for the distributions governing the randomizations in Procedure 3 is not important, and any other choice that assigns null probability to a subset of measure zero would work.

A.1.2.2 Computation of $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$

In what follows, we describe a randomized method (Procedure 4) to choose gain matrices $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$ that, in conjunction with W obtained from Procedure 3, are omniscience-achieving with probability one, provided that the conditions of Theorem 2.2.2 are satisfied. In fact, it follows from Theorem 2.4.10 that if the conditions of Theorem 2.2.2 are satisfied then the procedure will be selecting from a set in which almost all choices are omniscience-achieving.

Given a positive real c, we use $\mathbf{K} \sim U_{n \cdot r} \left((-c, c)^{n \cdot r} \right)$ to denote a randomization leading to a matrix \mathbf{K} in $\mathbb{R}^{n \times r}$ whose entries are the realizations of $n \cdot r$ independent random variables uniformly distributed in the interval (-c, c). For each i in \mathbb{V}_l , let $\overline{B}_i = e_i \otimes I_n$ and $\overline{C}_i = e_i^T \otimes C_i$, where e_i is the *i*-th column of the $|\mathbb{V}_l|$ -dimensional identity matrix. We choose $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$ as described below, where repeated randomizations are drawn from independent random variables.
Procedure 4: Computation of $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$

input : $\mathcal{G}' = (\mathbb{V}, \mathbb{E}')$, W given as in (2.11), and (A, C) given as in (1.1) output: $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$ 1 begin select \mathbb{V}^R as in Definition 2.2.1 2 for $l \in \{1, \dots, m_s\}$ do 3 /* computation of $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}_l}$ $i_l \in \mathbb{V}_l \cap \mathbb{V}^R$ */ 4 // a singleton for $i \in \mathbb{V}_l \setminus \{i_l\}$ do $\begin{bmatrix} \mathbf{K}_i \sim U_{n \cdot r_i} \left((-c, c)^{n \cdot r_i} \right) \\ \mu_i = 0 \end{bmatrix}$ 5 6 7 compute $\mathbf{K}_{i_l}, \mathbf{P}_{i_l}, \mathbf{Q}_{i_l}, \mathbf{S}_{i_l}$ with $\mu_{i_l} = |\mathbb{V}_l| - 1$ for which 8 $\begin{pmatrix} \mathbf{W}_l \otimes A - \sum_{i \in \mathbb{V}_l} \overline{B}_i \mathbf{K}_i \overline{C}_i & -\overline{B}_{i_l} \mathbf{P}_{i_l} \\ \mathbf{Q}_{i_l} \overline{C}_{i_l} & \mathbf{S}_{i_l} \end{pmatrix}$ is stable, provided they exist. /* computation of $\{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}_{m_o+1}}$ */ for $i \in \mathbb{V}_{m_s+1}$ do 9 $\mathbf{K}_i \leftarrow \mathbf{0}$ 10 11

A.2 Nondegeneracy of the Dynamic Matrix A

Here, we justify the nondegeneracy assumption on the dynamic matrix A in (1.1). Suppose that the dynamic matrix A is degenerate. Let M be the real matrix for which $J = M^{-1}AM$ is a real block diagonal matrix in the following form:

$$J = diag\left(J_1, \cdots, J_p\right) \tag{A.1}$$

where for each i in $\{1, \dots, p\}$, the submatrix J_i is the i-th real Jordan block. In particular, suppose that J_{p_0+1}, \dots, J_p are all the Jordan blocks associated with the zero eigenvalue. Notice that there exists a positive integer k_0 for which $J_i^k = 0$ for all $k \ge k_0$ and i in $\{p_0 + 1, \dots, p\}$.

By applying a similarity transform to the plant (1.1) with M, we obtain

$$\begin{pmatrix} x_a(k+1) \\ x_b(k+1) \end{pmatrix} = \begin{pmatrix} A_a & \mathbf{0} \\ \mathbf{0} & A_b \end{pmatrix} \begin{pmatrix} x_a(k) \\ x_b(k) \end{pmatrix}$$
$$y(k) = \begin{pmatrix} C_a & C_b \end{pmatrix} \begin{pmatrix} x_a(k) \\ x_b(k) \end{pmatrix}$$
(A.2)

where $\begin{pmatrix} x_a(k) \\ x_b(k) \end{pmatrix} = M^{-1}x(k), A_a = diag(J_1, \cdots, J_{p_0}), A_b = diag(J_{p_0+1}, \cdots, J_p),$ and $\begin{pmatrix} C_a & C_b \end{pmatrix} = CM$. Since the block diagonal elements of A_b are the Jordan blocks associated with the zero eigenvalue, it holds that $x_b(k) = 0$ for all $k \ge k_0$, and from (A.2)

we can derive the following state-space equation:

$$x_a(k+1) = A_a x_a(k)$$

$$y(k) = C_a x_a(k)$$
(A.3)

for $k \ge k_0$, where A_a is a nondegenerate matrix. For this reason, in what regards to achieving asymptotic omniscience, we may design a distributed observer (2.1) for (A.3) to asymptotically resolve the state $x_a(k)$, from which the state x(k) of (1.1) can be obtained using the relation $x(k) = M\begin{pmatrix} x_a(k) \\ 0 \end{pmatrix}$ for $k \ge k_0$.

A.3 Preliminary Concepts and Proof of Proposition 2.3.2

Let us define

$$x(k) = \begin{pmatrix} \chi^{(1)}(k) \\ x'(k) \end{pmatrix} \text{ where } x'(k) = \begin{pmatrix} \chi^{(2)}(k) - \chi^{(1)}(k) \\ \vdots \\ \chi^{(m_a)}(k) - \chi^{(1)}(k) \end{pmatrix}$$

We can re-write (2.3) as follows:

$$x(k+1) = Ax(k) + \sum_{i \in \mathbb{V}} B_i u_i(k)$$

$$y_i(k) = C_i x(k)$$
(A.4)

for each i in \mathbb{V} , where

$$A = \begin{pmatrix} F_o & F_{12} & \cdots & F_{1m_a} \\ \hline \mathbf{0} & A' & \end{pmatrix}$$
$$B_i = \begin{pmatrix} G_{1i}^T & (B_i')^T \end{pmatrix}^T$$
$$C_i = H_i \begin{pmatrix} \begin{pmatrix} \mathbf{1} & \mathbf{0} \\ \mathbf{1}_{m_a-1} & I_{m_a-1} \end{pmatrix} \otimes I_n \end{pmatrix}$$
(A.5)

In (A.5), F_o , $\{F_{1j}\}_{j=2}^{m_a}$, G_{1i} , H_i are from (2.3), and A' and B'_i are defined in (2.5b) and (2.5c), respectively.

To achieve asymptotic synchronization of the system (2.3), we need to find a distributed controller for which the partial state x'(k) of (A.4) converges to zero as $k \to \infty$. To find such distributed controller, we adopt the following procedure (also see Figure A.1 for an illustration):

- 1. Using the method described in Section 2.2, we first design a distributed observer (2.1) for the multi-agent system (A.4) subject to the pre-selected graph \mathcal{G} .
- 2. Then, using results on the synthesis of decentralized control systems, we find fully decentralized controllers for the multi-agent/distributed observer system obtained in Step 1.
- 3. Finally, we recover a distributed controller from the distributed observer and the fully decentralized controllers obtained in Step 1 and Step 2, respectively.



Figure A.1: Diagrams depicting a design procedure for finding a distributed controller.

Based on the aforementioned design procedure, we note that the resulting distributed controller has the following state-space representation:

$$\begin{pmatrix} \hat{x}_i(k+1) \\ z_i(k+1) \\ w_i(k+1) \end{pmatrix} = \begin{pmatrix} A \sum_{j \in \mathbb{N}_i} \mathbf{w}_{ij} \hat{x}_j(k) + \mathbf{K}_i \left(y_i(k) - C_i \hat{x}_i(k) \right) + \mathbf{P}_i z_i(k) \\ \mathbf{Q}_i \left(y_i(k) - C_i \hat{x}_i(k) \right) + \mathbf{S}_i z_i(k) \\ \mathbf{S}_i^d w_i(k) + \mathbf{Q}_i^d \hat{x}_i'(k) \end{pmatrix}$$
(A.6a)
$$u_i(k) = \mathbf{P}_i^d w_i(k) + \mathbf{K}_i^d \hat{x}_i'(k)$$
(A.6b)

for each *i* in \mathbb{V} , where $\hat{x}'_i(k) = \begin{pmatrix} \mathbf{0} & I_{(m_a-1)\cdot n} \end{pmatrix} \hat{x}_i(k)$, and *A* and *C_i* are defined in (A.5). It can be verified that (A.6) is a special case of (2.4). Hence, it remains to consider a parameter choice for (A.6) such that the resulting distributed controller synchronizes the system (2.3). In what follows, we describe particular choices of

$$\mathbf{W} = \left(\mathbf{w}_{ij}\right)_{i,j\in\mathbb{V}}, \left\{\mathbf{K}_{i}, \mathbf{P}_{i}, \mathbf{Q}_{i}, \mathbf{S}_{i}\right\}_{i\in\mathbb{V}} \text{ and } \left\{\mathbf{K}_{i}^{d}, \mathbf{P}_{i}^{d}, \mathbf{Q}_{i}^{d}, \mathbf{S}_{i}^{d}\right\}_{i\in\mathbb{V}}$$

for (A.6) in Appendix A.3.1 and Appendix A.3.2, respectively. A proof of Proposition 2.3.2 is then followed.

A.3.1 A Choice of
$$\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}}, \{\mathbf{K}_i, \mathbf{P}_i, \mathbf{Q}_i, \mathbf{S}_i\}_{i \in \mathbb{V}}$$

We design a distributed observer (2.1) for the system (A.4) subject to the given graph \mathcal{G} . The estimation error $\tilde{x}_i = x - \hat{x}_i$ of (2.1) evolves according to the following state-space equation:

$$\begin{pmatrix} \widetilde{x}(k+1) \\ z(k+1) \end{pmatrix} = \begin{pmatrix} \mathbf{W} \otimes A - \overline{\mathbf{K}} \ \overline{C} & -\overline{\mathbf{P}} \\ \overline{\mathbf{Q}} \ \overline{C} & \overline{\mathbf{S}} \end{pmatrix} \begin{pmatrix} \widetilde{x}(k) \\ z(k) \end{pmatrix} + \begin{pmatrix} \mathbf{1}_m \otimes \left(\sum_{i \in \mathbb{V}} B_i u_i(k)\right) \\ \mathbf{0} \end{pmatrix}$$
(A.7)

where

$$\widetilde{x} = \left(\widetilde{x}_{1}^{T} \dots \widetilde{x}_{m}^{T}\right)^{T}, \qquad z = \left(z_{1}^{T} \dots z_{m}^{T}\right)^{T},$$

$$\overline{C} = \left(\overline{C}_{1}^{T} \dots \overline{C}_{m}^{T}\right)^{T} \text{ with } \overline{C}_{i} = e_{i}^{T} \otimes C_{i}, \qquad (A.8)$$

$$\mathbf{W} = (\mathbf{w}_{ij})_{i,j \in \mathbb{V}},$$

$$\overline{\mathbf{K}} = diag\left(\mathbf{K}_{1}, \dots, \mathbf{K}_{m}\right), \qquad \overline{\mathbf{P}} = diag\left(\mathbf{P}_{1}, \dots, \mathbf{P}_{m}\right),$$

$$\overline{\mathbf{Q}} = diag\left(\mathbf{Q}_{1}, \dots, \mathbf{Q}_{m}\right), \qquad \overline{\mathbf{S}} = diag\left(\mathbf{S}_{1}, \dots, \mathbf{S}_{m}\right)$$

In (A.8), e_i is the *i*-th column of the *m*-dimensional identity matrix.

By writing (A.4) and (A.7) altogether and by omitting $\chi^{(1)}$ from x in (A.4), we can derive the following equation:²

$$\begin{pmatrix} x'(k+1)\\ \widetilde{x}(k+1)\\ z(k+1) \end{pmatrix} = \begin{pmatrix} A' & \mathbf{0} & \mathbf{0}\\ \mathbf{0} & \mathbf{W} \otimes A - \mathbf{\overline{K}} \, \overline{C} & -\mathbf{\overline{P}}\\ \mathbf{0} & \mathbf{\overline{Q}} \, \overline{C} & \mathbf{\overline{S}} \end{pmatrix} \begin{pmatrix} x'(k)\\ \widetilde{x}(k)\\ z(k) \end{pmatrix} + \begin{pmatrix} \sum_{i \in \mathbb{V}} B_i' u_i(k)\\ \mathbf{1}_m \otimes \left(\sum_{i \in \mathbb{V}} B_i u_i(k)\right)\\ \mathbf{0} \end{pmatrix}$$
(A.9a)

$$\hat{x}_{i}'(k) = E_{i} \begin{pmatrix} x'(k) \\ \tilde{x}(k) \\ z(k) \end{pmatrix}$$
(A.9b)

for each *i* in \mathbb{V} , where $E_i = \begin{pmatrix} I_{(m_a-1)\cdot n} & -e_i^T \otimes \begin{pmatrix} \mathbf{0} & I_{(m_a-1)\cdot n} \end{pmatrix} & \mathbf{0} \end{pmatrix}$, e_i is the *i*-th column of the *m*-dimensional identity matrix, and *A*, *A'*, *B_i*, *B'_i* and \overline{C}_i are defined in

²This is valid since (A.9) does not depend on $\chi^{(1)}$.

(A.5) and (A.8), respectively. In (A.9b), we note that

$$E_{i}\begin{pmatrix}x'(k)\\\tilde{x}(k)\\z(k)\end{pmatrix} = x'(k) - \begin{pmatrix}\mathbf{0} & I_{(m_{a}-1)\cdot n}\end{pmatrix}\tilde{x}_{i}(k) = \begin{pmatrix}\mathbf{0} & I_{(m_{a}-1)\cdot n}\end{pmatrix}\hat{x}_{i}(k)$$

Essentially, the state-space equation (A.9) describes a LTI system with state $\begin{pmatrix} x'^T & \tilde{x}^T & z^T \end{pmatrix}^T$, output vector \hat{x}'_i , and inputs $\{u_i\}_{i \in \mathbb{V}}$. If there is no input, i.e., $u_i = 0$ for all i in \mathbb{V} , and the matrix given by

$$\begin{pmatrix} \mathbf{W} \otimes A - \overline{\mathbf{K}} \ \overline{C} & -\overline{\mathbf{P}} \\ \overline{\mathbf{Q}} \ \overline{C} & \overline{\mathbf{S}} \end{pmatrix}$$
(A.10)

is stable, then we can see that the output $\hat{x}'_i(k)$ converges to x'(k) as $k \to \infty$.

The following Lemma states the stabilizability and detectability of (A.9).

Lemma A.3.1. Let a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ and a LTI system (A.4) be given. Suppose that assumptions (i) and (ii) of Proposition 2.3.2 hold. We can find $\mathbf{W}, \overline{\mathbf{K}}, \overline{\mathbf{P}}, \overline{\mathbf{Q}}, \overline{\mathbf{S}}$ in (A.9) for which the resulting system (A.9) is both stabilizable and detectable for all *i* in \mathbb{V} .

Proof. Notice that because of (ii) of Proposition 2.3.2, by Theorem 2.2.2 and the procedures in Appendix A.1.2, we can find \mathbf{W} , $\overline{\mathbf{K}}$, $\overline{\mathbf{P}}$, $\overline{\mathbf{Q}}$, $\overline{\mathbf{S}}$ for which the matrix (A.10) is stable. Under this choice of \mathbf{W} , $\overline{\mathbf{K}}$, $\overline{\mathbf{P}}$, $\overline{\mathbf{Q}}$, $\overline{\mathbf{S}}$, we show the stabilizability and detectability of (A.9). The stabilizability directly follows from (i) of Proposition 2.3.2. The detectability can be verified by the fact that if $u_i = 0$ for all i in \mathbb{V} , then it holds that $\lim_{k\to\infty} \|\hat{x}'_i(k) - x'(k)\| = 0$ for all i in \mathbb{V} and $\lim_{k\to\infty} \left\| \begin{pmatrix} \widetilde{x}(k) \\ z(k) \end{pmatrix} \right\| = 0$.

A.3.2 A Choice of $\left\{ \mathbf{K}_{i}^{d}, \mathbf{P}_{i}^{d}, \mathbf{Q}_{i}^{d}, \mathbf{S}_{i}^{d} \right\}_{i \in \mathbb{V}}$

Consider a set of fully decentralized controllers whose state-space representation is given as follows:

$$w_i(k+1) = \mathbf{S}_i^d w_i(k) + \mathbf{Q}_i^d \hat{x}_i'(k)$$

$$u_i(k) = \mathbf{P}_i^d w_i(k) + \mathbf{K}_i^d \hat{x}_i'(k)$$
(A.11)

for each i in \mathbb{V} .

Consider the closed-loop system obtained by interconnecting (A.9) and (A.11) where the parameters $\mathbf{W}, \overline{\mathbf{K}}, \overline{\mathbf{P}}, \overline{\mathbf{Q}}, \overline{\mathbf{S}}$ of (A.9) are chosen as described in Lemma A.3.1. By the stabilizability and detectability of (A.9) for all *i* in \mathbb{V} , using the results on the synthesis of decentralized control systems [63,64], we can find a parameter choice $\{\mathbf{K}_{i}^{d}, \mathbf{P}_{i}^{d}, \mathbf{Q}_{i}^{d}, \mathbf{S}_{i}^{d}\}_{i \in \mathbb{V}}$ for (A.11) that ensures the stability of the closed-loop system.

A.3.3 Proof of Proposition 2.3.2

Suppose that $\mathbf{W}, \overline{\mathbf{K}}, \overline{\mathbf{P}}, \overline{\mathbf{Q}}, \overline{\mathbf{S}}$ and $\{\mathbf{K}_i^d, \mathbf{P}_i^d, \mathbf{Q}_i^d, \mathbf{S}_i^d\}_{i \in \mathbb{V}}$ are respective parameter choices made by the procedures described in Appendix A.3.1 and Appendix A.3.2. By Lemma A.3.1 and the discussion in Appendix A.3.2, such parameter choices ensure the stability of the closed-loop system resulting from an interconnection of (A.9) and (A.11). We note that, under the same parameter choice, the stability of this closed-loop ensures the synchronization of the multi-agent system (2.3) via the distributed controller described by (A.6), which is a special case of (2.4). Therefore, with the aforementioned parameter choices, we conclude that the distributed controller (A.6) (hence (2.4)) synchronizes the multi-agent system (2.3). This proves the first statement.

Next, we prove the second statement of Proposition 2.3.2. We proceed by writing the state-space equation for agent 1 using (2.3a) and (A.6b) as follows:

$$\chi^{(1)}(k+1) = F_o \chi^{(1)}(k) + \sum_{j=2}^{m_a} F_{1j} \left(\chi^{(j)}(k) - \chi^{(1)}(k) \right) + \sum_{j=1}^m G_{1j} \left(\mathbf{P}_j^d w_j(k) + \mathbf{K}_j^d \hat{x}_j'(k) \right)$$
(A.12)

Without loss of generality, suppose that $F_o = \begin{pmatrix} F_{o,U} & \mathbf{0} \\ \mathbf{0} & F_{o,S} \end{pmatrix}$ where $F_{o,U}$ and $F_{o,S}$ are unstable and stable parts of F_o , respectively. Accordingly, we obtain a partition $\begin{pmatrix} \chi_U^{(1)} \\ \chi_S^{(1)} \end{pmatrix}$ of

$$\chi^{(1)}$$
 and a partition $\begin{pmatrix} G_U \\ G_S \end{pmatrix}$ of $(F_{12} \cdots F_{1m_a} G_{11} \mathbf{P}_1^d \cdots G_{1m} \mathbf{P}_m^d G_{11} \mathbf{K}_1^d \cdots G_{1m} \mathbf{K}_m^d).$

For notational convenience, let

$$x' = \begin{pmatrix} \chi^{(2)} - \chi^{(1)} \\ \vdots \\ \chi^{(m_a)} - \chi^{(1)} \end{pmatrix}, w = \begin{pmatrix} w_1 \\ \vdots \\ w_m \end{pmatrix}, \hat{x}' = \begin{pmatrix} \hat{x}'_1 \\ \vdots \\ \hat{x}'_m \end{pmatrix}$$

Since x'(k), w(k), $\hat{x}'(k)$ converge to zero exponentially as $k \to \infty$, it holds that $\lim_{k\to\infty} \left\| \chi_S^{(1)}(k) \right\| = 0$. Now, we consider the unstable dynamics of (A.12), which can be represented by the following state-space equation:

$$\chi_{U}^{(1)}(k+1) = F_{o,U}\chi_{U}^{(1)}(k) + G_{U}\begin{pmatrix} x'(k)\\ w(k)\\ \hat{x}'(k) \end{pmatrix}$$
(A.13)

Since eigenvalues of $F_{o,U}$ lie on or outside the unit circle in \mathbb{C} , we can verify that a solution

to (A.13) satisfies

$$(F_{o,U})^{-k} \chi_U^{(1)}(k) = \chi_U^{(1)}(0) + \sum_{l=0}^{k-1} (F_{o,U})^{-l-1} G_U \begin{pmatrix} x'(l) \\ w(l) \\ \hat{x}'(l) \end{pmatrix}$$
(A.14)

where the right hand side of (A.14) converges exponentially as $k \to \infty$. Let

$$\chi_{o,U} = \chi_U^{(1)}(0) + \sum_{l=0}^{\infty} (F_{o,U})^{-l-1} G_U \begin{pmatrix} x'(l) \\ w(l) \\ \hat{x}'(l) \end{pmatrix}$$

be the limit point of (A.14).

To complete the proof, let us consider the following state-space equation:

$$\chi_o(k+1) = F_o \chi_o(k), \ \chi_o(0) = \begin{pmatrix} \chi_{o,U} \\ \chi_{o,S} \end{pmatrix}$$

for any vector $\chi_{o,S}$ of a proper dimension. Since $F_{o,U}$ has the unit spectral radius by the assumption of the second statement, due to the exponential convergence rate of (A.14), we can see that $\lim_{k\to\infty} ||\chi^{(1)}(k) - \chi_o(k)|| = 0$ holds.

Using the fact that $\lim_{k\to\infty} \|\chi^{(i)}(k) - \chi^{(1)}(k)\| = 0$ holds for all i in $\mathbb{V}^I \setminus \{1\}$, we conclude that $\lim_{k\to\infty} \|\chi^{(i)}(k) - \chi_o(k)\| = 0$ holds for all i in \mathbb{V}^I . This proves the Proposition.

A.4 Proofs of Lemmas 2.4.3 and 2.4.4

Proof of Lemma 2.4.3: Since the matrix L is a WLM of the graph \mathcal{G} and the positive real number α' satisfies $\alpha' \leq \left(\max_{1 \leq i \leq |\mathbb{V}|} l_{ii}\right)^{-1}$, for every α in $(0, \alpha')$, the matrix \mathbf{W} is

stochastic. Hence, it remains to show that for almost every α in $(0, \alpha')$, $\mathbf{W} \otimes A$ satisfies the UEPP.

Let $\{v_1, \dots, v_s\}$ and $\{\lambda_1, \dots, \lambda_t\}$ be the sets of distinct eigenvalues of A and L, respectively. Under the choice $\mathbf{W} = I_{|\mathbb{V}|} - \alpha L$, we can observe that if $\mathbf{W} \otimes A$ does not satisfy the UEPP, then its nonzero eigenvalue can be expressed as a product

$$(1 - \alpha \lambda)\upsilon = (1 - \alpha \lambda')\upsilon'$$

for distinct λ, λ' in $\{\lambda_1, \dots, \lambda_t\}$ and for distinct v, v' in $\{v_1, \dots, v_s\}$. Since the sets of distinct eigenvalues of A and L are both finite, we conclude that the set of the values of α for which the UEPP does not hold is finite. Hence, for almost every α in $(0, \alpha')$, $\mathbf{W} \otimes A$ satisfies the UEPP.

Proof of Lemma 2.4.4: By the UEPP of $\mathbf{W} \otimes A$, for each nonzero eigenvalue λ of $\mathbf{W} \otimes A$, we can find the unique pair of eigenvalues $\lambda_{\mathbf{W}}$ and λ_A of \mathbf{W} and A, respectively, for which $\lambda = \lambda_{\mathbf{W}} \cdot \lambda_A$ holds. Since \mathbf{W} has all simple eigenvalues and $\mathbf{W} \otimes A$ satisfies the UEPP, we can show that there is a unique eigenvector (unique up to a scale factor), say v, associated with $\lambda_{\mathbf{W}}$, and the geometric multiplicities of λ and λ_A are equal³. Hence, we can see that an eigenvector q of $\mathbf{W} \otimes A$ associated with λ can be written as $q = v \otimes p$ where p is an eigenvector of A associated with λ_A . This proves the Lemma.

A.5 Preliminary Results and Proof of Theorem 2.4.5

In this section, we provide a proof of Theorem 2.4.5. The proof hinges on some results from structured linear system theory [110, 111]. To this end, we briefly review the

³A proof of this argument is along similar lines as that of Lemma 3.8 in [20]

structural controllability and observability of structured linear systems in Appendix A.5.1 and provide the detailed proof of Theorem 2.4.5 in Appendix A.5.3.

A.5.1 Structural Controllability and Observability

Consider a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ with $\mathbb{V} = \{1, \dots, |\mathbb{V}|\}$ and an associated structured linear system whose state-space representation is given as follows:

$$x(k+1) = [A]x(k) + [b_i]u(k)$$

$$y(k) = [c_j]^T x(k)$$
(A.15)

where $[A] \in \mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|}$ is a structure matrix, and $[b_i] \in \mathbb{R}^{|\mathbb{V}|}$ and $[c_j] \in \mathbb{R}^{|\mathbb{V}|}$ are structure vectors. Depending on respective sparse structures, entries of structure matrices and vectors are either zero or indeterminate. In particular, we suppose that [A] is consistent⁴ with the graph \mathcal{G} , and all entries of $[b_i]$ and $[c_j]$ are zero except the *i*-th entry and *j*-th entry, respectively. Under this setting, there are $(|\mathbb{E}| + 2)$ indeterminate entries of [A], $[b_i]$, and $[c_j]$, and if we allow each indeterminate entry to take a value in \mathbb{R} , then a choice of these entries can be represented by a vector in $\mathbb{R}^{|\mathbb{E}|+2}$. In other words, the vectors in $\mathbb{R}^{|\mathbb{E}|+2}$ specify all *numerical realizations* of (A.15).

The following Definition describes the structural controllability and observability of structured linear systems (A.15).

Definition A.5.1. Let a graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ and an associated structured linear system as in (A.15) be given. Let p be a vector in $\mathbb{R}^{|\mathbb{E}|+2}$ that specifies a numerical realization

⁴A structure matrix [A] is consistent with a graph \mathcal{G} if the (i, j)-th entry of [A] is indeterminate if $(j, i) \in \mathbb{E}$, and the entry is zero otherwise.

of (A.15). The pair $([A], [b_i])$ is said to be structurally controllable if for almost all p in $\mathbb{R}^{|\mathbb{E}|+2}$, the resulting numerical realizations of $([A], [b_i])$ are controllable. The structural observability is similarly defined for the pair $([A], [c_i]^T)$.

We can characterize the structural controllability and observability for the system (A.15) in terms of its associated graph as in the following Proposition.

Proposition A.5.2. Let a strongly connected graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ and an associated structured linear system as in (A.15) be given. Then, for each i, j in \mathbb{V} , the pair ($[A], [b_i]$) is structurally controllable and the pair ($[A], [c_j]^T$) is structurally observable.

Proof. The proof directly follows from relevant results from the structured linear system literature (see, for instance, Theorem 1 in [110]). The detail is omitted for brevity. \Box

A.5.2 A Key Lemma

The following Lemma is used in the proof of Theorem 2.4.5.

Lemma A.5.3. *Given a strongly connected graph* $\mathcal{G} = (\mathbb{V}, \mathbb{E})$ *with* $\mathbb{V} = \{1, \dots, |\mathbb{V}|\}$ *, for any fixed vertex r in* \mathbb{V} *, the following are true:*

- (i) There exists a WLM L_1 of \mathcal{G} for which the pair (L_1, e_r^T) is observable.
- (ii) There exists a WLM L_2 of \mathcal{G} for which the pair (L_2, e_r) is controllable.
- (iii) There exists a WLM L_3 of \mathcal{G} for which all eigenvalues of L_3 are simple.

where e_r is the *r*-th column of the |V|-dimensional identity matrix.

Proof. We provide a two-part proof: In the first part, we prove (i) and (ii) using Proposition A.5.2; and then we provide a constructive proof of (iii).

Proof of (i) and (ii): Consider a structured linear system that is associated with \mathcal{G} as in (A.15). By Proposition A.5.2, we can find numerical realizations (A_1, c_r^T) and (A_2, b_r) that are, respectively, observable and controllable. In particular, we may choose A_1 and A_2 to be irreducible and (element-wise) nonnegative.

We compute L_1 from A_1 by applying a special similarity transform used in [112]. This procedure is described as follows: By the Perron-Frobenius Theorem, we can find a right eigenvector \tilde{v} (of A_1) with all positive entries, which corresponds to the Perron-Frobenius eigenvalue $\tilde{\lambda}$. Let M be a diagonal matrix whose diagonal elements are the entries of \tilde{v} . Then, by applying a similarity transform to (A_1, c_r^T) with M, we obtain $(M^{-1}A_1M, c_r^TM)$. Since the observability is preserved under any similarity transform, the new pair $(M^{-1}A_1M, c_r^TM)$ is also observable. Note that $M^{-1}A_1M$ and A_1 have the same sparsity pattern, and so do c_r^T and c_r^TM .

Let us define

$$L_{1} = I_{|\mathbb{V}|} - \frac{1}{\bar{\lambda}} M^{-1} A_{1} M$$
 (A.16)

Notice that L_1 is a WLM of \mathcal{G} , and that eigenvectors of L_1 are same as those of $M^{-1}A_1M$. Since $(M^{-1}A_1M, c_r^TM)$ is observable, by the PBH rank test, we can see that (L_1, e_r^T) is observable.

By a similar argument, we can explicitly find a WLM L_2 of \mathcal{G} for which (L_2, e_r) is a controllable pair. This completes the first part of the proof. **Proof of (iii):** For a WLM L of \mathcal{G} , we represent L as follows:

$$L = \begin{pmatrix} l_1 & \dots & l_{|\mathbb{V}|} \end{pmatrix}^T \tag{A.17}$$

where l_i^T is the *i*-th row of *L*. By re-scaling each row of *L*, we construct a WLM L_3 of \mathcal{G} whose eigenvalues are all simple.

First of all, it is not difficult to show that for a positive real number α_1 , the following matrix has all simple eigenvalues except at the origin.

$$\left(\alpha_{1}l_{1} \quad \mathbf{0} \quad \cdots \quad \mathbf{0}\right)^{T} \in \mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|}, \tag{A.18}$$

where **0** is the $|\mathbb{V}|$ -dimensional zero vector. Suppose that for some positive real numbers $\alpha_1, \dots, \alpha_k$, the following matrix has all simple eigenvalues except at the origin.

$$\left(\alpha_1 l_1 \cdots \alpha_k l_k \quad \mathbf{0} \cdots \quad \mathbf{0}\right)^T \in \mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|}$$
(A.19)

Recall that eigenvalues of a matrix depend continuously on entries of the matrix. Since (A.19) has all simple eigenvalues except at the origin, for a sufficiently small positive real number α_{k+1} , the following matrix has all simple eigenvalues except at the origin.

$$\left(\alpha_{1}l_{1} \cdots \alpha_{k}l_{k} \quad \alpha_{k+1}l_{k+1} \quad \mathbf{0} \cdots \quad \mathbf{0}\right)^{T} \in \mathbb{R}^{|\mathbb{V}| \times |\mathbb{V}|}$$
(A.20)

By induction, we obtain

$$L_3 = \begin{pmatrix} \alpha_1 l_1 & \cdots & \alpha_{|\mathbb{V}|} l_{|\mathbb{V}|} \end{pmatrix}^T$$
(A.21)

that is a WLM of \mathcal{G} and has all simple eigenvalues except at the origin for the selected positive real numbers $\alpha_1, \dots, \alpha_{|\mathbb{V}|}$. Since \mathcal{G} is a strongly connected graph, the eigenvalue of L_3 at the origin is also simple [113]. This completes the second part of the proof. \Box

A.5.3 Proof of Theorem 2.4.5

To begin with, for the graph $\mathcal{G} = (\mathbb{V}, \mathbb{E})$, we define sets

 $\mathbb{L}_{1,r}^{c}(\mathcal{G}) \stackrel{def}{=} \left\{ L \in \mathbb{L}(\mathcal{G}) \mid (L, e_{r}^{T}) \text{ is not observable} \right\}$ $\mathbb{L}_{2,r}^{c}(\mathcal{G}) \stackrel{def}{=} \left\{ L \in \mathbb{L}(\mathcal{G}) \mid (L, e_{r}) \text{ is not controllable} \right\}$ $\mathbb{L}_{1}^{c}(\mathcal{G}) \stackrel{def}{=} \left\{ L \in \mathbb{L}(\mathcal{G}) \mid \text{A right eigenvector of } L \text{ has a zero entry} \right\}$ $\mathbb{L}_{2}^{c}(\mathcal{G}) \stackrel{def}{=} \left\{ L \in \mathbb{L}(\mathcal{G}) \mid \text{A left eigenvector of } L \text{ has a zero entry} \right\}$ $\mathbb{L}_{3}^{c}(\mathcal{G}) \stackrel{def}{=} \left\{ L \in \mathbb{L}(\mathcal{G}) \mid \text{A n eigenvalue of } L \text{ is not simple} \right\}$

and a natural bijective mapping

$$\pi: \mathbb{L}(\mathcal{G}) \to \mathbb{R}_{<0}^{|\mathbb{E}| - |\mathbb{V}|},$$

where e_r is the *r*-th column of the $|\mathbb{V}|$ -dimensional identity matrix, and $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$ is the set of the $(|\mathbb{E}| - |\mathbb{V}|)$ -dimensional vectors whose entries are all negative. To prove Theorem 2.4.5, it is sufficient to show that the sets $\pi(\mathbb{L}_1^c(\mathcal{G})), \pi(\mathbb{L}_2^c(\mathcal{G})), \pi(\mathbb{L}_3^c(\mathcal{G}))$ have the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$.

In [110], the observability is shown to be a *generic property* of structured linear systems. In words, unless every numerical realization of a given structured linear system is not observable, *almost* every numerical realization is observable. By an application of this principle, we can show that unless $\mathbb{L}_{1,r}^c(\mathcal{G}) = \mathbb{L}(\mathcal{G})$ holds, $\pi(\mathbb{L}_{1,r}^c(\mathcal{G}))$ has the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$. By a similar argument for the controllability of structured linear systems, we conclude that unless $\mathbb{L}_{2,r}^c(\mathcal{G}) = \mathbb{L}(\mathcal{G})$ holds, $\pi(\mathbb{L}_{2,r}^c(\mathcal{G}))$ has the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$.

Since \mathcal{G} is a strongly connected graph, by Lemma A.5.3, we can show that for any rin \mathbb{V} , $\mathbb{L}_{1,r}^{c}(\mathcal{G})$ and $\mathbb{L}_{2,r}^{c}(\mathcal{G})$ are proper subsets of $\mathbb{L}(\mathcal{G})$; hence, $\pi (\mathbb{L}_{1,r}^{c}(\mathcal{G}))$ and $\pi (\mathbb{L}_{2,r}^{c}(\mathcal{G}))$ have the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$. Since it holds that $\mathbb{L}_{1}^{c}(\mathcal{G}) = \bigcup_{r \in \mathbb{V}} \mathbb{L}_{1,r}^{c}(\mathcal{G})$ and $\mathbb{L}_{2}^{c}(\mathcal{G}) = \bigcup_{r \in \mathbb{V}} \mathbb{L}_{2,r}^{c}(\mathcal{G})$, we conclude that $\pi (\mathbb{L}_{1}^{c}(\mathcal{G}))$ and $\pi (\mathbb{L}_{2}^{c}(\mathcal{G}))$ have the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$.

Next, to prove that π ($\mathbb{L}_3^c(\mathcal{G})$) has the Lebesgue measure zero, we adopt the following argument from algebra (see, for instance, Chapter 14.6 of [114]). For a matrix L in $\mathbb{R}^{m \times m}$, all solutions to a polynomial equation

$$\Delta(\lambda) \stackrel{def}{=} det(L - \lambda I) = a_m \lambda^m + \dots + a_1 \lambda + a_0 = 0$$
 (A.22)

are distinct if the discriminant

$$D(\Delta) \stackrel{def}{=} a_m^{2m-1} \prod_{1 \le i < j \le m} (\lambda_i - \lambda_j)^2$$

is nonzero, where λ_i and λ_j are solutions to (A.22). This particular discriminant can be written as the resultant $R\left(\Delta, \frac{d}{d\lambda}\Delta\right)$ of $\Delta(\lambda)$ and $\frac{d}{d\lambda}\Delta(\lambda)$ as follows:

$$D(\Delta) = (-1)^{\frac{m(m-1)}{2}} \cdot R\left(\Delta, \frac{\mathrm{d}}{\mathrm{d}\lambda}\Delta\right)$$
(A.23)

The resultant R(f,g) of two polynomials f and g is a polynomial function of coefficients of f and g. Hence, by (A.23), we can see that the discriminant $D(\Delta)$ can be written as a polynomial function of coefficients of $\Delta(\lambda)$ and $\frac{d}{d\lambda}\Delta(\lambda)$. Since coefficients of $\Delta(\lambda)$ and $\frac{d}{d\lambda}\Delta(\lambda)$ are polynomial functions of entries of L, the discriminant $D(\Delta)$ is a polynomial function of entries of L.

Also, notice that for a polynomial function \widetilde{D} defined on $\mathbb{R}^{\widetilde{m}}$, solutions

$$\left\{ q \in \mathbb{R}^{\widetilde{m}} \, \big| \, \widetilde{D}\left(q\right) = 0 \right\}$$

to the polynomial equation $\widetilde{D}(q) = 0$ form either the entire space $\mathbb{R}^{\widetilde{m}}$ or a proper algebraic variety in $\mathbb{R}^{\widetilde{m}}$, which has the Lebesgue measure zero [115].

For a strongly connected graph \mathcal{G} , we have seen from Lemma A.5.3 that there exists $L_3 \in \mathbb{L}(\mathcal{G})$ whose eigenvalues are all simple. Therefore, $\mathbb{L}_3^c(\mathcal{G})$ is a proper subset of $\mathbb{L}(\mathcal{G})$, and by the aforementioned principles, $\pi(\mathbb{L}_3^c(\mathcal{G}))$ has the Lebesgue measure zero in $\mathbb{R}_{<0}^{|\mathbb{E}|-|\mathbb{V}|}$.

Appendix B: Auxiliary Results for Chapter 3

B.1 On Product Metric Space

Given a metric space (X, d), we define a metric \overline{d} on the product space X^k as follows: For $x_{1:k} = (x_1, \dots, x_k)$ and $y_{1:k} = (y_1, \dots, y_k)$ in X^k ,

$$\overline{d}(x_{1:k}, y_{1:k}) = \left[d^2(x_1, y_1) + \dots + d^2(x_k, y_k)\right]^{1/2}$$
(B.1)

Note that $(\mathbb{X}^k, \overline{d})$ is a (product) metric space.

For (X, d) and (X^k, \overline{d}) , the following are true:

- (F1) For each i in $\{1, \dots, k\}$, let \mathbb{K}_i be a compact subset of \mathbb{X} . Then $\mathbb{K}_1 \times \dots \times \mathbb{K}_k$ is a compact subset of \mathbb{X}^k .
- (F2) For a sequence $\left\{x_{1:k}^{(i)}\right\}_{i\in\mathbb{N}}$ in \mathbb{X}^k , $\left\{x_{1:k}^{(i)}\right\}_{i\in\mathbb{N}}$ converges to $x_{1:k}$ in \mathbb{X}^k , i.e.,

$$\lim_{i \to \infty} \overline{d} \left(x_{1:k}^{(i)}, x_{1:k} \right) = 0$$

if and only if $\left\{x_j^{(i)}\right\}_{i\in\mathbb{N}}$ converges to x_j for all j in $\{1, \cdots, k\}$, i.e.,

$$\lim_{i \to \infty} d\left(x_j^{(i)}, x_j\right) = 0$$

for all j in $\{1, \cdots, k\}$.

Consider a function $\mathcal{G} : \mathbb{X}^k \to \mathbb{R}$. By (F2), \mathcal{G} is continuous at $x_{1:k} \in \mathbb{X}^k$ if for any sequence $\left\{x_{1:k}^{(i)}\right\}_{i\in\mathbb{N}}$ for which $\lim_{i\to\infty} d\left(x_j^{(i)}, x_j\right) = 0$ holds for all j in $\{1, \dots, k\}$, it holds that $\lim_{i\to\infty} \left|\mathcal{G}\left(x_{1:k}^{(i)}\right) - \mathcal{G}\left(x_{1:k}\right)\right| = 0$.

B.2 On Randomized Policies

In our problem formulation described in Section 3.1, the randomized transmission policy $\mathcal{T}_k : \mathbb{N} \times \mathbb{X} \times \mathbb{X} \to \{0, 1\}$ dictates the random variable \mathbb{R}_k as in (3.1). Then given respective realizations τ_k , x_{τ_k} , and x_k of the last transmission time τ_k , state \mathbf{x}_{τ_k} of the underlying process at time τ_k , and state \mathbf{x}_k of the process at time k, it holds that

$$\mathbf{R}_{k} = \begin{cases} 0 & \text{with probability } \mathbb{P}\left(\boldsymbol{\mathcal{T}}_{k}\left(\boldsymbol{\tau}_{k}, \boldsymbol{x}_{\boldsymbol{\tau}_{k}}, \mathbf{x}_{k}\right) = 0 \middle| \boldsymbol{\tau}_{k} = \tau_{k}, \boldsymbol{x}_{\boldsymbol{\tau}_{k}} = x_{\tau_{k}}, \mathbf{x}_{k} = x_{k}\right) \\ 1 & \text{with probability } \mathbb{P}\left(\boldsymbol{\mathcal{T}}_{k}\left(\boldsymbol{\tau}_{k}, \boldsymbol{x}_{\boldsymbol{\tau}_{k}}, \mathbf{x}_{k}\right) = 1 \middle| \boldsymbol{\tau}_{k} = \tau_{k}, \boldsymbol{x}_{\boldsymbol{\tau}_{k}} = x_{\tau_{k}}, \mathbf{x}_{k} = x_{k}\right) \end{cases}$$

In Section 3.2, the randomized policy $\mathcal{P}_j : \mathbb{X} \to \{0, 1\}$ dictates the random variable \mathbf{R}_j as in (3.13). Then given a realization x_j of the state \mathbf{x}_j of the underlying process at time j, it holds that

$$\mathbf{R}_{j} = \begin{cases} 0 & \text{with probability } \mathbb{P}\left(\boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right) = 0 \mid \mathbf{x}_{j} = x_{j}\right) \\ 1 & \text{with probability } \mathbb{P}\left(\boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right) = 1 \mid \mathbf{x}_{j} = x_{j}\right) \end{cases}$$

Throughout the work, we restrict our attention to the policies in which

$$\mathbb{P}\left(\boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right)=0 \mid \mathbf{x}_{j}=x_{j}\right)$$

is a measurable function of x_j on the measurable space (X, \mathfrak{B}) . As a case in point,

consider a (deterministic) policy defined by

$$\mathcal{P}_j(\mathbf{x}_j) = \begin{cases} 0 & \text{if } \mathbf{x}_j \in \mathbb{D}_j \\ & & \\ 1 & \text{otherwise} \end{cases}$$

where $\mathbb{D}_j \in \mathfrak{B}$. It can be verified that

$$\mathbb{P}\left(\boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right)=0 \middle| \mathbf{x}_{j}=x_{j}\right) = \begin{cases} 0 & \text{if } x_{j} \in \mathbb{D}_{j} \\ \\ 1 & \text{otherwise} \end{cases}$$

is a measurable function of x_j .

B.3 Preliminary Concepts and Results

We first review some of key definitions and results from probability theory [93,116]. Let (\mathbb{X}, d) be the metric space defined in Section 3.1, and let \mathcal{T} and \mathfrak{B} be a topology and a Borel σ -algebra derived from the metric, respectively. Recall that (\mathbb{X}, d) is assumed to be complete, separable, and proper (see Assumption 3.1.4).

Definition B.3.1. Let μ be a probability measure on $(\mathbb{X}, \mathfrak{B})$. The probability measure is said to be tight if for each $\epsilon > 0$, there exists a compact subset \mathbb{K} of $(\mathbb{X}, \mathcal{T})$ for which $\mu(\mathbb{K}) > 1 - \epsilon$ holds.

The following is adopted from Theorem 7.1.4 in [93].

Lemma B.3.2. Any probability measure μ on $(\mathbb{X}, \mathfrak{B})$ is tight.

Definition B.3.3. A probability measure μ defined on (X, \mathfrak{B}) is said to be closed regular if for every \mathbb{A} in \mathfrak{B} , it holds that

$$\mu(\mathbb{A}) = \sup \left\{ \mu(\mathbb{F}) \, \big| \, \mathbb{F} \in \mathfrak{B} \, closed, \, \mathbb{F} \subset \mathbb{A} \right\} \tag{B.2}$$

From Theorem 7.1.3 in [93], we can state the following Lemma.

Lemma B.3.4. Any probability measure μ on $(\mathbb{X}, \mathfrak{B})$ is closed regular.

Remark B.3.5. For a probability measure μ ,

$$\mu\left(\mathbb{A}\right) = 1 - \mu\left(\mathbb{X} \setminus \mathbb{A}\right) \tag{B.3}$$

where \mathbb{A} in \mathfrak{B} . If μ is closed regular then for every $\delta > 0$, there exists a closed set \mathbb{F} for which $\mathbb{F} \subset \mathbb{X} \setminus \mathbb{A}$ and $\mu(\mathbb{X} \setminus \mathbb{A}) < \mu(\mathbb{F}) + \delta$. Let us define an open set $\mathbb{O} = \mathbb{X} \setminus \mathbb{F}$. We can see that \mathbb{O} satisfies $\mathbb{O} \supset \mathbb{A}$ and $\mu(\mathbb{O}) < \mu(\mathbb{A}) + \delta$. Hence we conclude that

$$\mu\left(\mathbb{A}\right) = \inf\left\{\mu\left(\mathbb{O}\right) \mid \mathbb{O} \in \mathfrak{B} \text{ open}, \mathbb{O} \supset \mathbb{A}\right\}$$
(B.4)

Definition B.3.6 (Convergence of Probability Measures). Let $\{\mu^{(i)}\}_{i\in\mathbb{N}}$ and μ be a sequence of probability measures and a probability measure defined on $(\mathbb{X}, \mathfrak{B})$, respectively, and $C_b(\mathbb{X})$ be the set of all bounded, continuous, real-valued functions on \mathbb{X} . The sequence is said to weakly converge to μ if it holds that

$$\lim_{i \to \infty} \int g \, \mathrm{d}\mu^{(i)} = \int g \, \mathrm{d}\mu$$

for every g in $\mathcal{C}_b(\mathbb{X})$. We denote the weak convergence as $\mu^{(i)} \xrightarrow{w} \mu$.

Definition B.3.7. Let $\{\mu^{(i)}\}_{i\in\mathbb{N}}$ be a sequence of probability measures defined on $(\mathbb{X}, \mathfrak{B})$. The probability measures are said to be uniformly tight if for each $\epsilon > 0$, there exists a compact subset \mathbb{K} of $(\mathbb{X}, \mathcal{T})$ for which $\mu^{(i)}(\mathbb{K}) > 1 - \epsilon$ holds for all i in \mathbb{N} .

A subset \mathbb{A} of \mathbb{X} is said to be a μ -continuity set if its boundary set has the zero measure with respect to μ , i.e., μ (bd (\mathbb{A})) = 0. The following is the *portmanteau theorem* (See Theorem 11.1.1 in [93]).

Theorem B.3.8. For a sequence $\{\mu^{(i)}\}_{i \in \mathbb{N}}$ of probability measures and a probability measure μ on $(\mathbb{X}, \mathfrak{B})$, the following are equivalent:

- 1. $\mu^{(i)} \xrightarrow{w} \mu$
- 2. $\limsup_{i\to\infty}\mu^{(i)}(\mathbb{F}) \leq \mu(\mathbb{F})$ for any closed subset \mathbb{F} of \mathbb{X}
- 3. $\liminf_{i\to\infty}\mu^{(i)}(\mathbb{O}) \ge \mu(\mathbb{O})$ for any open subset \mathbb{O} of \mathbb{X}
- 4. $\lim_{i\to\infty}\mu^{(i)}(\mathbb{A}) = \mu(\mathbb{A})$ for all μ -continuity subset \mathbb{A} of \mathbb{X} .

Based on Remark 3.2.11, we can state the following Lemma.

Lemma B.3.9. Given estimates $\hat{x}_{k:N}$, let $\mathcal{P}_{k:N}$ be non-degenerate policies satisfying $\mathcal{P}_{k:N} \in \mathfrak{P}(\hat{x}_{k:N})$. Consider compact sets $\{\mathbb{K}_j\}_{j=k}^N$ given by¹

$$\mathbb{K}_{j} = \left\{ x \in \mathbb{X} \mid d^{2}\left(x, \hat{x}_{j}\right) \leq c_{j}^{\prime} \right\}$$
(B.5)

Then under the policies $\mathcal{P}_{k:N}$ *it holds that*

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{K}_{j} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right) = 1$$
(B.6)

for all j in $\{k, \dots, N\}$.

The proof follows from the fact that for each j in $\{k, \dots, N\}$, \mathbb{K}_j contains the set $\overline{\mathbb{D}}_j$ defined in (3.27a) and Remark 3.2.11-1.

Remark B.3.10. Given policies $\mathcal{P}_{k:N}$, for each j in $\{k, \dots, N\}$, let us define

$$\mu_{j|j}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right)$$
(B.7a)

$$\mu_{j|j-1}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right)$$
(B.7b)

¹Due to the properness assumption on the metric space (\mathbb{X}, d) , every closed ball is a compact set.

where \mathbb{A} belongs to \mathfrak{B} . Then, the probability measure of the process $\{\mathbf{x}_j\}_{j=k}^N$ evolves as follows:

1. Policy update:

$$\mu_{j|j}\left(\mathbb{A}\right) = \frac{\int_{\mathbb{A}} \mathbb{P}\left(\mathbf{R}_{j} = 0 \mid \mathbf{x}_{j} = x, \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right) \, \mathrm{d}\mu_{j|j-1}}{\mathbb{P}\left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right)}$$

provided

$$\mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right)$$
$$= \int_{\mathbb{X}} \mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{x}_{j}=x, \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right) d\mu_{j|j-1} > 0$$

where $\mathbb{P}(\mathbf{R}_j = 0 | \mathbf{x}_j = x, \mathbf{R}_k = 0, \cdots, \mathbf{R}_{j-1} = 0)$ is the probability specified by the given policy \mathcal{P}_j , i.e.,

$$\mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{x}_{j}=x, \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right) = \mathbb{P}\left(\boldsymbol{\mathcal{P}}_{j}\left(\mathbf{x}_{j}\right)=0 \mid \mathbf{x}_{j}=x\right)$$

2. Process update:

$$\mu_{j|j-1}\left(\mathbb{A}\right) = \int_{\mathbb{X}} p_j\left(x,\mathbb{A}\right) \,\mathrm{d}\mu_{j-1|j-1}$$

where p_j is the transition probability of the process $\{\mathbf{x}_j\}_{j=k}^N$.

B.4 Proof of Proposition 3.2.14

To start with, we note that for each j in $\{k, \dots, N\}$, \mathcal{G}_j can be written as follows:

$$\mathcal{G}_{j}\left(x_{j-1}, \hat{x}_{j:N}\right) = \mathbb{E}_{\mathbf{x}_{j}}\left[\min\left\{d^{2}\left(\mathbf{x}_{j}, \hat{x}_{j}\right) + \mathcal{G}_{j+1}\left(\mathbf{x}_{j}, \hat{x}_{j+1:N}\right), c_{j}'\right\} \middle| \mathbf{x}_{j-1} = x_{j-1}\right]$$

with $\mathcal{G}_{N+1} = 0$.

We prove the Proposition by induction. First, notice that since $\mathcal{G}_{N+1} = 0$, it is a continuous function. Suppose that \mathcal{G}_{j+1} is a continuous function. To show that \mathcal{G}_j is continuous, we rewrite \mathcal{G}_j as follows:

$$\mathcal{G}_{j}\left(x_{j-1}, \hat{x}_{j:N}\right) = \mathbb{E}_{\mathbf{x}_{j}}\left[g\left(\mathbf{x}_{j}, \hat{x}_{j:N}\right) \mid \mathbf{x}_{j-1} = x_{j-1}\right]$$
(B.8)

where $g(x, \hat{x}_{j:N}) = \min \left\{ d^2(x, \hat{x}_j) + \mathcal{G}_{j+1}(x, \hat{x}_{j+1:N}), c'_j \right\}$. Note that g is a continuous function.

To verify the continuity of \mathcal{G}_j , let $\{x_{j-1}^{(i)}\}_{i\in\mathbb{N}}$ and $\{\hat{x}_{j:N}^{(i)}\}_{i\in\mathbb{N}}$ be sequences that converge to x_{j-1} and $\hat{x}_{j:N}$, respectively. For each set \mathbb{A} in \mathfrak{B} , let us define

$$\mu_j^{(i)}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_j \in \mathbb{A} \mid \mathbf{x}_{j-1} = x_{j-1}^{(i)}\right) \tag{B.9}$$

$$\mu_{j}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{x}_{j-1} = x_{j-1}\right)$$
(B.10)

By Assumption 3.1.5-2 and Theorem B.3.8, $\{\mu_j^{(i)}\}_{i\in\mathbb{N}}$ weakly converges to μ_j . Since (\mathbb{X}, d) is a complete, separable metric space (see Assumption 3.1.4), by the Skorokhod representation theorem [94], there exist a sequence of random variables $\{\mathbf{y}_j^{(i)}\}_{i\in\mathbb{N}}$ and a random variable \mathbf{y}_j all defined on a common probability space $(\Omega, \mathfrak{F}, \nu)$ in which the following three facts are true:

- (F1) $\mu_j^{(i)}$ is the probability measure of $\mathbf{y}_j^{(i)}$, i.e., $\nu\left(\left\{\omega \in \Omega \mid \mathbf{y}_j^{(i)}(\omega) \in \mathbb{A}\right\}\right) = \mu_j^{(i)}(\mathbb{A})$ for each \mathbb{A} in \mathfrak{B} .
- (F2) μ_j is the probability measure of \mathbf{y}_j , i.e., $\nu\left(\left\{\omega \in \Omega \mid \mathbf{y}_j(\omega) \in \mathbb{A}\right\}\right) = \mu_j(\mathbb{A})$ for each \mathbb{A} in \mathfrak{B} .
- (F3) $\left\{\mathbf{y}_{j}^{(i)}\right\}_{i\in\mathbb{N}}$ converges to \mathbf{y}_{j} almost surely.

From (F1) and (F2), we can derive

$$\mathcal{G}_{j}\left(x_{j-1}^{(i)}, \hat{x}_{j:N}^{(i)}\right) - \mathcal{G}_{j}\left(x_{j-1}, \hat{x}_{j:N}\right)$$

$$= \mathbb{E}_{\mathbf{x}_{j}}\left[g\left(\mathbf{x}_{j}, \hat{x}_{j:N}^{(i)}\right) \middle| \mathbf{x}_{j-1} = x_{j-1}^{(i)}\right] - \mathbb{E}_{\mathbf{x}_{j}}\left[g\left(\mathbf{x}_{j}, \hat{x}_{j:N}\right) \middle| \mathbf{x}_{j-1} = x_{j-1}\right]$$

$$= \int_{\Omega} g\left(\mathbf{y}_{j}^{(i)}(\omega), \hat{x}_{j:N}^{(i)}\right) \mathrm{d}\nu - \int_{\Omega} g\left(\mathbf{y}_{j}(\omega), \hat{x}_{j:N}\right) \mathrm{d}\nu \qquad (B.11)$$

Notice that by the fact that c'_j is a fixed constant, the sequence $\left\{g\left(\mathbf{y}_j^{(i)}(\cdot), \hat{x}_{j:N}^{(i)}\right)\right\}_{i \in \mathbb{N}}$ is uniformly bounded. i.e., for every $\omega \in \Omega$, it holds that $g\left(\mathbf{y}_j^{(i)}(\omega), \hat{x}_{j:N}^{(i)}\right) \leq c'_j$ for all iin \mathbb{N} .² Also, by the continuity of g and **(F3)**, it holds that

$$\lim_{i \to \infty} g\left(\mathbf{y}_j^{(i)}(\omega), \hat{x}_{j:N}^{(i)}\right) = g\left(\mathbf{y}_j(\omega), \hat{x}_{j:N}\right)$$

for almost every ω in Ω . Using the bounded convergence theorem (see Theorem 16.5 in [116]), we have that

$$\lim_{i \to \infty} \left| \mathcal{G}_j \left(x_{j-1}^{(i)}, \hat{x}_{j:N}^{(i)} \right) - \mathcal{G}_j \left(x_{j-1}, \hat{x}_{j:N} \right) \right|$$
$$= \lim_{i \to \infty} \left| \int_{\Omega} g \left(\mathbf{y}_j^{(i)}(\omega), \hat{x}_{j:N}^{(i)} \right) \mathrm{d}\nu - \int_{\Omega} g \left(\mathbf{y}_j(\omega), \hat{x}_{j:N} \right) \mathrm{d}\nu \right| = 0$$

Finally, by induction, we conclude that the functions $\{\mathcal{G}_j\}_{j=k}^N$ are all continuous. \Box

B.5 Proofs of Proposition 3.2.16 and Lemma 3.2.18

Lemma B.5.1. For each j_0 in $\{k, \dots, N\}$, there exist estimates $\hat{x}_{j_0:N}$ for which the set given by

$$\underline{\mathbb{D}}_{j_0} = \left\{ x_{j_0} \in \mathbb{X} \left| d^2 \left(x_{j_0}, \hat{x}_{j_0} \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_0+1}^* \left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N} \right) \right| \mathbf{x}_{j_0} = x_{j_0} \right] < c'_{j_0} \right\}$$
(B.12)

is non-empty, where $J_{j_0+1}^*$ is defined in (3.25).

²Note that g is a non-negative function.

Proof. Recall how $\{c'_j\}_{j=k}^N$ are determined by (3.8) with the solutions $\{\mathcal{T}_{j:N}^{< j-1>}\}_{j=k+1}^N$ and $\{\mathcal{E}_{j:N}^{< j-1>}\}_{j=k+1}^N$ to the preceding sub-problems – Sub-problem k+1 to Sub-problem N. Let us fix x_{j_0} in \mathbb{X} . Under the choice of $\hat{x}_{j_0} = x_{j_0}$ and $\hat{x}_l = \mathcal{E}_l^{< j_0>}(x_{j_0})$ for each l in $\{j_0 + 1, \dots, N\}$, by a similar argument as in Remark 3.2.2, we can see that

$$\mathbb{E}_{\mathbf{x}_{j_0+1}}\left[J_{j_0+1}^*\left(\mathbf{x}_{j_0+1}, \hat{x}_{j+1:N}\right) \middle| \mathbf{x}_{j_0} = x_{j_0}\right] = \mathbb{E}_{\mathbf{x}_{j_0+1}}\left[J_{j_0+1}^*\left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N}'\right) \middle| \mathbf{x}_{j_0} = 0\right]$$

where $\hat{x}'_l = \mathcal{E}_l^{\langle j_0 \rangle}(0)$ for each l in $\{j_0 + 1, \dots, N\}$. Hence, at $x_{j_0} = \hat{x}_{j_0}$, it holds that

$$d^{2}(x_{j_{0}}, \hat{x}_{j_{0}}) + \mathbb{E}_{\mathbf{x}_{j_{0}+1}} \left[J_{j_{0}+1}^{*}(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}) \mid \mathbf{x}_{j_{0}} = x_{j_{0}} \right]$$

$$= \mathbb{E}_{\mathbf{x}_{j_{0}+1}} \left[J_{j_{0}+1}^{*}(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}) \mid \mathbf{x}_{j_{0}} = x_{j_{0}} \right]$$

$$< c_{j_{0}} + \mathbb{E}_{\mathbf{x}_{j_{0}+1}} \left[J_{j_{0}+1}^{*}(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}) \mid \mathbf{x}_{j_{0}} = x_{j_{0}} \right]$$

$$= c_{j_{0}} + \mathbb{E}_{\mathbf{x}_{j_{0}+1}} \left[J_{j_{0}+1}^{*}(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}) \mid \mathbf{x}_{j_{0}} = 0 \right] = c_{j_{0}}'$$

This proves the Lemma.

Proof of Proposition 3.2.16: By contradiction, suppose that $\hat{x}_{k:N}^*$ is a global minimizer of (3.31) in which the policies $\mathcal{P}_{k:N}^*$ satisfying $\mathcal{P}_{k:N}^* \in \mathfrak{P}(\hat{x}_{k:N}^*)$ are degenerate. Let $j_0 \in \{k, \dots, N\}$ be the smallest integer for which

$$\mathbb{P}\left(\mathbf{R}_{j_{0}}^{*}=0 \mid \mathbf{R}_{k}^{*}=0, \cdots, \mathbf{R}_{j_{0}-1}^{*}=0\right)=0$$
(B.13)

holds. Since j_0 is the smallest such integer, by Remark B.3.10, the probability measure $\mu_{j_0|j_0-1}$ of \mathbf{x}_{j_0} is well-defined.

Using Lemma B.5.1, let us choose $\hat{x}^o_{j_0:N} \in \mathbb{X}^{N-j_0+1}$ for which the set given by

$$\underline{\mathbb{D}}_{j_0} = \left\{ x_{j_0} \in \mathbb{X} \left| d^2 \left(x_{j_0}, \hat{x}_{j_0}^o \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_0+1}^* \left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N}^o \right) \right| \mathbf{x}_{j_0} = x_{j_0} \right] < c_{j_0}' \right\}$$
(B.14)

is non-empty. Note that according to Corollary 3.2.15, the set $\underline{\mathbb{D}}_{j_0}$ is open; hence, from Assumption 3.1.5-1 and Remark B.3.10-2, we have that

$$\mathbb{P}\left(\mathbf{x}_{j_{0}} \in \underline{\mathbb{D}}_{j_{0}} \middle| \mathbf{R}_{k}^{*} = 0, \cdots, \mathbf{R}_{j_{0}-1}^{*} = 0\right) > 0$$
(B.15)

Consider functions $\mathcal{P}^o_{j_0:N}$ defined as

$$\mathcal{P}_{j}^{o}(x_{j}) = \begin{cases} 0 & \text{if } x_{j} \in \underline{\mathbb{D}}_{j} \\ \\ 1 & \text{otherwise} \end{cases}$$

where

$$\underline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{o} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{o} \right) \right| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$

Let us select new policies $\mathcal{P}'_{k:N}$ and estimates $\hat{x}'_{k:N}$ as follows:

$$\mathcal{P}'_{j} = \begin{cases} \mathcal{P}^{*}_{j} & \text{for } j \in \{k, \cdots, j_{0} - 1\} \\ \\ \mathcal{P}^{o}_{j} & \text{for } j \in \{j_{0}, \cdots, N\} \end{cases}$$
$$\hat{x}'_{j} = \begin{cases} \hat{x}^{*}_{j} & \text{for } j \in \{k, \cdots, j_{0} - 1\} \\ \\ \hat{x}^{o}_{j} & \text{for } j \in \{j_{0}, \cdots, N\} \end{cases}$$

By definition, under the new policies $\mathcal{P}'_{k:N}$, it holds that

$$\mathbb{P}\left(\mathbf{x}_{j_{0}} \in \underline{\mathbb{D}}_{j_{0}} \middle| \mathbf{R}_{k}' = 0, \cdots, \mathbf{R}_{j_{0}}' = 1\right) = 0$$
(B.16)

which implies that

$$\mathbb{P}\left(\mathbf{x}_{j_{0}} \in \underline{\mathbb{D}}_{j_{0}} \middle| \mathbf{R}_{k}' = 0, \cdots, \mathbf{R}_{j_{0}-1}' = 0\right)$$
$$= \mathbb{P}\left(\mathbf{x}_{j_{0}} \in \underline{\mathbb{D}}_{j_{0}} \middle| \mathbf{R}_{k}' = 0, \cdots, \mathbf{R}_{j_{0}}' = 0\right) \cdot \mathbb{P}\left(\mathbf{R}_{j_{0}}' = 0 \middle| \mathbf{R}_{k}' = 0, \cdots, \mathbf{R}_{j_{0}-1}' = 0\right)$$
(B.17)

By the fact that $\mathcal{P}'_j = \mathcal{P}^*_j$ for j in $\{k, \dots, j_0 - 1\}$, from (B.15) and (B.17), we can see that

$$\mathbb{P}\left(\mathbf{R}'_{j_0}=0 \mid \mathbf{R}'_k=0,\cdots,\mathbf{R}'_{j_0-1}=0\right) > 0$$
(B.18)

Due to (B.13), from Remark 3.2.8, we can see that

$$\mathbb{E}_{\mathbf{x}_{j_0}}\left[J_{j_0}\left(\mathbf{x}_{j_0}, \mathcal{P}^*_{j_0:N}, \hat{x}^*_{j_0:N}\right) \middle| \mathbf{R}^*_k = 0, \cdots, \mathbf{R}^*_{j_0-1} = 0\right] = c'_{j_0}$$

While, by the way new policies and estimates are defined for j in $\{j_0, \dots, N\}$, using (3.14) and (B.18), we can see that

$$\mathbb{E}_{\mathbf{x}_{j_0}} \left[J_{j_0} \left(\mathbf{x}_{j_0}, \mathcal{P}'_{j_0:N}, \hat{x}'_{j_0:N} \right) \middle| \mathbf{R}'_k = 0, \cdots, \mathbf{R}'_{j_0-1} = 0 \right]$$
$$= \mathbb{E}_{\mathbf{x}_{j_0}} \left[J_{j_0}^* \left(\mathbf{x}_{j_0}, \hat{x}'_{j_0:N} \right) \middle| \mathbf{R}'_k = 0, \cdots, \mathbf{R}'_{j_0-1} = 0 \right] < c'_{j_0}$$

These relations imply that

$$\mathbb{E}_{\mathbf{x}_{j_0}} \left[J_{j_0} \left(\mathbf{x}_{j_0}, \mathcal{P}'_{j_0:N}, \hat{x}'_{j_0:N} \right) \middle| \mathbf{R}'_k = 0, \cdots, \mathbf{R}'_{j_0-1} = 0 \right] \\ < \mathbb{E}_{\mathbf{x}_{j_0}} \left[J_{j_0} \left(\mathbf{x}_{j_0}, \mathcal{P}^*_{j_0:N}, \hat{x}^*_{j_0:N} \right) \middle| \mathbf{R}^*_k = 0, \cdots, \mathbf{R}^*_{j_0-1} = 0 \right]$$

Using the facts that $\mathcal{P}'_j = \mathcal{P}^*_j$ and $\hat{x}'_j = \hat{x}^*_j$ for j in $\{k, \dots, j_0 - 1\}$, and j_0 is the smallest integer for which (B.13) holds, from (3.14), we can infer that

$$\mathcal{G}\left(\hat{x}_{k:N}^{\prime}\right) \leq \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}^{\prime}, \hat{x}_{k:N}^{\prime}\right)\right] < \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}^{*}, \hat{x}_{k:N}^{*}\right)\right] = \mathcal{G}\left(\hat{x}_{k:N}^{*}\right)$$

which violates the optimality of the global minimizer. Therefore, the global minimizer has to be non-degenerate. $\hfill \Box$

Lemma B.5.2. Consider policies
$$\left\{ \boldsymbol{\mathcal{P}}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$$
 and estimates $\left\{ \hat{x}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ that satisfy
 $\boldsymbol{\mathcal{P}}_{k:N}^{(i)} \in \mathfrak{P}\left(\hat{x}_{k:N}^{(i-1)} \right)$

Suppose that the policies are strictly non-degenerate, i.e., there exists $\epsilon > 0$ for which

$$\mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0,\cdots,\mathbf{R}_{j-1}^{(i)}=0\right) \geq \epsilon$$
(B.19)

holds for all i in \mathbb{N} and j in $\{k, \dots, j_0\}$. Then the sequence $\{\hat{x}_j^{(i)}\}_{i \in \mathbb{N}}$ is bounded for all j in $\{k, \dots, j_0\}$.

Proof. By contradiction, suppose that there exists j' in $\{k, \dots, j_0\}$ such that for a subsequence $\left\{\hat{x}_{j'}^{(i_l-1)}\right\}_{l\in\mathbb{N}}$ of $\left\{\hat{x}_{j'}^{(i)}\right\}_{i\in\mathbb{N}}$, it holds that

$$d\left(0,\hat{x}_{j'}^{(i_l-1)}\right) \stackrel{l\to\infty}{\longrightarrow} \infty \tag{B.20}$$

For each l in \mathbb{N} , let us choose a compact set $\mathbb{K}_{j'}^{(i_l)} = \left\{ x \in \mathbb{X} \mid d^2\left(x, \hat{x}_{j'}^{(i_l-1)}\right) \leq c'_j \right\}$. Then, according to Lemma B.3.9, under the policies $\left\{ \mathcal{P}_{k:N}^{(i_l)} \right\}_{l \in \mathbb{N}}$, it holds that

$$\mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_l)} \middle| \mathbf{R}_k^{(i_l)} = 0, \cdots, \mathbf{R}_{j'}^{(i_l)} = 0\right) = 1$$
(B.21)

for all l in \mathbb{N} . Using (B.19), we can derive the following:

$$\mathbb{P}\left(\mathbf{R}_{j'}^{(i_{l})} = 0 \middle| \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
= \frac{\mathbb{P}\left(\mathbf{R}_{j'}^{(i_{l})} = 0 \middle| \mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})}, \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
\mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})} \middle| \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
\leq \mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})} \middle| \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
= \frac{\mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})}, \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
= \frac{\mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})}, \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_{l})} = 0\right) \\
\leq \epsilon^{k-j'} \cdot \mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_{l})}\right) \tag{B.22}$$

holds for all l in \mathbb{N} . Hence, by Lemma B.3.2 and (B.20), we can see that

$$\lim_{l \to \infty} \mathbb{P}\left(\mathbf{x}_{j'} \in \mathbb{K}_{j'}^{(i_l)}\right) = 0 \tag{B.23}$$

In conjunction with (B.22), we conclude that

$$\lim_{l \to \infty} \mathbb{P}\left(\mathbf{R}_{j'}^{(i_l)} = 0 \,\middle|\, \mathbf{R}_k^{(i_l)} = 0, \cdots, \mathbf{R}_{j'-1}^{(i_l)} = 0\right) = 0 \tag{B.24}$$

This contradicts the fact that (B.19) holds for all j in $\{k, \dots, j_0\}$.

Proof of Lemma 3.2.18: For a positive real r, let us define

$$\mathbb{K}_{r} \stackrel{def}{=} \left\{ \hat{x}_{k:N} \in \mathbb{X}^{N-k+1} \, \middle| \, d\left(0, \hat{x}_{j}\right) \le r \text{ for all } j \text{ in } \{k, \cdots, N\} \right\}$$
(B.25)

To prove the Lemma, it is sufficient to show that there exists r > 0 for which with $\mathbb{K} = \mathbb{K}_r$, the statement of the Lemma is true. By contradiction, suppose that there exists a sequence $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}} \subset \mathbb{X}^{N-k+1}$ that satisfies the following hypotheses:

(H1) For each element $\hat{x}_{k:N}^{(i)}$ of the sequence, it holds that $\hat{x}_{k:N}^{(i)} \notin \mathbb{K}_i$.

(H2) For every $\hat{x}_{k:N}$ in \mathbb{K}_i , it holds that $\mathcal{G}(\hat{x}_{k:N}) > \mathcal{G}(\hat{x}_{k:N}^{(i)})$.

We constructively show that the hypothesis (H2) is violated for sufficiently large iin \mathbb{N} . To proceed, let us select policies $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$ that satisfy $\mathcal{P}_{k:N}^{(i)} \in \mathfrak{P}\left(\hat{x}_{k:N}^{(i-1)}\right)$. Let $j_0 \in \{k, \dots, N\}$ be the smallest integer such that there is a subsequence $\left\{ \mathcal{P}_{k:j_0}^{(i)} \right\}_{l \in \mathbb{N}}$ of $\left\{ \mathcal{P}_{k:j_0}^{(i)} \right\}_{i \in \mathbb{N}}$ satisfying³

$$\lim_{l \to \infty} \mathbb{P}\left(\mathbf{R}_{j_0}^{(i_l)} = 0 \,\middle|\, \mathbf{R}_k^{(i_l)} = 0, \cdots, \mathbf{R}_{j_0-1}^{(i_l)} = 0\right) = 0 \tag{B.26}$$

Note that according to (3.14), (B.26) implies that

$$\lim_{l \to \infty} \mathbb{E}_{\mathbf{x}_{j_0}} \left[J_{j_0} \left(\mathbf{x}_{j_0}, \mathcal{P}_{j_0:N}^{(i_l)}, \hat{x}_{j_0:N}^{(i_l-1)} \right) \ \middle| \ \mathbf{R}_k^{(i_l)} = 0, \cdots, \mathbf{R}_{j_0-1}^{(i_l)} = 0 \right] = c'_{j_0} \tag{B.27}$$

³Such j_0 always exists; otherwise according to Lemma B.5.2, the sequence $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}}$ is bounded, which violates (H1).

Also, according to Lemma B.5.2, the sequence $\left\{\hat{x}_{j}^{(i)}\right\}_{i\in\mathbb{N}}$ is bounded for all j in $\{k, \cdots, j_{0}-1\}$.

Using Lemma B.5.1, let us choose $\hat{x}_{j_0:N}^o \in \mathbb{X}^{N-j_0+1}$ for which the set given by

$$\underline{\mathbb{D}}_{j_0} = \left\{ x_{j_0} \in \mathbb{X} \left| d^2 \left(x_{j_0}, \hat{x}_{j_0}^o \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_0+1}^* \left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N}^o \right) \right| \mathbf{x}_{j_0} = x_{j_0} \right] < c_{j_0}' \right\}$$
(B.28)

is non-empty, where $J_{j_0+1}^*$ is defined in (3.25). Note that by Proposition 3.2.14

$$d^{2}\left(x_{j_{0}}, \hat{x}_{j_{0}}^{o}\right) + \mathbb{E}_{\mathbf{x}_{j_{0}+1}}\left[J_{j_{0}+1}^{*}\left(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}^{o}\right) \ \middle| \ \mathbf{x}_{j_{0}} = x_{j_{0}}\right]$$

is a continuous function of x_{j_0} . Hence, for some $\epsilon > 0$, the set defined by

$$\mathbb{B} = \left\{ x_{j_0} \in \mathbb{X} \left| d^2 \left(x_{j_0}, \hat{x}_{j_0}^o \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_0+1}^* \left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N}^o \right) \right| \mathbf{x}_{j_0} = x_{j_0} \right] < c'_{j_0} - \epsilon \right\}$$
(B.29)

is non-empty and open.

Consider functions $\mathcal{P}_{j_0:N}^o$ defined as

$$\mathcal{P}_{j}^{o}(x_{j}) = \begin{cases} 0 & \text{if } x_{j} \in \underline{\mathbb{D}}_{j} \\ \\ 1 & \text{otherwise} \end{cases}$$

where

$$\underline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{o} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{o} \right) \right| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$

Let us select sequences of policies $\left\{ \boldsymbol{\mathcal{P}}_{k:N}^{\prime(i)} \right\}_{i \in \mathbb{N}}$ and estimates $\left\{ \hat{x}_{k:N}^{\prime(i)} \right\}_{i \in \mathbb{N}}$ as follows:

$$\mathcal{P}_{j}^{\prime(i)} = \begin{cases} \mathcal{P}_{j}^{(i)} & \text{for } j \in \{k, \cdots, j_{0} - 1\} \\ \\ \mathcal{P}_{j}^{o} & \text{for } j \in \{j_{0}, \cdots, N\} \end{cases}$$
$$\hat{x}_{j}^{\prime(i)} = \begin{cases} \hat{x}_{j}^{(i)} & \text{for } j \in \{k, \cdots, j_{0} - 1\} \\ \\ \hat{x}_{j}^{o} & \text{for } j \in \{j_{0}, \cdots, N\} \end{cases}$$

We argue that for sufficiently large *i*, it holds that $\hat{x}_{k:N}^{\prime(i-1)} \in \mathbb{K}_{i-1}$ and $\mathcal{G}\left(\hat{x}_{k:N}^{\prime(i-1)}\right) < \mathcal{G}\left(\hat{x}_{k:N}^{\prime(i-1)}\right)$. This contradicts the hypothesis (**H2**), and completes the proof of the Lemma. In what follows, we show that this argument is valid.

Since the sequence $\left\{\hat{x}_{j_0-1}^{(i-1)}\right\}_{i\in\mathbb{N}}$ is bounded, by Lemma B.3.9, and by the fact that $\mathcal{P}_{j}^{\prime(i)} = \mathcal{P}_{j}^{(i)}$ for i in \mathbb{N} and j in $\{k, \cdots, j_0 - 1\}$, there exists a compact set \mathbb{K}_{j_0-1} for which

$$\mathbb{P}\left(\mathbf{x}_{j_{0}-1} \in \mathbb{K}_{j_{0}-1} \middle| \mathbf{R}_{k}^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0}-1}^{\prime(i)} = 0\right)$$
$$= \mathbb{P}\left(\mathbf{x}_{j_{0}-1} \in \mathbb{K}_{j_{0}-1} \middle| \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j_{0}-1}^{(i)} = 0\right) = 1$$

holds for all i in \mathbb{N} . Hence, due to Assumption 3.1.5 and the compactness of \mathbb{K}_{j_0-1} , for some $\delta_{j_0} > 0$, it holds that

$$\mathbb{P}\left(\mathbf{x}_{j_{0}} \in \mathbb{B} \mid \mathbf{R}_{k}^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0}-1}^{\prime(i)} = 0\right) \\
= \int_{\mathbb{K}_{j_{0}-1}} p_{j_{0}}\left(x, \mathbb{B}\right) \, \mathrm{d}\mu_{j_{0}-1|j_{0}-1}^{\prime(i)} \ge \delta_{j_{0}} \cdot \mu_{j_{0}-1|j_{0}-1}^{\prime(i)}\left(\mathbb{K}_{j_{0}-1}\right) = \delta_{j_{0}} \tag{B.30}$$

for all *i* in \mathbb{N} , where the set \mathbb{B} is given in (B.29) and the probability measure $\mu_{j_0-1|j_0-1}^{\prime(i)}$ is defined as

$$\mu_{j_0-1|j_0-1}^{\prime(i)}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j_0-1} \in \mathbb{A} \mid \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_0-1}^{\prime(i)} = 0\right)$$

for each \mathbb{A} in \mathfrak{B} .

Since the sequence $\left\{\hat{x}_{j}^{(i-1)}\right\}_{i\in\mathbb{N}}$ is bounded for all j in $\{k, \dots, j_{0} - 1\}$, for sufficiently large i, we can see that $\hat{x}_{k:N}^{\prime(i-1)} \in \mathbb{K}_{i-1}$. In addition, by the way the policies $\left\{\mathcal{P}_{k:N}^{\prime(i)}\right\}_{i\in\mathbb{N}}$ and estimates $\left\{\hat{x}_{k:N}^{\prime(i)}\right\}_{i\in\mathbb{N}}$ are defined, using (3.14), (B.29) and (B.30), we can derive the following relations:

$$\mathbb{E}_{\mathbf{x}_{j_0}}\left[J_{j_0}\left(\mathbf{x}_{j_0}, \mathcal{P}_{j_0:N}^{\prime(i)}, \hat{x}_{j_0:N}^{\prime(i-1)}\right) \ \middle| \ \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_0-1}^{\prime(i)} = 0\right]$$

$$= \left(\mathbb{E}_{\mathbf{x}_{j_0}} \left[d^2 \left(\mathbf{x}_{j_0}, \hat{x}_{j_0}^{\circ} \right) \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_0}^{\prime(i)} = 0 \right] \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_{0}+1}^* \left(\mathbf{x}_{j_0+1}, \hat{x}_{j_0+1:N}^{\circ} \right) \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_0}^{\prime(i)} = 0 \right] \right) + \mathbb{E}_{\mathbf{x}_{j_0}} \left[\mathbf{R}_{j_0}^{\prime(i)} = 0 \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) \right) + c_{j_0}^{\prime} \cdot \left(1 - \mathbb{P} \left(\mathbf{R}_{j_0}^{\prime(i)} = 0 \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) \right) \right) \stackrel{(1)}{\leq} \left(\mathbb{E}_{\mathbf{x}_{j_0}} \left[d^2 \left(\mathbf{x}_{j_0}, \hat{x}_{j_0}^{\circ} \right) \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0, \mathbf{x}_{j_0} \in \mathbb{B} \right] \right) + \mathbb{E}_{\mathbf{x}_{j_0+1}} \left[J_{j_{0+1}}^* \left(\mathbf{x}_{j_{0+1}}, \hat{x}_{j_{0+1:N}}^{\circ} \right) \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0, \mathbf{x}_{j_0} \in \mathbb{B} \right] \right) \cdot \mathbb{P} \left(\mathbf{x}_{j_0} \in \mathbb{B} \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) + c_{j_0}^{\prime} \cdot \left(1 - \mathbb{P} \left(\mathbf{x}_{j_0} \in \mathbb{B} \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) \right) + c_{j_0}^{\prime} \cdot \left(1 - \mathbb{P} \left(\mathbf{x}_{j_0} \in \mathbb{B} \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) \right) + c_{j_0}^{\prime} \cdot \left(1 - \mathbb{P} \left(\mathbf{x}_{j_0} \in \mathbb{B} \middle| \mathbf{R}_k^{\prime(i)} = 0, \cdots, \mathbf{R}_{j_{0-1}}^{\prime(i)} = 0 \right) \right)$$

(3)
(3)

holds for all i in \mathbb{N} . To obtain (1), we use the fact that

$$d^{2}\left(\mathbf{x}_{j_{0}}, \hat{x}_{j_{0}}^{o}\right) + \mathbb{E}_{\mathbf{x}_{j_{0}+1}}\left[J_{j_{0}+1}^{*}\left(\mathbf{x}_{j_{0}+1}, \hat{x}_{j_{0}+1:N}^{o}\right) \ \middle| \ \mathbf{x}_{j_{0}}\right] < c_{j_{0}}'$$

if $\mathbf{R}_{j_0}^{\prime(i)} = 0$ (or equivalently $\mathbf{x}_{j_0} \in \underline{\mathbb{D}}_{j_0}$), and \mathbb{B} is a subset of $\underline{\mathbb{D}}_{j_0}$; whereas (2) and (3) follow from (B.29) and (B.30), respectively. By a similar argument as in the proof of Proposition 3.2.16, from (B.27) and (B.31), we can observe that for sufficiently large i, there exists $\hat{x}_{k:N}^{(i-1)}$ in \mathbb{K}_{i-1} for which it holds that

$$\mathcal{G}\left(\hat{x}_{k:N}^{\prime(i-1)}\right) \leq \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}^{\prime(i)}, \hat{x}_{k:N}^{\prime(i-1)}\right)\right] < \mathbb{E}_{\mathbf{x}_{k}}\left[J_{k}\left(\mathbf{x}_{k}, \mathcal{P}_{k:N}^{(i)}, \hat{x}_{k:N}^{(i-1)}\right)\right] = \mathcal{G}\left(\hat{x}_{k:N}^{(i-1)}\right)$$

B.6 Proof of Lemma 3.2.24

Lemma B.6.1. Let $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}}$ be a sequence of estimates that converges to $\hat{x}_{k:N}$. The following hold for all j in $\{k, \dots, N\}$:

$$\overline{\mathbb{D}}_{j} \supset \bigcap_{i \in \mathbb{N}} \bigcup_{l \ge i} \overline{\mathbb{D}}_{j}^{(l)}$$
(B.32a)

$$\underline{\mathbb{D}}_{j} \subset \bigcup_{i \in \mathbb{N}} \bigcap_{l \ge i} \underline{\mathbb{D}}_{j}^{(l)}$$
(B.32b)

where

$$\overline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N} \right) \middle| \mathbf{x}_{j} = x_{j} \right] \le c_{j}^{\prime} \right\}$$
$$\underline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N} \right) \middle| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$

and

$$\overline{\mathbb{D}}_{j}^{(i)} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{(i)} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{(i)} \right) \left| \mathbf{x}_{j} = x_{j} \right] \le c_{j}^{\prime} \right\} \right.$$
$$\underline{\mathbb{D}}_{j}^{(i)} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{(i)} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{(i)} \right) \left| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$

where J_{i+1}^* is defined in (3.25).

Proof. Let x_j be an element of $\bigcap_{i \in \mathbb{N}} \bigcup_{l \ge i} \overline{\mathbb{D}}_j^{(l)}$. By definition, there exists an infinite index set $\{i_l\}_{l \in \mathbb{N}}$ for which $x_j \in \overline{\mathbb{D}}_j^{(i_l)}$ holds for all l in \mathbb{N} . Hence, we can see that

$$d^{2}\left(x_{j}, \hat{x}_{j}^{(i_{l})}\right) + \mathbb{E}_{\mathbf{x}_{j+1}}\left[J_{j+1}^{*}\left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{(i_{l})}\right) \middle| \mathbf{x}_{j} = x_{j}\right] \le c_{j}'$$
(B.33)

holds for all l in \mathbb{N} . Using Proposition 3.2.14 and by the fact that $\left\{\hat{x}_{k:N}^{(i)}\right\}_{i\in\mathbb{N}}$ converges to $\hat{x}_{k:N}$, we can derive

$$d^{2}(x_{j}, \hat{x}_{j}) + \mathbb{E}_{\mathbf{x}_{j+1}}\left[J_{j+1}^{*}\left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}\right) \middle| \mathbf{x}_{j} = x_{j}\right] \leq c_{j}'$$
(B.34)
which shows that $x_j \in \overline{\mathbb{D}}_j$. This proves (B.32a).

To show that (B.32b) is true, we consider

$$\underline{\mathbb{D}}_{j}^{c} \supset \bigcap_{i \in \mathbb{N}} \bigcup_{l \ge i} \left(\underline{\mathbb{D}}_{j}^{(l)} \right)^{c}$$
(B.35)

As the rest of the proof is similar to the above arguments, we omit the detail for brevity.

Proof of Lemma 3.2.24: Under the policies $\left\{ \mathcal{P}_{k:N}^{(i)} \right\}_{i \in \mathbb{N}}$, let us define

$$\mu_{j|j-1}^{(i)}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)$$
(B.36a)

$$\mu_{j|j}^{(i)}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j}^{(i)} = 0\right)$$
(B.36b)

for each i in \mathbb{N} and j in $\{k, \dots, N\}$, where \mathbb{A} belongs to \mathfrak{B} . Since $\{\hat{x}_{k:N}^{(i-1)}\}_{i\in\mathbb{N}}$ is a convergent sequence, according to Lemma B.3.9, there exist compact subsets $\{\mathbb{K}_j\}_{j=k}^N$ for which

$$\mu_{j|j}^{(i)}(\mathbb{K}_j) = 1 \tag{B.37}$$

holds for all i in \mathbb{N} and j in $\{k, \dots, N\}$. Hence, the probability measures $\left\{\mu_{j|j}^{(i)}\right\}_{i\in\mathbb{N}}$ are uniformly tight in the sense of Definition B.3.7 for all j in $\{k, \dots, N\}$. According to Theorem 11.5.4 in [93], for each j in $\{k, \dots, N\}$, there exists a subsequence $\left\{\mu_{j|j}^{(i)}\right\}_{l\in\mathbb{N}}$ of $\left\{\mu_{j|j}^{(i)}\right\}_{i\in\mathbb{N}}$ that weakly converges to a probability measure $\mu_{j|j}$. In addition, since $\mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0, \dots, \mathbf{R}_{j-1}^{(i)}\right)$ takes a value in a compact set $[\epsilon, 1]$, for an infinite index set $\{i_{l}\}_{l\in\mathbb{N}}$, it holds that

$$\lim_{l \to \infty} \mathbb{P}\left(\mathbf{R}_{j}^{(i_{l})} = 0 \mid \mathbf{R}_{k}^{(i_{l})} = 0, \cdots, \mathbf{R}_{j-1}^{(i_{l})} = 0\right) = q_{j}$$

where q_j takes a value in $[\epsilon, 1]$. For these reasons, without loss of generality, we prove the Lemma under the following assumptions: For each j in $\{k, \dots, N\}$,

(F1) There exists a probability measure $\mu_{j|j}$ such that $\mu_{j|j}^{(i)} \xrightarrow{w} \mu_{j|j}$ holds.

(F2)
$$\lim_{i\to\infty} \mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0,\cdots,\mathbf{R}_{j-1}^{(i)}=0\right)=q_{j}$$
 whose value belongs to $[\epsilon,1]$.

We proceed by showing that there exist policies $\boldsymbol{\mathcal{P}}_{k:N}$ for which

(A1) For every \mathbb{A} in \mathfrak{B} , it holds that

$$\mu_{j|j}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right)$$
(B.38a)

and

$$q_j = \mathbb{P}\left(\mathbf{R}_j = 0 \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_{j-1} = 0\right)$$
(B.38b)

where \mathbf{R}_j is dictated by $\boldsymbol{\mathcal{P}}_j$ for each j in $\{k, \cdots, N\}$.

(A2) $\mathcal{P}_{k:N}$ belongs to $\mathfrak{P}(\hat{x}'_{k:N})$, where $\hat{x}'_{k:N}$ is the limit of $\left\{\hat{x}^{(i-1)}_{k:N}\right\}_{i\in\mathbb{N}}$.

For our purpose, we define

$$\overline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{\prime} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{\prime} \right) \left| \mathbf{x}_{j} = x_{j} \right] \le c_{j}^{\prime} \right\}$$
(B.39a)
$$\underline{\mathbb{D}}_{j} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{\prime} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{\prime} \right) \left| \mathbf{x}_{j} = x_{j} \right] < c_{j}^{\prime} \right\}$$
(B.39b)

and

$$\overline{\mathbb{D}}_{j}^{(i)} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{(i-1)} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{(i-1)} \right) \right| \mathbf{x}_{j} = x_{j} \right] \le c_{j}^{\prime} \right\}$$
(B.40a)

$$\underline{\mathbb{D}}_{j}^{(i)} = \left\{ x_{j} \in \mathbb{X} \left| d^{2} \left(x_{j}, \hat{x}_{j}^{(i-1)} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{(i-1)} \right) \right| \mathbf{x}_{j} = x_{j} \right] < c_{j}' \right\}$$
(B.40b)

for each i in \mathbb{N} and j in $\{k, \dots, N\}$, where J_{j+1}^* is defined in (3.25). Note that according to Corollary 3.2.15, the sets (B.39a) and (B.40a) are closed, and the sets (B.39b) and (B.40b) are open.

We first make the following two claims to show that (A1) is true.

Claim 1: For each \mathbb{A} in \mathfrak{B} , let us define

$$\mu_{j|j-1}\left(\mathbb{A}\right) \stackrel{def}{=} \int_{\mathbb{X}} p_j\left(x, \mathbb{A}\right) \, \mathrm{d}\mu_{j-1|j-1} \tag{B.41}$$

where p_j is the transition probability of the process $\{\mathbf{x}_j\}_{j=k}^N$. Then, $\mu_{j|j-1}$ is a probability measure on $(\mathbb{X}, \mathfrak{B})$, and the following holds:

$$\lim_{i \to \infty} \mu_{j|j-1}^{(i)} \left(\mathbb{A} \right) = \mu_{j|j-1} \left(\mathbb{A} \right)$$

for all \mathbb{A} in \mathfrak{B} .

To prove the claim, based on Remark B.3.10-2, we note that

$$\mu_{j|j-1}^{(i)}(\mathbb{A}) = \int_{\mathbb{X}} p_j(x,\mathbb{A}) \, \mathrm{d}\mu_{j-1|j-1}^{(i)}$$
(B.42)

holds for each \mathbb{A} in \mathfrak{B} , and by definition, for each x in \mathbb{X} , $\mathbb{A} \mapsto p_j(x, \mathbb{A})$ is a probability measure on $(\mathbb{X}, \mathfrak{B})$. In conjunction with Assumption 3.1.5-2, we can see that $x \mapsto p_j(x, \mathbb{A})$ is a bounded, continuous function. Hence, using (F1), we have that

$$\lim_{i \to \infty} \mu_{j|j-1}^{(i)} \left(\mathbb{A}\right) = \lim_{i \to \infty} \int_{\mathbb{X}} p_j\left(x, \mathbb{A}\right) \, \mathrm{d}\mu_{j-1|j-1}^{(i)}$$
$$= \int_{\mathbb{X}} p_j\left(x, \mathbb{A}\right) \, \mathrm{d}\mu_{j-1|j-1} = \mu_{j|j-1}\left(\mathbb{A}\right)$$
(B.43)

Lastly, the claim that $\mu_{j|j-1}$ is a probability measure on $(\mathbb{X}, \mathfrak{B})$ follows from the fact that $\mathbb{A} \mapsto p_j(x, \mathbb{A})$ is a probability measure on $(\mathbb{X}, \mathfrak{B})$.

Claim 2: There exists a measurable function $f_j : \mathbb{X} \to [0, 1]$ for which

$$\mu_{j|j}\left(\mathbb{A}\right) = \frac{\int_{\mathbb{A}} f_j \,\mathrm{d}\mu_{j|j-1}}{q_j} \tag{B.44}$$

holds for all \mathbb{A} in \mathfrak{B} , where $\mu_{j|j-1}$ is defined in (B.41).

Based on Lemma B.3.8 and Remark B.3.10-1, for any open set \mathbb{O} , we can see that the following relations hold:

$$\mu_{j|j}\left(\mathbb{O}\right) \leq \liminf_{i \to \infty} \mu_{j|j}^{(i)}\left(\mathbb{O}\right) \stackrel{(1)}{\leq} \lim_{i \to \infty} \frac{\mu_{j|j-1}^{(i)}\left(\mathbb{O}\right)}{\mathbb{P}\left(\mathbf{R}_{j}^{(i)}=0 \mid \mathbf{R}_{k}^{(i)}=0, \cdots, \mathbf{R}_{j-1}^{(i)}=0\right)}$$

$$\stackrel{(2)}{=} \frac{\mu_{j|j-1}\left(\mathbb{O}\right)}{q_{j}}$$
(B.45)

where (1) follows from Remark B.3.10-1, and (2) follows from Claim 1 and (F2). We argue that

$$\mu_{j|j}\left(\mathbb{A}\right) \le \frac{\mu_{j|j-1}\left(\mathbb{A}\right)}{q_j} \tag{B.46}$$

holds for any set \mathbb{A} in \mathfrak{B} . To justify the argument, by contradiction, suppose that for a set \mathbb{A} in \mathfrak{B} it holds that

$$\mu_{j|j}\left(\mathbb{A}\right) > \frac{\mu_{j|j-1}\left(\mathbb{A}\right)}{q_j} \tag{B.47}$$

By the closed regularity theorem (see Theorem 7.1.3 in [93]) and Remark B.3.5, we can choose an open set \mathbb{O} containing \mathbb{A} for which

$$\mu_{j|j}(\mathbb{O}) \ge \mu_{j|j}(\mathbb{A}) > \frac{\mu_{j|j-1}(\mathbb{O})}{q_j}$$
$$\ge \frac{\mu_{j|j-1}(\mathbb{A})}{q_j}$$
(B.48)

holds. This contradicts (B.45).

Notice that (B.46) implies that $\mu_{j|j}$ is absolutely continuous with respect to $\mu_{j|j-1}$. According to the Radon-Nikodym theorem, there is a measurable function $f_j : \mathbb{X} \to \mathbb{R}_+$ for which

$$\mu_{j|j}\left(\mathbb{A}\right) = \frac{\int_{\mathbb{A}} f_j \,\mathrm{d}\mu_{j|j-1}}{q_j} \tag{B.49}$$

holds for all \mathbb{A} in \mathfrak{B} . In addition, it can be verified that $f_j(x) \leq 1$ for almost every x in \mathbb{X} ; otherwise (B.46) would be violated. \Box

Proof of (A1): Define policies $\mathcal{P}_{k:N}$ as follows:

$$\mathbb{P}\left(\mathcal{P}_{j}(\mathbf{x}_{j})=0 \mid \mathbf{x}_{j}=x\right) = f_{j}(x)$$
(B.50)

Then we can verify that under the policies $\boldsymbol{\mathcal{P}}_{k:N}$

$$\mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{x}_{j}=x, \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right) = f_{j}(x)$$

and from (B.49), we have that

$$\mu_{j|j}\left(\mathbb{A}\right) = \frac{\int_{\mathbb{A}} \mathbb{P}\left(\mathbf{R}_{j} = 0 \mid \mathbf{x}_{j} = x, \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right) d\mu_{j|j-1}}{q_{j}}$$
(B.51)

Since it holds that $\mu_{k|k-1}(\mathbb{A}) = \mathbb{P}(\mathbf{x}_k \in \mathbb{A})$, from (B.51), we can see that

$$q_k = \mathbb{P}\left(\mathbf{R}_k = 0\right)$$

Hence, in conjunction with Remark B.3.10-1, we can see that

$$\mu_{k|k}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{k} \in \mathbb{A} \mid \mathbf{R}_{k} = 0\right)$$

Then, using (B.41) and Remark B.3.10-2, we can observe that

$$\mu_{k+1|k}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{k+1} \in \mathbb{A} \mid \mathbf{R}_{k} = 0\right)$$

By repeating this verification for each j in $\{k, \dots, N\}$, we conclude that

$$\mu_{j|j}\left(\mathbb{A}\right) = \mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right)$$

and

$$q_j = \mathbb{P}\left(\mathbf{R}_j = 0 \mid \mathbf{R}_k = 0, \cdots, \mathbf{R}_{j-1} = 0\right)$$

hold for all j in $\{k, \cdots, N\}$.

Henceforth, we make two additional claims under the policies $\mathcal{P}_{k:N}$ determined as in (B.50) to show that (A2) is valid.

Claim 3: For any Borel measurable subset \mathbb{A} contained in $\overline{\mathbb{D}}_{j}^{c}$, it holds that

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0\right) = 0$$

To prove the claim, let \mathbb{O} be an open set contained in $\overline{\mathbb{D}}_{j}^{c}$. By Remark 3.2.11-1 and Remark B.3.10-1, we can derive the following:

$$\mu_{j|j}^{(i)}\left(\mathbb{O}\right) = \mu_{j|j}^{(i)}\left(\mathbb{O}\cap\overline{\mathbb{D}}_{j}^{(i)}\right) \leq \frac{\mu_{j|j-1}^{(i)}\left(\mathbb{O}\cap\overline{\mathbb{D}}_{j}^{(i)}\right)}{\mathbb{P}\left(\mathbf{R}_{j}^{(i)} = 0 \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)}$$
(B.53)

By applying Lemma B.3.8, we obtain

$$\mu_{j|j}(\mathbb{O}) \leq \liminf_{i \to \infty} \mu_{j|j}^{(i)}(\mathbb{O})$$

$$\leq \liminf_{i \to \infty} \frac{\mu_{j|j-1}^{(i)}\left(\mathbb{O} \cap \overline{\mathbb{D}}_{j}^{(i)}\right)}{\mathbb{P}\left(\mathbf{R}_{j}^{(i)} = 0 \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)}$$

$$\leq \liminf_{i \to \infty} \frac{\mu_{j|j-1}^{(i)}\left(\mathbb{O} \cap \left(\bigcup_{l \geq i} \overline{\mathbb{D}}_{j}^{(l)}\right)\right)}{\mathbb{P}\left(\mathbf{R}_{j}^{(i)} = 0 \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)}$$

$$\stackrel{(1)}{\leq} \frac{\mu_{j|j-1}\left(\mathbb{O} \cap \left(\bigcup_{l \geq i_{0}} \overline{\mathbb{D}}_{j}^{(l)}\right)\right)}{\mathbb{P}\left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right)}$$
(B.54)

holds for all i_0 in \mathbb{N} , where (1) follows from Claim 1, (A1), and the fact that $\left\{\bigcup_{l\geq i}\overline{\mathbb{D}}_j^{(l)}\right\}_{i\in\mathbb{N}}$ is a decreasing sequence of measurable sets. Hence, from Lemma B.6.1, we have that

$$\mu_{j|j}\left(\mathbb{O}\right) \leq \frac{\mu_{j|j-1}\left(\mathbb{O}\cap\left(\bigcap_{i\in\mathbb{N}}\bigcup_{l\geq i}\overline{\mathbb{D}}_{j}^{(l)}\right)\right)}{\mathbb{P}\left(\mathbf{R}_{j}=0 \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{j-1}=0\right)} = 0$$
(B.55)

Since $\overline{\mathbb{D}}_j^c$ is an open set, by selecting $\mathbb{O} = \overline{\mathbb{D}}_j^c$, we conclude that

$$\mu_{j|j}\left(\mathbb{A}\right) \le \mu_{j|j}\left(\overline{\mathbb{D}}_{j}^{c}\right) = 0 \tag{B.56}$$

holds for every Borel measurable subset \mathbb{A} of $\overline{\mathbb{D}}_{j}^{c}$.

Claim 4: Suppose that $\mathbb{P}\left(\mathbf{R}_{j}=1 \mid \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right) > 0$. Then, for any Borel measurable subset \mathbb{A} contained in $\underline{\mathbb{D}}_{j}$, it holds that

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1\right) = 0$$

To prove the claim, let \mathbb{F} be a closed set contained in $\underline{\mathbb{D}}_j$. Notice that by Remark 3.2.11-2

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{F} \cap \underline{\mathbb{D}}_{j}^{(i)} \middle| \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j}^{(i)} = 0\right)$$
$$= \frac{\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{F} \cap \underline{\mathbb{D}}_{j}^{(i)} \middle| \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)}{\mathbb{P}\left(\mathbf{R}_{j}^{(i)} = 0 \middle| \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0\right)}$$
(B.57)

Using (B.57) and Theorem B.3.8, we can derive the following:

$$\mu_{j|j} (\mathbb{F}) \geq \limsup_{i \to \infty} \mu_{j|j}^{(i)} (\mathbb{F})$$

$$\geq \limsup_{i \to \infty} \mu_{j|j}^{(i)} \left(\mathbb{F} \cap \underline{\mathbb{D}}_{j}^{(i)} \right)$$

$$= \limsup_{i \to \infty} \frac{\mu_{j|j-1}^{(i)} \left(\mathbb{F} \cap \underline{\mathbb{D}}_{j}^{(i)} \right)}{\mathbb{P} \left(\mathbf{R}_{j}^{(i)} = 0 \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0 \right)}$$

$$\geq \limsup_{i \to \infty} \frac{\mu_{j|j-1}^{(i)} \left(\mathbb{F} \cap \left(\bigcap_{l \geq i} \underline{\mathbb{D}}_{j}^{(l)} \right) \right)}{\mathbb{P} \left(\mathbf{R}_{j}^{(i)} = 0 \mid \mathbf{R}_{k}^{(i)} = 0, \cdots, \mathbf{R}_{j-1}^{(i)} = 0 \right)}$$

$$\stackrel{(1)}{\geq} \frac{\mu_{j|j-1} \left(\mathbb{F} \cap \left(\bigcap_{l \geq i_{0}} \underline{\mathbb{D}}_{j}^{(l)} \right) \right)}{\mathbb{P} \left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)}$$
(B.58)

holds for all i_0 in \mathbb{N} , where (1) follows from Claim 1, (A1), and the fact that $\left\{\bigcap_{l\geq i} \underline{\mathbb{D}}_j^{(l)}\right\}_{i\in\mathbb{N}}$

is an increasing sequence of measurable sets. Hence, from Lemma B.6.1, we have that

$$\mu_{j|j} (\mathbb{F}) \geq \frac{\mu_{j|j-1} \left(\mathbb{F} \cap \left(\bigcup_{i \in \mathbb{N}} \bigcap_{l \geq i} \underline{\mathbb{D}}_{j}^{(l)} \right) \right)}{\mathbb{P} \left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)}$$
$$= \frac{\mu_{j|j-1} (\mathbb{F})}{\mathbb{P} \left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)}$$
(B.59)

Using this relation, we can see that

$$\mu_{j|j-1} (\mathbb{F})$$

$$= \mathbb{P} \left(\mathbf{x}_{j} \in \mathbb{F} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)$$

$$= \mathbb{P} \left(\mathbf{x}_{j} \in \mathbb{F} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right) \cdot \mathbb{P} \left(\mathbf{R}_{j} = 0 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)$$

$$+ \mathbb{P} \left(\mathbf{x}_{j} \in \mathbb{F} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1 \right) \cdot \mathbb{P} \left(\mathbf{R}_{j} = 1 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)$$

$$\geq \mu_{j|j-1} (\mathbb{F})$$

$$+ \mathbb{P} \left(\mathbf{x}_{j} \in \mathbb{F} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1 \right) \cdot \mathbb{P} \left(\mathbf{R}_{j} = 1 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0 \right)$$
(B.60)

Using the fact that $\mathbb{P}\left(\mathbf{R}_{j}=1 \mid \mathbf{R}_{k}=0, \cdots, \mathbf{R}_{j-1}=0\right) > 0$, we can derive that

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{F} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1\right) = 0$$
(B.61)

From the closed regularity theorem (see Theorem 7.1.3 in [93]), for any Borel measurable set \mathbb{A} contained in $\underline{\mathbb{D}}_j$, we obtain

$$\mathbb{P}\left(\mathbf{x}_{j} \in \mathbb{A} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1\right) = 0$$
(B.62)

Proof of (A2): Recall the definitions of

$$\mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}\left(\mathbf{x}_{j}, \boldsymbol{\mathcal{P}}_{j:N}, \hat{x}_{j:N}'\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right]$$

and J_j^* given in (3.14) and (3.25), respectively.

For j = N, using Claim 3 and Claim 4, we can derive the following:

$$\mathbb{E}_{\mathbf{x}_{N}}\left[d^{2}\left(\mathbf{x}_{N}, \hat{x}_{N}'\right) \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{N} = 0\right]$$
$$= \mathbb{E}_{\mathbf{x}_{N}}\left[\min\left\{d^{2}\left(\mathbf{x}_{N}, \hat{x}_{N}'\right), c_{N}'\right\} \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{N} = 0\right]$$
(B.63a)

and

$$c'_{N} = \mathbb{E}_{\mathbf{x}_{N}} \left[\min \left\{ d^{2} \left(\mathbf{x}_{N}, \hat{x}'_{N} \right), c'_{N} \right\} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{N} = 1 \right]$$
(B.63b)

provided that $\mathbb{P}\left(\mathbf{R}_{N}=1 \mid \mathbf{R}_{k}=0,\cdots,\mathbf{R}_{N-1}=0\right) > 0$. From (3.14), (3.25), and (B.63), we can derive that

$$\mathbb{E}_{\mathbf{x}_{N}}\left[J_{N}\left(\mathbf{x}_{N}, \boldsymbol{\mathcal{P}}_{N}, \hat{x}_{N}'\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{N-1} = 0\right]$$
$$= \mathbb{E}_{\mathbf{x}_{N}}\left[J_{N}^{*}\left(\mathbf{x}_{N}, \hat{x}_{N}'\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{N-1} = 0\right]$$
(B.64)

Suppose that

$$\mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1} \left(\mathbf{x}_{j+1}, \mathcal{P}_{j+1:N}, \hat{x}'_{j+1:N} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
$$= \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}'_{j+1:N} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
(B.65)

holds. Then, using Claim 3 and Claim 4, we can derive the following:

$$\mathbb{E}_{\mathbf{x}_{j}} \left[d^{2} \left(\mathbf{x}_{j}, \hat{x}_{j}^{\prime} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1} \left(\mathbf{x}_{j+1}, \mathcal{P}_{j+1:N}, \hat{x}_{j+1:N}^{\prime} \right) \middle| \mathbf{x}_{j} \right] \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
$$= \mathbb{E}_{\mathbf{x}_{j}} \left[\min \left\{ d^{2} \left(\mathbf{x}_{j}, \hat{x}_{j}^{\prime} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J_{j+1}^{*} \left(\mathbf{x}_{j+1}, \hat{x}_{j+1:N}^{\prime} \right) \middle| \mathbf{x}_{j} \right], c_{j}^{\prime} \right\} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
$$= \mathbb{E}_{\mathbf{x}_{j}} \left[J_{j}^{*} \left(\mathbf{x}_{j}, \hat{x}_{j:N}^{\prime} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 0 \right]$$
(B.66a)

and

$$c'_{j} = \mathbb{E}_{\mathbf{x}_{j}} \left[\min \left\{ d^{2} \left(\mathbf{x}_{j}, \hat{x}'_{j} \right) + \mathbb{E}_{\mathbf{x}_{j+1}} \left[J^{*}_{j+1} \left(\mathbf{x}_{j+1}, \hat{x}'_{j+1:N} \right) \middle| \mathbf{x}_{j} \right], c'_{j} \right\} \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1 \right]$$

$$= \mathbb{E}_{\mathbf{x}_{j}} \left[J^{*}_{j} \left(\mathbf{x}_{j}, \hat{x}'_{j:N} \right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j} = 1 \right]$$
(B.66b)

provided that $\mathbb{P}\left(\mathbf{R}_{j} = 1 \mid \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right) > 0$. From (3.14), (3.25), and (B.66), we can derive that

$$\mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}\left(\mathbf{x}_{j}, \boldsymbol{\mathcal{P}}_{j:N}, \hat{x}_{j:N}'\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right]$$
$$= \mathbb{E}_{\mathbf{x}_{j}}\left[J_{j}^{*}\left(\mathbf{x}_{j}, \hat{x}_{j:N}'\right) \middle| \mathbf{R}_{k} = 0, \cdots, \mathbf{R}_{j-1} = 0\right]$$
(B.67)

By induction, we conclude that (B.67) holds for all j in $\{k, \dots, N\}$. By Definition 3.2.5 and the fact that

$$\min_{\boldsymbol{\mathcal{P}}_{k:N}'} \mathbb{E}_{\mathbf{x}_{k}} \left[J_{k} \left(\mathbf{x}_{k}, \boldsymbol{\mathcal{P}}_{k:N}', \hat{x}_{k:N}' \right) \right] = \mathbb{E}_{\mathbf{x}_{k}} \left[J_{k}^{*} \left(\mathbf{x}_{k}, \hat{x}_{k:N}' \right) \right]$$

it holds that $\boldsymbol{\mathcal{P}}_{k:N} \in \boldsymbol{\mathfrak{P}}(\hat{x}'_{k:N}).$

B.7 Proof of Theorem 3.3.1

Notice that under (3.41), the cost-to-go of (3.2) from time k can be written as follows:

with $\mathcal{J}_{N+1} = 0$, where

$$\mathbf{K} = \begin{cases} \min\left\{j \in \{k, \cdots, N\} \middle| \mathbf{R}_j = 1\right\} & \text{if } \mathbf{R}_j = 1 \text{ for some } j \in \{k, \cdots, N\} \\ N & \text{otherwise} \end{cases}$$

First we consider the case where for each k in $\{1, \dots, N\}$, $\mathcal{T}_{k:N}^{* < k-1>}$ and $\mathcal{E}_{k:N}^{* < k-1>}$ are a jointly optimal solution of Sub-problem k. We will show that for any transmission polices $\mathcal{T}_{1:N}$ and estimation rules $\mathcal{E}_{1:N}$,

$$\mathcal{J}_{k}\left(x_{k-1}, \mathcal{T}_{k:N}^{*}, \mathcal{E}_{k:N}^{*}\right) \leq \mathcal{J}_{k}\left(x_{k-1}, \mathcal{T}_{k:N}, \mathcal{E}_{k:N}\right)$$
(B.69)

holds for all x_{k-1} in \mathbb{X} and all k in $\{1, \dots, N\}$.

For k = N, note that (B.68) can be written as

$$\mathcal{J}_{N}\left(x_{N-1}, \mathcal{T}_{N}, \mathcal{E}_{N}\right)$$
$$= \mathbb{E}\left[d^{2}\left(\mathbf{x}_{N}, \hat{\mathbf{x}}_{N}\right) + c_{N} \cdot \mathbf{R}_{N} \mid \mathbf{x}_{N-1} = x_{N-1}, \mathcal{T}_{N}^{< N-1>}, \mathcal{E}_{N}^{< N-1>}\right]$$

By joint optimality of the solution $\mathcal{T}_{N}^{* < N-1 >}$ and $\mathcal{E}_{N}^{* < N-1 >}$, we can see that

$$\mathcal{J}_{N}\left(x_{N-1}, \mathcal{T}_{N}^{*}, \mathcal{E}_{N}^{*}\right) \leq \mathcal{J}_{N}\left(x_{N-1}, \mathcal{T}_{N}, \mathcal{E}_{N}\right)$$

holds for all x_{N-1} in \mathbb{X} .

Suppose that

$$\mathcal{J}_{j+1}\left(x_{j}, \mathcal{T}_{j+1:N}^{*}, \mathcal{E}_{j+1:N}^{*}\right) \leq \mathcal{J}_{j+1}\left(x_{j}, \mathcal{T}_{j+1:N}, \mathcal{E}_{j+1:N}\right)$$
(B.70)

holds for all x_j in \mathbb{X} and all j in $\{k, \dots, N-1\}$. Let $\{c_j^{\prime*}\}_{j=k}^N$ be the stopping costs determined by (3.8) with the jointly optimal solutions $\{\mathcal{T}_{j:N}^{*<j-1>}\}_{j=k+1}^N$ and $\{\mathcal{E}_{j:N}^{*<j-1>}\}_{j=k+1}^N$ and let $\{c_j^{\prime}\}_{j=k}^N$ be constants determined by

$$c'_{j} = c_{j} + \inf_{x_{j} \in \mathbb{X}} \mathcal{J}_{j+1}\left(x_{j}, \mathcal{T}_{j+1:N}, \mathcal{E}_{j+1:N}\right)$$
(B.71)

From Remark 3.1.10 and (3.8), we note that the stopping costs $\{c'_j^*\}_{j=k}^N$ satisfy

$$c_{j}^{\prime*} = c_{j} + \mathcal{J}_{j+1}\left(x_{j}, \mathcal{T}_{k+1:N}^{*}, \mathcal{E}_{k+1:N}^{*}\right)$$

which does not depend on x_j . Then, from (B.68), we can derive the following relations

for each x_{k-1} in \mathbb{X} :

$$\begin{aligned}
\mathcal{J}_{k}\left(x_{k-1}, \mathcal{T}_{k:N}^{*}, \mathcal{E}_{k:N}^{*}\right) \\
&= \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}^{\prime*} \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{* < k-1 >}, \mathcal{E}_{k:N}^{* < k-1 >}\right] \\
\stackrel{(1)}{\leq} \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}^{\prime*} \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1 >}, \mathcal{E}_{k:N}^{< k-1 >}\right] \\
\stackrel{(2)}{\leq} \mathbb{E}\left[\sum_{j=k}^{\mathbf{K}} d^{2}\left(\mathbf{x}_{j}, \hat{\mathbf{x}}_{j}\right) + c_{\mathbf{K}}^{\prime} \cdot \mathbf{R}_{\mathbf{K}} \middle| \mathbf{x}_{k-1} = x_{k-1}, \mathcal{T}_{k:N}^{< k-1 >}, \mathcal{E}_{k:N}^{< k-1 >}\right] \\
\stackrel{(3)}{\leq} \mathcal{J}_{k}\left(x_{k-1}, \mathcal{T}_{k:N}, \mathcal{E}_{k:N}\right)
\end{aligned} \tag{B.72}$$

where (1) follows from the fact that $\mathcal{T}_{k:N}^{* < k-1 >}$ and $\mathcal{E}_{k:N}^{* < k-1 >}$ are a jointly optimal solution of Sub-problem k, (2) follows from Remark 3.1.11 and the fact that $c'_j \le c'_j$ holds for all j in $\{k, \dots, N\}$, and (3) follows from (B.68) and (B.71).

By induction, we can see that (B.69) holds for all x_{k-1} in \mathbb{X} and all k in $\{1, \dots, N\}$. Hence, we conclude that the solution $\mathcal{T}_{1:N}^*$ and $\mathcal{E}_{1:N}^*$ determined by (3.41) is jointly optimal for (3.2).

To prove the statement for person-by-person optimality, we note that for every k in $\{1, \dots, N\}$, with $\mathcal{T}_{k:N}^{* < k-1 >} \left(\mathcal{E}_{k:N}^{* < k-1 >} \right)$ fixed, $\mathcal{E}_{k:N}^{* < k-1 >} \left(\mathcal{T}_{k:N}^{* < k-1 >} \right)$ is a global minimizer of (3.4) for Sub-problem k. By a similar argument as for the joint optimality case, we can observe that with $\mathcal{T}_{k:N}^{*}$ fixed, $\mathcal{E}_{k:N}^{*}$ determined by (3.41) minimizes (B.68), and vice versa for all x_{k-1} in X and all k in $\{1, \dots, N\}$. This proves that $\mathcal{T}_{1:N}^{*}$ and $\mathcal{E}_{1:N}^{*}$ are person-by-person optimal for (3.2).

Appendix C: Auxiliary Results for Chapter 4

C.1 Proof of Proposition 4.2.4

We proceed by showing that for any C_1 function $S_{RD} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ satisfying (**P1**), the condition (**P2**) does not hold. Then, from Theorem 4.2.3, we conclude that the replicator dynamics (4.3) are not passive.

Let us re-write (4.3) in the following form:

$$\dot{x}_{i} = \sum_{j=1}^{n} x_{i} x_{j} \left(p_{i} - p_{j} \right)$$
(C.1)

Note that any C^1 function S_{RD} satisfying (P1) should be of the form

$$S_{RD}(p,x) = \frac{1}{4} \sum_{i=1}^{n} \sum_{j=1}^{n} x_i x_j (p_i - p_j)^2 + S(x)$$

where S is a C^1 function. By taking a partial derivative of S_{RD} with respect to x, we obtain

$$\nabla_x^T S_{RD}(p, x) V(p, x) = \begin{bmatrix} \left(\frac{1}{2} \sum_{j=1}^n x_j (p_1 - p_j)^2 \\ \vdots \\ \frac{1}{2} \sum_{j=1}^n x_j (p_n - p_j)^2 \right) + \nabla_x S(x) \end{bmatrix}^T \begin{pmatrix} \sum_{j=1}^n x_1 x_j (p_1 - p_j) \\ \vdots \\ \sum_{j=1}^n x_n x_j (p_n - p_j) \end{pmatrix}$$

Let us choose $x_j = 0$ for all $j \ge 3$. Then, we obtain

$$\nabla_x^T S_{RD}(p,x) V(p,x) = \begin{pmatrix} \frac{1}{2} x_2 (p_1 - p_2)^2 + \frac{\partial S}{\partial x_1}(x) \\ \frac{1}{2} x_1 (p_1 - p_2)^2 + \frac{\partial S}{\partial x_2}(x) \end{pmatrix}^T \begin{pmatrix} x_1 x_2 (p_1 - p_2) \\ -x_1 x_2 (p_1 - p_2) \end{pmatrix}$$
$$= -\frac{1}{2} x_1 x_2 (p_1 - p_2) \left[(x_1 - x_2) (p_1 - p_2)^2 + 2 \left(\frac{\partial S}{\partial x_2}(x) - \frac{\partial S}{\partial x_1}(x) \right) \right]$$

Note that for fixed x_1, x_2 (except for the points at which $\nabla_x^T S_{RD}(p, x)V(p, x) = 0$ holds for all p in \mathbb{R}^n), there exists $p \in \mathbb{R}^n$ for which $\nabla_x^T S_{RD}(p, x)V(p, x) > 0$ holds. Therefore, the function S_{RD} does not satisfy (**P2**). Since we made an arbitrary choice of S_{RD} , we conclude that no C^1 function satisfies (**P1**) and (**P2**) for the replicator dynamics.

C.2 Proof of Proposition 4.2.5

We first note that the condition (A) implies so-called the Strict Positive Correlation (SPC) [102] given by

$$V(p,x) \neq 0$$
 implies $p^T V(p,x) > 0$ (SPC)

Let $\gamma : \mathbb{R}^n \to \mathbb{R}$ be a C^1 function for which (I) holds. It can be verified that γ satisfies

$$\nabla_p \gamma(\hat{p}) = V(p, x) \tag{C.2a}$$

$$\nabla_x^T \gamma(\hat{p}) V(p, x) = -\left(\mathbf{1}^T \varrho(\hat{p})\right) \left(p^T V(p, x)\right)$$
(C.2b)

Let us select a candidate storage function as $S_{EPT}(p, x) = \gamma(\hat{p})$ where $\hat{p} = p - \mathbf{1} \cdot p^T x$. Due to (C.2a), the function S_{EPT} satisfies (P1). In conjunction with the fact that $\varrho(\hat{p}) = \mathbf{0}$ implies $V(p, x) = \mathbf{0}$, due to (SPC) and (C.2b), we can see that (P2) holds for $\eta = 0$ and the equality in (P2) holds only if V(p, x) = 0. According to the statement (S2) in Theorem 4.2.3, to complete the proof, it remains to show that γ is non-negative.

We argue that the following inequality holds for every \hat{p} in \mathbb{R}^n :

$$\gamma(\hat{p}) \ge \gamma(\mathbf{0}) \tag{C.3}$$

Then, without loss of generality by setting $\gamma(\mathbf{0}) = 0$, we conclude that $S_{EPT}(p, x) = \gamma(\hat{p})$ is non-negative. In what follows, we show that (C.3) is valid.

We first claim that (C.3) holds for all (p, x) in \mathbb{S} , where \mathbb{S} is the set of equilibrium points of (4.10). Note that for each x in \mathbb{X} , due to (**SPC**), it holds that $V(\mathbf{0}, x) = \mathbf{0}$, i.e., $\mathbf{0} \in \mathbb{S}_x$. By the fact that $\nabla_p \gamma(\hat{p}) = V(p, x)$, for fixed x, the following equality holds for all p in \mathbb{R}^n :

$$\gamma\left(\hat{p}\right) = \gamma\left(\mathbf{0}\right) + \int_{0}^{1} \dot{\mathbf{p}}^{T}(s) V(\mathbf{p}(s), x) \,\mathrm{d}s \tag{C.4}$$

where $\mathbf{p} : [0,1] \to \mathbb{R}$ is a parameterization of a piece-wise smooth path from 0 to p. According the path-connectedness assumption (see Section 4.1.2.2), for each p in \mathbb{S}_x , there is a path from 0 to p in which the entire path is contained in \mathbb{S}_x , i.e., $V(\mathbf{p}(s), x) = \mathbf{0}$ holds for all s in [0, 1]; hence

$$\gamma(\hat{p}) - \gamma(\mathbf{0}) = \int_0^1 \dot{\mathbf{p}}^T(s) V(\mathbf{p}(s), x) \, \mathrm{d}s = \mathbf{0}$$
 (C.5)

holds for every p in S_x . Since (C.5) holds for every (p, x) in S, this proves the claim.

To see that (C.3) also holds for (p, x) in $(\mathbb{R}^n \times \mathbb{X}) \setminus \mathbb{S}$, by contradiction, let us assume that there is $(p', x') \notin \mathbb{S}$ for which $S_{EPT}(p', x') = \gamma(\hat{p}') < \gamma(\mathbf{0})$. Let $x(\cdot)$ be a population state trajectory induced by (4.10) for the initial condition x(0) = x' and constant payoff p(t) = p' for all t in \mathbb{R}_+ . By (SPC) and (C.2b), the value of $S_{EPT}(p', x(t))$ is strictly decreasing unless V(p', x(t)) = 0. By the hypothesis that $S_{EPT}(p', x') < \gamma(0)$ and by (C.5), for every (p, x) in S, it holds that

$$S_{EPT}(p', x') < S_{EPT}(p, x)$$

and the trajectory $(p', x(\cdot))$ never converges to S. On the other hand, by LaSalle's Theorem [101], since p(t) is constant and the population state x(t) is contained in a compact set, x(t) converges to an invariant subset of $\{x \in \mathbb{X} \mid \nabla_x^T S_{EPT}(p', x) V(p', x) = 0\}$. By (SPC) and (C.2b), the invariant set is contained in S. This contradicts the fact that the trajectory $(p', x(\cdot))$ does not converge to S; hence $\gamma(\hat{p}) \geq \gamma(\mathbf{0})$ holds for all (p, x) in $(\mathbb{R}^n \times \mathbb{X}) \setminus \mathbb{S}$.

C.3 Proof of Proposition 4.2.7

The analysis used in Theorem 2.1 of [106] suggests that the following hold:

$$z^{T}\nabla_{p}\left[\max_{y\in\operatorname{int}(\mathbb{X})}(p^{T}y-v(y))\right] = z^{T}C(p)$$
(C.6a)

$$z^T \nabla_x v(x) = z^T p$$
 if and only if $C(p) = x$ (C.6b)

for all p in \mathbb{R}^n , x in \mathbb{X} , and z in $T\mathbb{X}$. Using (C.6), we can show that

$$\nabla_p S_{PBR}(p, x) = C(p) - x = V(p, x) \tag{C.7}$$

and

$$\nabla_x^T S_{PBR}(p, x) V(p, x) = -(p - \nabla_x v(x))^T V(p, x)$$
$$= -(\nabla_y v(y) - \nabla_x v(x))^T (y - x)$$
(C.8)

where y = C(p). By the fact that v is strictly convex, it holds that $\nabla_x^T S_{PBR}(p, x)V(p, x) \le 0$ where the equality holds only if V(p, x) = 0. According to Theorem 4.2.3, we conclude that the PBR dynamics are strictly passive.

Furthermore, if the perturbation v is strongly convex, i.e.,

$$\left(\nabla_y v(y) - \nabla_x v(x)\right)^T (y - x) \ge \eta' \cdot \|y - x\|^2$$

holds for all x, y in \mathbb{X} , then from (C.8), we can derive that

$$\nabla_x^T S_{PBR}(p, x) V(p, x) \le -\eta' \cdot \|V(p, x)\|^2 \tag{C.9}$$

Hence, by Theorem 4.2.3, we conclude that the PBR dynamics are strictly output passive and satisfies the passivity inequality (4.7) for $\eta = \eta'$.

C.4 Proof of Proposition 4.2.8

The first part of the statement directly follows from the condition (**P1**) and the fact that at a global minimizer (p^*, x^*) of S_{ED} , it holds that $\nabla_p S_{ED}(p^*, x^*) = 0$.

Now suppose that the dynamics satisfy (**NS**). To prove the second statement, it is sufficient to show that at each equilibrium point (p, x) of (4.2), it holds that $S_{ED}(p, x) =$ 0. To this end, let us consider an *anti-coordination game* whose payoff function is given by $F_{x_o}(x) = -(x - x_o)$ for a fixed $x_o \in \mathbb{X}$. Notice that x_o is a unique Nash equilibrium of the game. In what follows, we show that $S_{ED}(p_o, x_o) = 0$ holds for any choice of x_o from \mathbb{X} and p_o from \mathbb{S}_{x_o} .

Let (p^*, x^*) be a global minimizer of S_{ED} , i.e., $S_{ED}(p^*, x^*) = 0$. By the first part of the statement and (**NS**), we have that $V(s \cdot p^*, x^*) = 0$ for all s in \mathbb{R}_+ ; hence it holds that

$$S_{ED}(\mathbf{0}, x^*) = S_{ED}(p^*, x^*) - \int_0^1 (p^*)^T V(s \cdot p^*, x^*) \, \mathrm{d}s = 0$$

By the continuity of S_{ED} , for each $\epsilon > 0$ there exists $\delta > 0$ for which it holds that $S_{ED} (\delta \cdot F_{x_o}(x^*), x^*) < \epsilon.$

According to the passivity conditions (P1) and (P2) for $\eta = 0$, the following relation holds for every positive constant δ :

$$\frac{\mathrm{d}}{\mathrm{d}t} S_{ED}(\delta \cdot F_{x_o}(x(t)), x(t))$$

$$\leq \delta \cdot V^T(\delta \cdot F_{x_o}(x(t)), x(t)) DF_{x_0}(x(t)) \dot{x}(t)$$

$$= -\delta \cdot \|V(\delta \cdot F_{x_o}(x(t)), x(t))\|^2$$
(C.10)

where the trajectory $x(\cdot)$ starts from x^* . By an application of LaSalle's theorem [101] and by (NS), we can verify that $(\delta \cdot F_{x_o}(x(t)), x(t))$ converges to $(0, x_o)$ as $t \to \infty$. In addition, due to (C.10), we have that

$$S_{ED}\left(\mathbf{0}, x_{o}\right) \leq S_{ED}\left(\delta \cdot F_{x_{0}}(x^{*}), x^{*}\right) < \epsilon$$

Since this holds for every $\epsilon > 0$, we conclude that $S_{ED}(\mathbf{0}, x_o) = 0$. By the fact that $V(s \cdot p_o, x_o) = \mathbf{0}$ for all s in \mathbb{R}_+ if p_o belongs to \mathbb{S}_{x_o} , we can see that

$$S_{ED}(p_o, x_o) = S_{ED}(\mathbf{0}, x_o) + \int_0^1 p_o^T V(s \cdot p_o, x_o) \, \mathrm{d}s = 0$$
 (C.11)

holds for every p_o in \mathbb{S}_{x_o} .

Since the choice of x_o from X in constructing the game was arbitrary, we conclude that (C.11) holds for every (p_o, x_o) in S. This proves the Proposition.

C.5 Proof of Proposition 4.2.10

We first construct a game *F* based on the *Hypnodisk game* [104], which is described by the following payoff function:

$$\begin{pmatrix} H_1(x_1, x_2, x_3) \\ H_2(x_1, x_2, x_3) \\ H_3(x_1, x_2, x_3) \end{pmatrix} = \cos\left(\theta(x_1, x_2, x_3)\right) \begin{pmatrix} x_1 - \frac{1}{3} \\ x_2 - \frac{1}{3} \\ x_3 - \frac{1}{3} \end{pmatrix} + \frac{\sqrt{3}}{3} \sin\left(\theta(x_1, x_2, x_3)\right) \begin{pmatrix} x_2 - x_3 \\ x_3 - x_1 \\ x_1 - x_2 \end{pmatrix} + \frac{1}{3} \begin{pmatrix} 1 \\ 1 \\ 1 \\ 1 \end{pmatrix}$$

where

$$\theta(x_1, x_2, x_3) = \pi \cdot \left[1 - b \left(\sum_{i=1}^3 \left| x_i - \frac{1}{3} \right|^2 \right) \right]$$

and b is a bump function that satisfies

- 1. b(a) = 1 if $a \le R_I^2$
- 2. b(a) = 0 if $a \ge R_0^2$
- 3. b(a) is decreasing if $R_I^2 < a < R_O^2$

with $0 < R_I < R_O < \frac{1}{\sqrt{6}}$. Consider a payoff function F' given by

$$F'_{i}(x) = \begin{cases} H_{i}(x_{1}, x_{2}, x_{3} + \dots + x_{n}) & \text{if } i \in \{1, 2\} \\ H_{3}(x_{1}, x_{2}, x_{3} + \dots + x_{n}) & \text{otherwise} \end{cases}$$
(C.12)

Note that the set of Nash equilibria for F' is given by

$$\left\{ x \in \mathbb{X} \mid x_1 = x_2 = x_3 + \dots + x_n = \frac{1}{3} \right\}$$
(C.13)

Since θ is a smooth function, F' is continuously differentiable, and its differential map DF' is bounded, i.e., for some $\delta > 0$, it holds that $z^T DF'(x)z < \delta \cdot z^T z$ for all x in \mathbb{X} and z in $T\mathbb{X}$. Finally, for a given constant $\nu > 0$, we define a new payoff function by $F_{\nu} = \frac{\nu}{\delta} \cdot F'$.

Using the payoff function F_{ν} , we prove the statement of the Proposition. By contradiction, suppose that there is a payoff monotonic dynamic, i.e., the dynamic satisfies both (**NS**) and (**PC**), that is strictly output passive. By definition, the dynamic satisfies the passivity inequality (4.7) for $\eta > 0$. Consider a population game described by F_{ν} in which $\nu < \eta$ holds. Then the time derivative of the storage function $S_{ED}(F_{\nu}(x(t)), x(t))$ satisfies

$$\frac{\mathrm{d}}{\mathrm{d}t} S_{ED}(F_{\nu}(x(t)), x(t)) \le -(\eta - \nu) \cdot \|V(F_{\nu}(x(t)), x(t))\|$$

By an application of LaSalle's theorem [101] and by (NS), we can verify that x(t) converges to the set of Nash equilibria given as in (C.13).

On the other hand, when x(t) is contained in the set

$$\left\{ x \in \mathbb{X} \mid \left(x_1 - \frac{1}{3} \right)^2 + \left(x_2 - \frac{1}{3} \right)^2 + \left(x_3 + \dots + x_n - \frac{1}{3} \right)^2 \le R_I^2 \right\}$$

by the condition (PC), it holds that

$$F_{\nu}^{T}(x(t))V(F_{\nu}(x(t)), x(t)) = \frac{\nu}{2\delta} \cdot \frac{\mathrm{d}}{\mathrm{d}t} \left[\left(x_{1}(t) - \frac{1}{3} \right)^{2} + \left(x_{2}(t) - \frac{1}{3} \right)^{2} + \left(x_{3}(t) + \dots + x_{n}(t) - \frac{1}{3} \right)^{2} \right] \ge 0$$
(C.14)

Hence, the population state x(t) never converges to the set of Nash equilibria. This is a contradiction. Therefore, the dynamic cannot be strictly output passive.

C.6 Proof of Proposition 4.2.12

The sufficiency directly follows from the choice of $E_{ED} = S_{ED}$ and the inequalities in (4.7) and (4.16), where S_{ED} is a storage function of the passive evolutionary dynamic. To prove the necessity, we consider a set of population games represented by cumulative payoff functions $\dot{p} = Ax$, where A is a symmetric matrix in $\mathbb{R}^{n \times n}$. It can be verified that each game satisfies (4.16) for a C^1 function

$$S_G(x) = \frac{1}{2} \max_{x \in \mathbb{X}} x^T A x - \frac{1}{2} x^T A x$$

Let us select a candidate Lyapunov function $E = S_G + E_{ED}$ for a C^1 function $E_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$. The time derivative of E leads to the following:

$$\frac{\mathrm{d}}{\mathrm{d}t}E(p,x) = \nabla_x^T S_G(x)\dot{x} + \nabla_p^T E_{ED}(p,x)\dot{p} + \nabla_x^T E_{ED}(p,x)\dot{x}$$
$$= -V^T(p,x)Ax + \nabla_p^T E_{ED}(p,x)Ax + \nabla_x^T E_{ED}(p,x)V(p,x)$$
$$= (\nabla_p E_{ED}(p,x) - V(p,x))^T Ax + \nabla_x^T E_{ED}(p,x)V(p,x) \le 0 \qquad (C.15)$$

Since x belongs to X and A could be any symmetric matrix in $\mathbb{R}^{n \times n}$, the inequality in (C.15) holds for every choice of A if and only if $\nabla_p E_{ED}(p, x) = V(p, x)$ and $\nabla_x^T E_{ED}(p, x)V(p, x) \leq 0$ hold. By Theorem 4.2.3, we conclude that the dynamic is passive with a storage function $S_{ED} = E_{ED}$.

C.7 Proof of Corollary 4.2.13

The sufficiency directly follows from the definition of passivity. To prove the necessity, by contradiction, suppose that the dynamic is not passive. Then, by Proposition 4.2.12, for any choice of S_{ED} , we can construct a game identified by a cumulative payoff function given as in (4.14) for which

$$S_G(x(t)) + S_{ED}(p(t), x(t)) > S_G(x(t_0)) + S_{ED}(p(t_0), x(t_0))$$
(C.16)

holds, where S_G is given in (4.15). According to (4.16), this yields that

$$S_{ED}(p(t), x(t)) > S_{ED}(p(t_0), x(t_0)) + \int_{t_0}^t F^T(x(\tau))\dot{x}(\tau) \,\mathrm{d}\tau$$
(C.17)

which contradicts the fact that the dynamic satisfies the passivity inequality (4.7) for every cumulative payoff function (4.14).

C.8 Proof of Proposition 4.2.14

As the revision protocol depends only on the the gradient $\nabla_x u(p, x) = p - \nabla_x v(x)$ and the population state x, we represent the revision protocol and vector field of the dynamic as ϱ_{ji} ($\nabla_x u(p, x), x$) and V ($\nabla_x u(p, x), x$), respectively. Since unperturbed dynamics are passive, by Theorem 4.2.3, we can find a storage function $S_{ED} : \mathbb{R}^n \times \mathbb{X} \to \mathbb{R}_+$ for which the conditions (**P1**) and (**P2**) hold for $\eta = 0$. In what follows, we show that the resulting perturbed dynamics are strictly passive with a storage function

$$\widetilde{S}_{ED}(p,x) \stackrel{def}{=} S_{ED}(p - \nabla_x v(x), x)$$

We first compute the gradient of \tilde{S}_{ED} with respect to p and x as follows:

$$\nabla_{p}\widetilde{S}_{ED}(p,x) = \nabla_{p'}S_{ED}(p',x)\Big|_{p'=p-\nabla_{x}v(x)} = V\left(p-\nabla_{x}v(x),x\right)$$
(C.18)

and

$$\nabla_{x}\widetilde{S}_{ED}(p,x)$$

$$= \nabla_{x}^{T}(p - \nabla_{x}v(x)) \left. \nabla_{p'}S_{ED}(p',x) \right|_{p'=p-\nabla_{x}v(x)} + \nabla_{x}S_{ED}(p',x) \right|_{p'=p-\nabla_{x}v(x)}$$

$$= -\left(\nabla_{x}^{2}v(x)\right)^{T}V(p - \nabla_{x}v(x),x) + \nabla_{x}S_{ED}(p',x) \Big|_{p'=p-\nabla_{x}v(x)}$$
(C.19)

Using (C.19), we can derive the following:

$$\nabla_x^T \widetilde{S}_{ED}(p, x) V\left(p - \nabla_x v(x), x\right)$$

$$= -V^T \left(p - \nabla_x v(x), x\right) \nabla_x^2 v(x) V\left(p - \nabla_x v(x), x\right) + \nabla_x^T S_{ED}(p', x) V(p', x) \Big|_{p'=p - \nabla_x v(x)}$$

$$\stackrel{(1)}{\leq} -V^T \left(p - \nabla_x v(x), x\right) \nabla_x^2 v(x) V\left(p - \nabla_x v(x), x\right)$$
(C.20)

where (1) is due to passivity of the unperturbed dynamics. Since v satisfies $z^T \nabla_x^2 v(x) z > 0$ for all x in \mathbb{X} and nonzero z in $T\mathbb{X}$,

$$\nabla_x^T \widetilde{S}_{ED}(p, x) V\left(p - \nabla_x v(x), x\right) = 0$$

holds only if $V(p - \nabla_x v(x), x) = 0$. Using Theorem 4.2.3, we can see that the perturbed dynamics are strictly passive.

Now suppose that v is strongly convex satisfying

$$z^T \nabla_x^2 v(x) z \ge \eta \cdot z^T z \tag{C.21}$$

for all x in X and z in TX, where η' is a positive constant. From (C.20), we can see that

$$\nabla_x^T \widetilde{S}_{ED}(p, x) V\left(p - \nabla_x v(x), x\right)$$

$$\leq -\eta \cdot V^T \left(p - \nabla_x v(x), x\right) V\left(p - \nabla_x v(x), x\right)$$

holds. Hence, from Theorem 4.2.3, we conclude that the perturbed dynamics are strongly output passive. $\hfill \Box$

C.9 Proof of Theorem 4.3.1

First of all, we note that for any sequence $\{(p^{(l)}, x^{(l)})\}_{l \in \mathbb{N}}$ in $\mathbb{R}^n \times \mathbb{X}$, if

$$\nabla_x^T S_{ED}\left(p^{(l)}, x^{(l)}\right) V(p^{(l)}, x^{(l)}) + \nu \cdot V^T\left(p^{(l)}, x^{(l)}\right) V\left(p^{(l)}, x^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0$$

then, in both (CL1) and (CL2), we can see that $\nabla_x^T S_{PBR}(p^{(l)}, x^{(l)}) V(p^{(l)}, x^{(l)}) \stackrel{l \to \infty}{\longrightarrow} 0.$ Hence, in conjunction with (A2), we have that the following relation: For any sequence $\{(p^{(l)}, x^{(l)})\}_{l \in \mathbb{N}}$ in $\mathbb{R}^n \times \mathbb{X}$,

(A2')
$$\nabla_x^T S_{ED}\left(p^{(l)}, x^{(l)}\right) V\left(p^{(l)}, x^{(l)}\right) + \nu \cdot V^T\left(p^{(l)}, x^{(l)}\right) V\left(p^{(l)}, x^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0$$

implies $S_{ED}\left(p^{(l)}, x^{(l)}\right) \stackrel{l \to \infty}{\longrightarrow} 0$

To prove the Theorem, for each $\epsilon > 0$ let us define a set given by

$$\mathbb{O}_{\epsilon} \stackrel{def}{=} \left\{ t > 0 \; \middle| \; S_{ED}(p(t), x(t)) > \frac{\epsilon}{2} \right\}$$

Since $S_{ED}(p(t), x(t)) > \frac{\epsilon}{2}$ holds for all t in \mathbb{O}_{ϵ} , by the contrapositive of (A2'), there exists $\delta_1 > 0$ for which

$$\nabla_x^T S_{ED}(p(t), x(t)) V(p(t), x(t)) + \nu \cdot V^T(p(t), x(t)) V(p(t), x(t)) \le -\delta_1$$

holds¹ for all t in \mathbb{O}_{ϵ} . Note that using (4.22), we can derive the following relations:

$$S_{ED}(p(t), x(t)) - S_{ED}(p(0), x(0)) - \alpha$$

$$\leq \int_{0}^{t} \frac{\mathrm{d}}{\mathrm{d}\tau} S_{ED}(p(\tau), x(\tau)) \,\mathrm{d}\tau + \int_{0}^{t} \left[-\dot{p}^{T}(\tau) \dot{x}(\tau) + \nu \cdot \dot{x}^{T}(\tau) \dot{x}(\tau) \right] \,\mathrm{d}\tau$$

$$= \int_{0}^{t} \left[\nabla_{x}^{T} S_{ED}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) + \nu \cdot V^{T}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) \right] \,\mathrm{d}\tau$$

(C.22)

Since S_{ED} is a non-negative function, we can infer that (C.22) is lower-bounded by $-S_{ED}(p(0), x(0)) - \alpha$ for all $t \ge 0$, which yields that

$$-S_{ED}(p(0), x(0)) - \alpha$$

$$\leq \int_{0}^{\infty} \left[\nabla_{x}^{T} S_{ED}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) + \nu \cdot V^{T}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) \right] d\tau$$

$$\leq \int_{\mathbb{O}_{\epsilon}} \left[\nabla_{x}^{T} S_{ED}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) + \nu \cdot V^{T}(p(\tau), x(\tau)) V(p(\tau), x(\tau)) \right] d\tau$$

$$\leq -\delta_{1} \cdot \mathcal{L}(\mathbb{O}_{\epsilon})$$
(C.23)

where $\mathcal{L}(\mathbb{O}_{\epsilon})$ is the Lebesgue measure of \mathbb{O}_{ϵ} . Hence, we have that $\mathcal{L}(\mathbb{O}_{\epsilon}) \leq \frac{S_{ED}(p(0),x(0)) + \alpha}{\delta_1}$.

Since \mathbb{O}_{ϵ} is an open set, we can represent \mathbb{O}_{ϵ} as an union of disjoint open intervals, i.e., $\mathbb{O}_{\epsilon} = \bigcup_{i \in \mathbb{N}} \mathbb{I}_i$ where $\{\mathbb{I}_i\}_{i \in \mathbb{N}}$ is a set of disjoint open intervals. Notice that by our construction of \mathbb{O}_{ϵ} , by letting $\mathbb{I}_i = (a_i, b_i)$, we have that $S_{ED}(p(a_i), x(a_i)) \leq \frac{\epsilon}{2}$ and $S_{ED}(p(t), x(t)) > \frac{\epsilon}{2}$ for $t \in \mathbb{I}_i$. Since \mathbb{O}_{ϵ} has finite Lebesgue measure, it holds that $\lim_{i \to \infty} \mathcal{L}(\mathbb{I}_i) = 0.$

$$\nabla_x^T S_{ED}(\boldsymbol{p}(t), \boldsymbol{x}(t)) V(\boldsymbol{p}(t), \boldsymbol{x}(t)) + \boldsymbol{\nu} \cdot \boldsymbol{V}^T(\boldsymbol{p}(t), \boldsymbol{x}(t)) V(\boldsymbol{p}(t), \boldsymbol{x}(t)) \leq 0$$

¹Notice that under the configurations as in (CL1) or (CL2), it holds that

In what follows, we show that for each $\epsilon > 0$, there exists $T_{\epsilon} > 0$ for which $S_{ED}(p(t), x(t)) < \epsilon$ holds for all $t \ge T_{\epsilon}$, and we conclude that $\lim_{t\to\infty} S_{ED}(p(t), x(t)) = 0$. To achieve this, by contradiction, suppose that there exists an infinite subset \mathbb{J} of \mathbb{N} for which

$$\max_{t \in cl(\mathbb{I}_j)} S_{ED}(p(t), x(t)) \ge \epsilon$$

for each j in \mathbb{J} , where $\operatorname{cl}(\mathbb{I}_j)$ is the closure of \mathbb{I}_j . Let $\overline{t}_j \in \operatorname{cl}(\mathbb{I}_j)$ be for which

$$S_{ED}(p(\bar{t}_j), x(\bar{t}_j)) = \max_{t \in cl(\mathbb{I}_j)} S_{ED}(p(t), x(t))$$

holds. By letting $\mathbb{I}_j = (a_j, b_j)$, we can derive the following:

$$S_{ED}(p(\bar{t}_j), x(\bar{t}_j)) - S_{ED}(p(a_j), x(a_j))$$

$$= \int_{a_j}^{\bar{t}_j} \frac{\mathrm{d}}{\mathrm{d}\tau} S_{ED}(p(\tau), x(\tau)) \,\mathrm{d}\tau$$

$$\stackrel{(1)}{\leq} \int_{a_j}^{\bar{t}_j} \dot{p}^T(\tau) V(p(\tau), x(\tau)) \,\mathrm{d}\tau$$

$$\stackrel{(2)}{<} M \cdot \delta_2 \cdot \mathcal{L}(\mathbb{I}_j) \qquad (C.24)$$

The inequality (1) can be derived using Theorem 4.2.3. To see that (2) holds, recall that from (C.22), $S_{ED}(p(t), x(t)) \leq S_{ED}(p(0), x(0)) + \alpha$ holds for all $t \geq 0$, and that \dot{p} is bounded, i.e, there is a positive real M for which $\|\dot{p}(t)\| < M$ holds for all t in \mathbb{R}_+ . Hence, according to the contrapositive of (A1), we can derive $\dot{p}^T(\tau)V(p(\tau), x(\tau)) < M \cdot \delta_2$ for some $\delta_2 > 0$, which yields (2).

Since $S_{ED}(p(a_j), x(a_j)) \leq \frac{\epsilon}{2}$ and $\lim_{j\to\infty} \mathcal{L}(\mathbb{I}_j) = 0$, from (C.24), we can see that $S_{ED}(p(\bar{t}_j), x(\bar{t}_j)) < \epsilon$ for sufficiently large j in \mathbb{J} . This contradicts the hypothesis that $S_{ED}(p(\bar{t}_j), x(\bar{t}_j)) \geq \epsilon$ holds for all j in \mathbb{J} . Since this holds for every $\epsilon > 0$, we can infer that for each $\epsilon > 0$, there exists $T_{\epsilon} > 0$ for which $S_{ED}(p(t), x(t)) < \epsilon$ holds for all $t \ge T_{\epsilon}$.

C.10 Proof of Proposition 4.3.2

Suppose that $\nu_2 > 0$. We proceed by showing that the following relations hold:

$$\int_{0}^{t} \dot{p}^{T}(\tau) \dot{x}(\tau) d\tau
= \int_{0}^{t} \dot{x}^{T}(\tau) DF_{1}^{T}(x(\tau)) \dot{x}(\tau) d\tau + \int_{0}^{t} \dot{x}^{T}(\tau-d) DF_{2}^{T}(x(\tau-d)) \dot{x}(\tau) d\tau
\stackrel{(1)}{\leq} \nu_{1} \int_{0}^{t} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau + \frac{\nu_{2}}{2} \int_{0}^{t} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau
+ \frac{1}{2\nu_{2}} \int_{0}^{t} \dot{x}^{T}(\tau-d) DF_{2}^{T}(x(\tau-d)) DF_{2}(x(\tau-d)) \dot{x}(\tau-d) d\tau
\stackrel{(2)}{\leq} \nu_{1} \int_{0}^{t} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau + \frac{\nu_{2}}{2} \int_{0}^{t} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau + \frac{\nu_{2}}{2} \int_{0}^{t} \dot{x}^{T}(\tau-d) \dot{x}(\tau-d) d\tau
\leq (\nu_{1}+\nu_{2}) \int_{0}^{t} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau + \frac{\nu_{2}}{2} \int_{-d}^{0} \dot{x}^{T}(\tau) \dot{x}(\tau) d\tau \qquad (C.25)$$

To show (1), we use (4.25a) and the following fact:

$$\dot{x}^{T}(\tau)DF_{2}(x(\tau-d))\dot{x}(\tau-d)$$

$$\leq \frac{\nu_{2}}{2}\dot{x}^{T}(\tau)\dot{x}(\tau) + \frac{1}{2\nu_{2}}\dot{x}^{T}(\tau-d)DF_{2}^{T}(x(\tau-d))DF_{2}(x(\tau-d))\dot{x}(\tau-d)$$

To see (2), we use (4.25b). Then, by defining $\alpha = \frac{\nu_2}{2} \cdot \int_{-d}^{0} \dot{x}^T(\tau) \dot{x}(\tau) d\tau$, we can see that the payoff function (4.24) satisfies (4.22) with $\nu = \nu_1 + \nu_2$.

If $\nu_2 = 0$ then we can see that $DF_2(x)z = 0$ for all x in \mathbb{X} and z in $T\mathbb{X}$. Based on the above analysis, by defining $\alpha = 0$, we can conclude that (4.24) satisfies (4.22) with $\nu = \nu_1$.

C.11 Proof of Proposition 4.3.3

First of all, we note that with a choice of P as $\hat{\mathbf{p}}(s) = s \cdot \hat{p}$, according to (**A**) of the EPT dynamics, we can see that $S(\hat{p}) = \int_0^1 \hat{p}^T \varrho(s \cdot \hat{p}) \, ds \ge 0$. We proceed the proof by choosing a storage function for the dynamics as $S_{EPT}(p, x) = S(\hat{p})$, and show that (**A1**) and (**A2**) of Theorem 4.3.1 hold. To start with, we verify that the contrapositive of (**A1**) is true. Notice that

$$\|V(p,x)\| = \left\| \varrho(\hat{p}) - x \mathbf{1}^T \varrho(\hat{p}) \right\|$$

$$\leq \|\varrho(\hat{p})\| + \|x\| \left| \mathbf{1}^T \varrho(\hat{p}) \right|$$

$$\leq (\sqrt{n} + 1) \cdot \|\varrho(\hat{p})\| \qquad (C.26)$$

where the last inequality follows from the Cauchy-Schwarz inequality. Suppose that there exists $\epsilon > 0$ for which $S_{EPT}(p, x) < \epsilon$ holds. By the contrapositive of (C1), there exists $\delta' > 0$ for which $\|\varrho(\hat{p})\| < \delta'$ holds. By (C.26), it holds that $\|V(p, x)\| < (\sqrt{n}+1)\cdot\delta' = \delta$; hence the contrapositive of (A1) holds.

To verify that (A2) of Theorem 4.3.1 holds, notice that

$$\nabla_x^T S_{EPT}(p, x) V(p, x) = -\left(\mathbf{1}^T \varrho(\hat{p})\right) \left(\hat{p}^T \varrho(\hat{p})\right)$$

We claim that if $S_{EPT}(p, x) \ge \epsilon$ holds for a positive real ϵ , then there exist $\delta_1, \delta_2 > 0$ for which $\mathbf{1}^T \varrho(\hat{p}) \ge \delta_1$ and $\hat{p}^T \varrho(\hat{p}) \ge \delta_2$ hold. Hence we have that

$$\nabla_x^T S_{EPT}(p, x) V(p, x) \le -\delta_1 \cdot \delta_2 = -\delta$$

This proves that the contrapositive of (A2) is true. We prove the claim by contradiction. Suppose that there exists a sequence $\{(p^{(l)}, x^{(l)})\}_{l \in \mathbb{N}} \subset \mathbb{R}^n \times \mathbb{X}$ for which either $\lim_{l\to\infty} \mathbf{1}^T \varrho\left(\hat{p}^{(l)}\right) = 0 \text{ or } \lim_{i\to\infty} \left(\hat{p}^{(l)}\right)^T \varrho\left(\hat{p}^{(l)}\right) = 0 \text{ or both and } S_{EPT}(p^{(l)}, x^{(l)}) \ge \epsilon$ hold for all $l \in \mathbb{N}$. Then it follows from (C2) that $\lim_{l\to\infty} S_{EPT}\left(p^{(l)}, x^{(l)}\right) = 0$, which contradicts our hypothesis that $S_{EPT}(p^{(l)}, x^{(l)}) \ge \epsilon$ holds for all $l \in \mathbb{N}$. \Box

C.12 Proof of Proposition 4.3.5

According to Proposition 4.2.6, we can choose a storage function for the pairwise comparison dynamics as

$$S_{PC}(p,x) = \sum_{i=1}^{n} \sum_{j=1}^{n} S_j(p_j - p_i, x_i)$$
(C.27)

The partial derivative of (C.27) with respect to x satisfies²

$$\nabla_x^T S_{PC}(p,x) V(p,x) = \sum_{i=1}^n \sum_{j=1}^n \left[x_i \varrho_j (p_j - p_i) \sum_{k=1}^n \int_{p_k - p_i}^{p_k - p_j} \varrho_k(s) \, \mathrm{d}s \right]$$
(C.28)

To prove the Proposition, we show that (A1) and (A2) of Theorem 4.3.1 hold. We first show that the contrapositive of (A1) of Theorem 4.3.1 is true. Suppose that for some $\epsilon > 0$, it holds that $S_{PC}(p, x) < \epsilon$. Then $S_j(p_j - p_i, x_i) < \epsilon$ holds for all i, j in $\{1, \dots, n\}$, and by (C1), there exists $\delta' > 0$ for which $x_i \varrho_j (p_j - p_i) < \delta'$ holds for all $i, j \in \{1, \dots, n\}$. From the following inequality

$$|V_i(p,x)| \le \sum_{j=1}^n x_j \varrho_i(p_i - p_j) + \sum_{j=1}^n x_i \varrho_j(p_j - p_i)$$
(C.29)

we have that $|V_i(p, x)| < 2n \cdot \delta'$ which implies that $||V(p, x)|| < 2n^{3/2} \cdot \delta' = \delta$. This shows that the contrapositive of (A1) is true.

Next, we consider the contrapositive of (A2). Suppose that $S_{PC}(p, x) \ge \epsilon$ for some ²This fact directly follows from the analysis in the proof of Theorem 7.1 in [14]

 $\epsilon > 0$. We note that according to (C.27), there exists $\delta_1 > 0$, e.g., $\delta_1 = \frac{\epsilon}{n^2}$, for which

$$\max_{i,j\in\{1,\cdots,n\}} S_j \left(p_j - p_i, x_i \right) = \max_{i,j\in\{1,\cdots,n\}} x_i \int_0^{p_j - p_i} \varrho_j(s) \, \mathrm{d}s \ge \delta_1 \tag{C.30}$$

holds. Let $i^*, j^* \in \{1, \cdots, n\}$ be indices that satisfy

$$x_{i^*} \int_0^{p_{j^*} - p_{i^*}} \varrho_{j^*}(s) \, \mathrm{d}s = \max_{i, j \in \{1, \cdots, n\}} x_i \int_0^{p_j - p_i} \varrho_j(s) \, \mathrm{d}s \tag{C.31}$$

Then by the contrapositive of (C2), there exists $\delta_2 > 0$ for which it holds that

$$\varrho_{j^*}(p_{j^*} - p_{i^*}) \ge \delta_2 \tag{C.32}$$

Since $x_i \varrho_j (p_j - p_i) \int_{p_k - p_i}^{p_k - p_j} \varrho_k(s) ds \le 0$ holds, according to (C.28), we can derive

the following:

$$\nabla_x^T S_{PC}(p, x) V(p, x) \le x_{i^*} \varrho_{j^*}(p_{j^*} - p_{i^*}) \int_{p_{k^*} - p_{i^*}}^{p_{k^*} - p_{j^*}} \varrho_{k^*}(s) \,\mathrm{d}s$$

for any i^*, j^*, k^* in $\{1, \dots, n\}$. In particular, we choose i^*, j^* that satisfy (C.31) and $k^* = j^*$. Then, in conjunction with (C.30) and (C.32), we have that

$$\nabla_x^T S_{PC}(p, x) V(p, x) \le x_{i^*} \varrho_{j^*}(p_{j^*} - p_{i^*}) \int_{p_{j^*} - p_{i^*}}^0 \varrho_{j^*}(s) \, \mathrm{d}s \le -\delta_1 \cdot \delta_2 = -\delta_1 \cdot \delta_2$$

This verifies (A2) of Theorem 4.3.1. This completes the proof of the Proposition. \Box

C.13 Proof of Proposition 4.3.7

From Proposition 4.2.7, we have seen that

$$S_{PBR}(p,x) = \max_{y \in \text{int}(\mathbb{X})} \left(p^T y - v(y) \right) - \left(p^T x - v(x) \right)$$

is a storage function of the PBR dynamics. We verify that the assumptions (A1) and (A2) of Theorem 4.3.1 hold.

The verification of (A1) directly follows from the facts that $||V(p, x)|| \le ||C(p)|| + ||x||$ and C(p) belongs to X for every p in \mathbb{R}^n . To show (A2), consider the following relations:

$$S_{PBR}(p,x) = (p^{T}C(p) - v(C(p))) - (p^{T}x - v(x))$$

= $p^{T}(C(p) - x) - (v(C(p)) - v(x))$
 $\stackrel{(1)}{\leq} (p - \nabla_{x}v(x))^{T}(C(p) - x)$
 $\stackrel{(2)}{=} -\nabla_{x}^{T}S_{PBR}(p,x)V(p,x)$ (C.33)

To obtain (1), we use the fact that v is a convex function; and to show (2), we use (C.8) in the proof of Proposition 4.2.7. Therefore, from (C.33), we conclude that (A2) is true. \Box

Bibliography

- [1] M. Emelianenko, L. Ju, and A. Rand, "Nondegeneracy and weak global convergence of the Lloyd algorithm in \mathbb{R}^D ," *SIAM Journal on Numerical Analysis*, vol. 46, Mar. 2008.
- [2] Q. Du, M. Emelianenko, and L. Ju, "Convergence of the Lloyd algorithm for computing centroidal voronoi tesellations," *SIAM Journal on Numerical Analysis*, vol. 44, 2006.
- [3] Q. Du, V. Faber, and M. Gunzburger, "Centroidal voronoi tessellations: Applications and algorithms," *SIAM Review*, vol. 41, Dec. 1999.
- [4] J. W. Weibull, *Evolutionary Game Theory*. MIT Press, 1995.
- [5] J. Hofbauer and K. Sigmund, "Evolutionary game dynamics," *Bulletin of the American Mathematical Society*, vol. 40, no. 4, pp. 479–519, July 2003.
- [6] J. C. Willems, "The generation of lyapunov functions for input-output stable systems," *SIAM Journal on Control*, vol. 9, no. 1, Feb. 1971.
- [7] A. van der Schaft, *L2-Gain and Passivity Techniques in Nonlinear Control.* Springer, 2000.
- [8] G. W. Brown and J. von Neumann, "Solutions of games by differential equations," in *Contributions to the Theory of Games I*. Princeton University Press, 1950, pp. 73–79.
- [9] P. D. Taylor and L. B. Jonker, "Evolutionarily stable strategies and game dynamics," *Mathematical Biosciences*, vol. 40, 1978.
- [10] E. Zeeman, "Population dynamics from game theory," in *Global Theory of Dynamical Systems*, ser. Lecture Notes in Mathematics, Z. Nitecki and C. Robinson, Eds. Springer Berlin Heidelberg, 1980, vol. 819, pp. 471–497.
- [11] I. Gilboa and A. Matsui, "Social stability and equilibrium," *Econometrica*, vol. 59, no. 3, May 1991.

- [12] J. M. Swinkels, "Adjustment dynamics and rational play in games," *Games and Economic Behavior*, vol. 5, July 1993.
- [13] K. Ritzberger and J. W. Weibull, "Evolutionary selection in normal-form games," *Econometrica*, vol. 63, no. 6, Nov. 1995.
- [14] J. Hofbauer and W. H. Sandholm, "Stable games and their dynamics," *Journal of Economic Theory*, vol. 144, 2009.
- [15] M. J. Fox and J. S. Shamma, "Population games, stable games, and passivity," *Games*, vol. 2, 2013.
- [16] E. Wei, E. W. Justh, and P. Krishnaprasad, "Pursuit and an evolutionary game," Proceedings of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 2009.
- [17] D. Pais and N. E. Leonard, "Pursuit and evasion: Evolutionary dynamics and collective motion," in AIAA Guidance, Navigation, and Control Conference, Aug. 2010.
- [18] K. Binmore and L. Samuelson, "Evolutionary drift and equilibrium selection," *Review of Economic Studies*, pp. 363–393, 1999.
- [19] M. A. nowak and K. Sigmund, "Evolutionary dynamics of biological games," Science, vol. 303, Feb. 2004.
- [20] S. Park and N. C. Martins, "An augmented observer for the distributed estimation problem for LTI systems," in 2012 American Control Conference, June 2012.
- [21] R. Olfati-Saber, "Distributed Kalman filter with embedded consensus filters," in 44th IEEE Conference on Decision and Control, and the European Control Conference, Dec. 2005.
- [22] —, "Distributed Kalman filtering for sensor networks," in 46th IEEE Conference on Decision and Control, Dec. 2007, pp. 5492–5498.
- [23] M. S. Mahmoud and H. M. Khalid, "Distributed Kalman filtering: a bibliographic review," *IET Control Theory and Applications*, vol. 7, Mar. 2013.
- [24] R. Olfati-Saber, J. A. Fax, and R. M. Murray, "Consensus and cooperation in networked multi-agent systems," *Proceedings of the IEEE*, vol. 95, no. 1, Jan. 2007.
- [25] R. Carli, A. Chiuso, L. Schenato, and S. Zampieri, "Distributed Kalman filtering based on consensus strategies," *IEEE Journal on Selected Areas in Communication*, no. 4, May 2008.
- [26] R. Olfati-Saber, "Kalman-consensus filter: Optimality, stability, and performance," in *Joint 48th IEEE Conference on Decision and Control and 28th Chinese Control Conference*, Dec. 2009.

- [27] M. Kamgarpour and C. Tomlin, "Convergence properties of a decentralized Kalman filter," in 47th IEEE Conference on Decision and Control, Dec. 2008.
- [28] U. A. Khan and A. Jadbabaie, "On the stability and optimality of distributed Kalman filters with finite-time data fusion," in *2011 American Control Conference*, June 2011.
- [29] P. Alriksson and A. Rantzer, "Distributed Kalman filtering using weighted averaging," in *In Proceedings of the 17th International Symposium on Mathematical Theory of Networks and Systems*, 2006.
- [30] F. S. Cattivelli and A. H. Sayed, "Diffusion strategies for distributed Kalman filtering and smoothing," *IEEE Transactions on Automatic Control*, vol. 55, no. 9, Sept. 2010.
- [31] U. A. Khan, S. Kar, A. Jadbabaie, and J. M. F. Moura, "On connectivity, observability, and stability in distributed estimation," in *49th IEEE Conference on Decision and Control*, Dec. 2010.
- [32] S. Kar and J. M. F. Moura, "Gossip and distributed kalman filtering: Weak consensus under weak detectability," *IEEE Trans. Signal Processing*, vol. 59, no. 4, Apr. 2011.
- [33] H. Bai, R. A. Freeman, and K. M. Lynch, "Distributed Kalman filtering using the internal model average consensus estimator," in 2011 American Control Conference, June 29 - July 01 2011, pp. 1500–1505.
- [34] U. A. Khan and A. Jadbabaie, "Coordinated networked estimation strategies using structured systems theory," in 2011 50th IEEE Conference on Decision and Control and European Control Conference, Dec. 2011.
- [35] I. Matei and J. S. Baras, "Consensus-base linear distributed filtering," *Automatica*, Aug. 2012.
- [36] S. Das and J. M. F. Moura, "Distributed state estimation in multi-agent networks," in 2013 IEEE International Conference on Acoustics, Speech and Signal Processing, May 2013.
- [37] U. A. Khan and J. M. F. Moura, "Distributing the Kalman filter for large-scale systems," *IEEE Trans. Signal Processing*, vol. 56, no. 10, Oct. 2008.
- [38] S. S. Stanković, M. S. Stanković, and D. M. Stipanović, "Consensus based overlapping decentralized estimator," *IEEE Trans. Automat. Contr.*, vol. 54, no. 2, Feb. 2009.
- [39] M. Farina, G. Ferrari-Trecate, and R. Scattolini, "Distributed moving horizon estimation for linear constrained systems," *IEEE Trans. Automat. Contr.*, vol. 55, pp. 2462–2475, Nov. 2010.

- [40] A. S. M. Vamsi and N. Elia, "Design of distributed estimators over arbitrary causal networks," in 2011 50th IEEE Conference on Decision and Control and European Control Conference, Dec. 2011.
- [41] L. Orihuela, P. Millán, C. Vivas, and F. R. Rubio, "Reduced-order H_2/H_{∞} distributed observer for sensor networks," *International Journal of Control*, vol. 86, no. 10, June 2013.
- [42] V. Ugrinovskii, "Conditions for dectectability in distributed consensus-based observer networks," in *51st IEEE Conference on Decision and Control*, Dec. 2012.
- [43] M. Doostmohammadian and U. A. Khan, "On the genericity properties in distributed estimation: Topology design and sensor placement," *IEEE Journal of Selected Topics in Signal Processing*, vol. 7, no. 2, Apr. 2013.
- [44] S. Park and N. C. Martins, "Necessary and sufficient conditions for the stabilizability of a class of LTI distributed observers," in 2012 IEEE Conference on Decision and Control, Dec. 2012.
- [45] A. Gopalakrishnan, N. S. Kaisare, and S. Narasimhan, "Incorporating delayed and infrequent measurements in extended Kalman filter based nonlinear state estimation," *Journal of Process Control*, pp. 119–129, 2011.
- [46] W. Ren, R. W. Beard, and E. M. Atkins, "A survey of consensus problems in multiagent coordination," in *American Control Conference*, June 2005.
- [47] R. Olfati-Saber, J. A. Fax, and R. M. Murray, "Consensus and cooperation in networked multi-agent systems," *Proceedings of the IEEE*, vol. 95, no. 1, Jan. 2007.
- [48] A. Arenas, A. Díaz-Guilera, J. Kurths, Y. Moreno, and C. Zhou, "Synchronization in complex networks," *Physics Reports*, vol. 469, Dec. 2008.
- [49] W. Ren and Y. Cao, *Distributed Coordination of Multi-agent Networks*. Springer, 2011.
- [50] X. Li, X. Wang, and G. Chen, "Pinning a complex dynamical network to its equilibrium," *IEEE Trans. Circuits Syst. I*, vol. 51, no. 10, Oct. 2004.
- [51] T. Liu, D. J. Hill, and J. Zhao, "Synchronization of dynamical networks by network control," *IEEE Trans. Automat. Contr.*, vol. 57, no. 6, June 2012.
- [52] S. E. Tuna, "Conditions for synchronizability in arrays of coupled linear systems," *IEEE Trans. Automat. Contr.*, vol. 54, no. 10, Oct. 2009.
- [53] L. Scardovi and R. Sepulchre, "Synchronization in networks of identical linear systems," *Automatica*, vol. 45, 2009.
- [54] Z. Li, Z. Duan, G. Chen, and L. Huang, "Consensus of multiagent systems and synchronization of complex networks: A unified viewpoint," *IEEE Trans. Circuits Syst. I*, vol. 57, no. 1, Jan. 2010.
- [55] T. Yang, A. A. Stoorvogel, and A. Saberi, "Consensus for multi-agent systems synchronization and regulation for complex networks," in 2011 American Control Conference, June 2011.
- [56] T. Yang, S. Roy, Y. Wan, and A. Saberi, "Constructing consensus controllers for networks with identical general linear agents," *International Journal of Robust and Nonlinear Control*, vol. 21, 2011.
- [57] K. You and L. Xie, "Network topology and communication data rate for consensusability of discrete-time multi-agent systems," *IEEE Trans. Automat. Contr.*, vol. 56, no. 10, Oct. 2011.
- [58] H. Zhang, F. L. Lewis, and A. Das, "Optimal design for synchronization of cooperative systems: State feedback, observer and output feedback," *IEEE Trans. Automat. Contr.*, vol. 56, no. 8, Aug. 2011.
- [59] Y. Su and J. Huang, "Two consensus problems for discrete-time multi-agent systems with switching network topology," *Automatica*, vol. 48, 2012.
- [60] F. Blaabjerg, R. Teodorescu, M. Liserre, and A. V. Timbus, "Overview of control and grid synchronization for distributed power generation systems," *IEEE Trans. Ind. Electron.*, vol. 53, no. 5, Oct. 2006.
- [61] M. Rohden, A. Sorge, M. Timme, and D. Witthaut, "Self-organized synchronization in decentralized power grids," *Phys. Rev. Lett.*, vol. 109, Aug. 2012.
- [62] F. Dörfler and F. Bullo, "Synchronization in complex networks of phase oscillatoers: A survey," *Automatica*, vol. 50, June 2014.
- [63] S.-H. Wang and E. J. Davison, "On the stabilization of decentralized control systems," *IEEE Trans. Automat. Contr.*, vol. AC-18, no. 5, Oct. 1973.
- [64] E. J. Davison and T. N. Chang, "Decentralized stabilization and pole assignment for general proper systems," *IEEE Trans. Automat. Contr.*, vol. 35, no. 6, June 1990.
- [65] B. D. O. Anderson and D. J. Clements, "Algebraic characterization of fixed modes in decentralized control," *Automatica*, vol. 17, no. 5, pp. 703–712, 1981.
- [66] E. J. Davison and U. Ozguner, "Characterizations of decentralized fixed modes for interconnected systems," *Automatica*, vol. 19, no. 2, pp. 169–182, 1983.
- [67] W. J. Rugh, *Linear System Theory*, 2nd ed. Prentice Hall, 1995.
- [68] J. P. Corfmat and A. S. Morse, "Decentralized control of linear multivariable systems," *Automatica*, vol. 12, pp. 479–495, 1976.
- [69] F. M. Brasch and J. B. Pearson, "Pole placement using dynamic compensators," *IEEE Trans. Automat. Contr.*, vol. AC-15, no. 1, Feb. 1970.

- [70] A. Molin and S. Hirche, "An iterative algorithm for optimal event-triggered estimation," in 4th IFAC Conference on Analysis and Design of Hybrid Systems, June 2012.
- [71] R. Radner, "Team decision problems," Ann. Math. Statist., vol. 33, no. 3, pp. 857– 881, Sept. 1962.
- [72] Y. Xu and J. P. Hespanha, "Optimal communication logics in networked control systems," in 43rd IEEE Conference on Decision and Control, Dec. 2004.
- [73] R. Cogill, S. Lall, and J. P. Hespanha, "A constant factor approximation algorithm for event-based sampling," in *American Control Conference*, July 2007.
- [74] G. M. Lipsa and N. C. Martins, "Remote state estimation with communication costs for first-order lti systems," *IEEE Trans. Automat. Contr.*, vol. 56, Sept. 2011.
- [75] L. Li and M. Lemmon, "Performance and average sampling period of sub-optimal triggering event in event triggered state estimation," in *50th IEEE Conference on Decision and Control and European Control Conference*, Dec. 2011.
- [76] A. Nayyar, T. Basar, D. Teneketzis, and V. V. Veeravalli, "Optimal strategies for communication and remote estimation with an energy harvesting sensor," *IEEE Trans. Automat. Contr.*, vol. 58, Nov. 2013.
- [77] L. Li, Z. Wang, and M. Lemmon, "Polynomial approximation of optimal event triggers for state estimation problems using SOSTOOLS," in *American Control Conference*, June 2013.
- [78] S. Park and N. Martins, "Individually optimal solutions to a remote state estimation problem with communication costs," in *Decision and Control (CDC)*, 2014 IEEE 53rd Annual Conference on, Dec 2014, pp. 4014–4019.
- [79] O. C. Imer and T. Basar, "Optimal estimation with limited measurements," in 44th *IEEE Conference on Decision and Control, and the European Control Conference* 2005, Dec. 2005.
- [80] H. Sandberg, M. Rabi, M. Skoglund, and K. H. Johansson, "Estimation over heterogeneous sensor networks," in 47th IEEE Conference on Decision and Control, Dec. 2008.
- [81] L. Li, M. Lemmon, and X. Wang, "Event-triggered state estimation in vector linear processes," in *American Control Conference*, June 2010.
- [82] L. Shi, K. H. Johansson, and L. Qiu, "Time and event-based sensor scheduling for networks with limited communication resources," in 18th IFAC World Congress, Aug. 2011.
- [83] C. Yang and L. Shi, "Deterministic sensor data scheduling under limited communication resource," *IEEE Trans. Signal Processing*, vol. 59, no. 10, Oct. 2011.

- [84] L. Shi, P. Cheng, and J. Chen, "Sensor data scheduling for optimal state estimation with communication energy constraint," *Automatica*, vol. 47, 2011.
- [85] M. Rabi, G. V. Moustakides, and J. S. Baras, "Adaptive sampling for linear state estimation," *SIAM Journal on Control and Optimization*, vol. 50, no. 2, 2012.
- [86] J. Sijs and M. Lazar, "Event based state estimation with time synchronous updates," *IEEE Trans. Automat. Contr.*, vol. 57, no. 10, Oct. 2012.
- [87] S. Trimpe and R. D'Andrea, "Event-based state estimation with variance-based triggering," in *51st IEEE Conference on Decision and Control*, Dec. 2012.
- [88] J. Wu, K. H. Johansson, and L. Shi, "An improved hybrid sensor schedule for remote state estimation under limited communication resources," in *IEEE 51st Annual Conference on Deicision and Control*, Dec. 2012.
- [89] J. Wu, Q.-S. Jia, K. H. Johansson, and L. Shi, "Event-based sensor data scheduling: Trade-off between communication rate and estimation quality," *IEEE Trans. Automat. Contr.*, vol. 58, no. 4, Apr. 2013.
- [90] M. Xia, V. Gupta, and P. J. Antsaklis, "Networked state estimation over a shared communication medium," in *American Control Conference*, June 2013.
- [91] Z. Ren, P. Cheng, J. Chen, L. Shi, and H. Zhang, "Dynamic sensor transmission power scheduling for remote state estimation," *Automatica*, vol. 50, Apr. 2014.
- [92] D. Han, Y. Mo, J. Wu, S. Weerakkody, B. Sinopoli, and L. Shi, "Stochastic eventtriggered sensor schedule for remote state estimation," *Automatic Control, IEEE Transactions on*, vol. PP, no. 99, pp. 1–1, 2015.
- [93] R. M. Dudley, *Real Analysis and Probability*, 2nd ed. Cambridge University Press, 2002, cambridge Books Online.
- [94] A. V. Skorokhod, "Limit theorems for stochastic processes," *Theory of Probability and Its Applications*, vol. 1, no. 3, pp. 261–290, 1956.
- [95] W. H. Sandholm, *Population Games and Evolutionary Dynamics*. MIT Press, 2010.
- [96] M. J. Smith, "The stability of a dynamic model of traffic assignmentan application of a method of lyapunov," *Transportation Science*, vol. 18, no. 3, pp. 245–252, 1984.
- [97] J. Hofbauer and W. H. Sandholm, "Evolution in games with randomly disturbed payoffs," *Journal of Economic Theory*, vol. 132, Jan. 2007.
- [98] J. C. Willems, "Dissipative dynamical systems part I: General theory," *Archive for Rational Mechanics and Analysis*, vol. 45, 1972.

- [99] D. Hill and P. Moylan, "The stability of nonlinear dissipative systems," *IEEE Trans. Automat. Contr.*, vol. 21, 1976.
- [100] W. H. Sandholm, "Probabilistic interpretations of integrability for game dynamics," *Dynamic Games and Applications*, Apr. 2013.
- [101] H. K. Khalil, Nonlinear Systems. Pretince Hall, 2001.
- [102] W. H. Sandholm, "Excessive payoff dynamics and other well-behaved evolutionary dynamics," *Journal of Economic Theory*, vol. 124, Oct. 2005.
- [103] ——, "Pairwise comparison dynamics and evolutionary foundations for nash equilibrium," *Games*, vol. 1, no. 1, 2010.
- [104] J. Hofbauer and W. H. Sandholm, "Survival of dominated strategies under evolutionary dynamics," *Theoretical Economics*, vol. 6, Sept. 2011.
- [105] L.-G. Mattsson and J. W. Weibull, "Probabilistic choice and procedurally bounded rationality," *Games and Economic Behavior*, vol. 41, pp. 61–78, 2002.
- [106] J. Hofbauer and W. H. Sandholm, "On the global convergence of stochastic fictitious play," *Econometrica*, vol. 70, no. 6, Nov. 2002.
- [107] D. McFadden, "Conditional logit analysis of qualitative choice behavior," in *In Frontiers in Econometrics*, P. Zarembka, Ed. Academic Press: New York, 1974, pp. 105–142.
- [108] A. Lindbeck, S. Nyberg, and J. W. Weibull, "Social norms and economic incentives in the welfare state," *The Quarterly Journal of Economics*, Feb. 1999.
- [109] T. H. Cormen, C. E. Leiserson, R. L. Rivest, and C. Stein, *Introduction to Algorithms*, 3rd ed. The MIT Press, 2009.
- [110] J.-M. Dion, C. Commault, and J. van der Woude, "Generic properties and control of linear structured systems: a survey," *Automatica*, vol. 39, pp. 1125–1144, 2003.
- [111] K. J. Reinschke, Multivariable Control: A Graph-Theoretic Approach. Springer Verlag, 1988.
- [112] F. Pasqualetti, A. Bicchi, and F. Bullo, "Consensus computation in unreliable networks: A system theoretic approach," *IEEE Transactions on Automatic Control*, vol. 57, no. 1, Jan. 2012.
- [113] J. S. Caughman and J. J. P. Veerman, "Kernels of directed graph Laplacian," *The Electronic Journal of Combinatorics*, vol. 13, 2006.
- [114] D. S. Dummit and R. M. Foote, Abstract Algebra. Wiley, 2004.
- [115] J. Fogaty, Invariant Theory. W. A. Benjamin, 1969.
- [116] P. Billingsley, *Probability and measure*, 3rd ed. Wiley, 1995.